Mission: Lifelong learners will be inspired and developed through effective teaching in a

safe and caring environment.

Vision: Respect - Pride - Excellence for All



# AGENDA BOARD OF EDUCATION - REGULAR MEETING

Instructional Planning Center/Huron Arena Tuesday, October 11, 2022

5:30 p.m.



- 1. Call to Order
- 2. Roll Call
- 3. Pledge of Allegiance
- 4. Adoption of the Agenda
- Dates to Remember

October 24 Board of Education Meeting - 5:30pm - IPC

November 2 Early Release

November 11 Veteran's Day Holiday - No School

November 14 Board of Education Meeting – 5:30pm – IPC

Nov 23, 24 & 25 Holiday Break - No School

November 28 Board of Education Meeting – 5:30pm – IPC

- 6. Community Input on Items Not on the Agenda
  - See Policy BFB Public Participation at Board Meetings for more information
- Conflict Disclosure and Consideration of Waivers The School Board will review the
  disclosures and determine if the transactions or the terms of the contracts are fair,
  reasonable, and not contrary to the public interest.
  a)
- 8. CONSENT AGENDA

The superintendent recommends approval of the following:

- a) Approval and/or Corrections of Minutes of Previous Meetings
- b) Approval and/or Corrections of the Financial Report
- c) Consideration and Approval of the Bills
- d) New Hires to the District

Classified personnel and substitute teachers must be approved in order to be covered by our workmen's compensation plan.

- 1) Vanya Munce/National Honor Society Supervisor/\$1,896 per year
- 2) Jaycee Clark/Special Education Para-Educator/Madison/\$19.16 per hour
- Paw Hser Eh/Substitute Teacher \$160 per day/Substitute Para-Educator -\$19.16 per hour
- Patricia Shoemaker/ Substitute Teacher \$160 per day/Substitute Para-Educator - \$19.16 per hour
- e) Resignations for Board Approval
  - Lisa McCarty/Instructional Coach-Title Teacher-Buchanan/33 years
- f) Contracts for Board Approval
  - 1)
- g) HEA Request to be Recognized for Certified Negotiations
- h) Advertising Agreement Renewal Tiger Stadium: Iverson CDJR-Huron

(The consent agenda may be approved with one motion. However, if a board member wishes to separate an item for discussion, he/she may do so.

### 9. CELEBRATE SUCCESSES IN THE DISTRICT

### Congratulations to:

- a) Huron Girls Tennis All-Conference Flight #5 Singles Say Ma, Senior
- b) <u>Huron Girls Tennis Teams for Finishing 4<sup>th</sup> at State Tournament.</u> Place winners are:

### Singles:

Flight 1: AnneClaire Rubish - 4th place

Flight 2: Elizabeth Heinen- 5th place

Flight 3: Bianca Medina- 5th place

Flight 6: Bethany Engelhart - 6th place

Flight 5: Say Ma- 4th place

Flight 6: Ann Hoek- 3rd place

**Doubles:** 

Flight 1: AnneClaire Rubish and Elizabeth Heinen- 3rd place

Flight 2: Bianca Medina and Bethany Engelhart- Consolation Champs

Flight 3: Say Ma and Ann Hoek- 2nd place

c) Huron Boys Soccer Team for Winning Huron's First ESD Boys Soccer Title

d) <u>Dianne (Administrative Assistant @ESL) & Jim Tapken</u> on the birth of their grandson, Henry Douglas Mills born September 30<sup>th</sup> in Tempe, AZ

#### Thank You to:

- HSD Principals October is National Principals Month
- HSD Custodians School Custodian Appreciation Day, October 2
- HSD Instructional Coaches Day, October 6
- School Lunch Week, October 10-14
- Boss's Day, October 16
- School Bus Safety Week, October 17-21
- Red Ribbon Week, October 23-31

### 10. REPORTS TO THE BOARD

- a) Business Manager's Report
- b) <u>Superintendent's Report</u>

### 11. OLD BUSINESS

a) HSD Policy DDA Gifts and Donations to District - 1st Reading

### 12. NEW BUSINESS

- a) Policy IHCDD Opioid Antagonists Administration Plan Introduction
- b) <u>ILG Architects Contracts for ESSER III and Capital Outlay Projects</u> ESSER III Projects and Cost Estimates
  - Tennis Court Expansion and New Fence \$900,000 \$1,100,000.
     JLG Fee \$89,000.
  - Building Controls for Buchanan, Madison, Washington \$700,000.
     JLG Fee \$49,500.
  - Air Condition Tiger Activity Center \$260,000. JLG Fee \$28,000.

### **Capital Outlay Projects and Cost Estimates**

- Resurface Track at Tiger Stadium \$250,000 -\$525,000. JLG Fee \$75,000.
- Replace Tiger Activity Center Ceiling Insulation \$150,000. JLG Fee \$8,500.

### 13. ADJOURNMENT

#### TEACHER'S CONTRACT

Huron School District No. 2-2, Huron, South Dakota

Vanya Munce September 30, 2022

YOU ARE DEREBY OFFICIALAY NOTHITIOD, that you have been elected as a Teacher in the fluron School District. No. 2-2, whose address is City of Duron on the annual salary basis of \$ 71518 for the sphool term, or the remaining part thereof, of the designated number of teaching days, inclusive of days arranged for pre-school planning, beginning <u>8/12/2022</u> and subject to the calendar, or modifications of the same, as adopted by the Board of Education. The salary is to be paid the twentieth day of each of the twelve calendar months.

Your election is subject to the school taws of the Smite of South Daketa and to the salary schedule and contractual elements rules and regulations of the Board of Education of the Huron School District No. 2-2, which are hereby by reference, incorporated in and made a part of this contract as though set forth herein at length, subject to the right of said Board to terminate the contract for cause, to be determined upon by the Board, and subject to your right to resign upon giving fairty (30) days notice thereof, in writing to said Board within 15 days of issuance of contract.

It is further contracted and agreed that your failure to complete the term of reaching prescribed herein for any cause, including out not limited to dismissal or resignation, constitutes a financial damage to the Suran School District No. 2-2 and that from the nature of the case it might be improvided for difficult to fix the actual damage. THIGREFORM, it is enderstood and agreed that your failure to complete the term provided herein shall result in the following liquidated duranges: failures occurring May 15 through May 31 for the casning year, damages shall be assessed at \$2,000.00, for breaking of a contract June 11 through Line 20, \$750.00 and for breaking of a contract June 21 through Line 20, \$750.00 and for breaking of a contract June 21 through Liny 31, damages shall be assessed at \$2,000.00 and breaking of contract August 1 and for the duration of the first semester, damages shall be assessed at \$3,000.00. Damages will be assessed at \$1,500.00 for breaking of a contract anytime during the 2<sup>rd</sup> semester. The Board reserves the right to request the Department of Education to suspend the employee's conflictation for one year in lieu of monetary damages in accordance with SDCL 15-42-9. Teachers who are not full-time employees of the district shall be assessed damages at a percentage which mesches their percent of employment.

It is further understood and agreed that resignations shall not become effective until approved by the Board of Education at the next meeting following receipt of said resignation. Further, it is hereby agreed that you will pay to the fitteen School District No. 2-2, or the Huran School District No. 2-2 will withhold or appropriate from any monies coved by them to you, and you hereby authorize such withholding or appropriation, the appropriate such herein above set furth as liquidated damages due to your faithire to complete said term.

This agreement becomes a binding contract when signed by the leacher and the Board of Education

Fired Nov 2019-JDC Math Tulor, 40-80 mins per day depending on enrolment, \$47.62 per nour 2021-2022\* The goove salary induces \$1586 (ES-0) Natl Honor Society Advisor

\*\*\*\*CONTRACT MUST BE SIGNED AND SETURNED TO THE SUPERINTENDENT'S OFFICE BY WED, OCT 5, 2022\*\*\*

SCHOOL DISTRICT NO. 2-2 OF THE CITY OF LIURON, BRADLE COUNTY, SOUTH DAKOTA

ATTEST:

By

Business Manager of the School District

Chairman of School District Board

TO THE BOARD OF EDUCATION OF THE MURON SCHOOL DISTRICT NO. 2-2

I hereby accept the position microtreed in the integrang contract of hiring in the Public Schools of Floren. South Dakote, at the salaty and upon and under the terus and conditions of the above and foregoing contract and have carefully resoluted and contract and such properties of the above and foregoing contract and have carefully resoluted and further than the properties of the above and foregoing contract and have carefully resoluted and further than the properties of the above and foregoing and the solution of this common. I accept that my pay with conse on October 1, a sumy employing may be terminated or suspended without the Caste of South Dakote. The document of this common.

pay until such time that  $\Gamma$  neet the deditionion requirer area of ( ) a job  $\Gamma$ 

Witness my hand this . 5th day of . OG other . 2022 Witness: M. chilli. Dick Soyn

CTTY OF HURON, BEADLE COUNTY, SOUTH DAKOTAL

Sign here: Varya Munco

I eacher



Peggy Heinz
Principal
5th and Mellette SW
Huron, SD 57350
P: (605) 353-7875
F: (605) 353-7877
Peggy.Heinz@k12.sd.us

Dear Mrs. Heinz and Dr. Steinhoff:

It is with mixed emotions that I write this letter. I will be retiring from the Instructional Coach position at Buchanan K-1 Center at the end of the 2022-2023 school year.

I have been in the district for 33 years and have enjoyed working with all of the different staff over the years. I love helping students learn to read and enjoy all of the staff at Buchanan. However, my eye condition is making it harder to do my job effectively and I need to step away from a career that has been wonderful to me!

I would like to thank the Huron School District for allowing me to do a job I have loved and will miss it!

Sincerely,

Lisa McCarty
Lisa McCarty



Mike.Carda@k12.sd.us

October 3, 2022

To whom it may concern,

On behalf of the HEA, I am writing to reduce tracognition for Certified Regulations for the 2023-24 academic school year. academic school year.

Thank you,

Michael Carde

HEA President

<u> Ç</u>ikizit<u>a</u>n

gigitalian gigi maretay

## TIGER STADIUM ADVERTISING AGREEMENT RENEWAL

| This adverti | sing agreement | renewal is made a  | nd entered into th | is 29th   | .∙<br>day of      |
|--------------|----------------|--|--------------------|-----------|-------------------|
| -7m/         | <u>)</u> 2022, | i renewal is made a<br>by and among IV<br>UCT 2-2, ("Owner | ERSON CDJR-1       | HURON ("/ | Advertiser"), and |
| HURON SÉ     | HOOL DISTR     | UCT 2-2, ("Owner   | ·").               |           | ,,                |

WHEREAS, Advertiser and Owner have entered into the Advertising Agreement for Tiger Stadium dated July 17, 2015 (as amended, the "Advertising Agreement") (a copy of the "Advertising Agreement" is attached); and

WHEREAS, the parties desire to extend and modify the Advertising Agreement as set forth herein.

Now, therefore, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

- Renewal/Extension of Term. Effective as of the date hereof, the Term set forth in the Advertising Agreement shall be extended to run for 1 (one) year from and after August 1, 2022, and ending on July 31, 2023.
- Advertising Fees. Effective as of the date hereof, the Advertising Fees for such extended term shall be \$1,500 (one thousand five hundred dollars) per year, payable in advance in accordance with the attached Payment Schedule.
- Ratification Except as expressly modified hereby, the remaining terms and conditions
  of the Advertising Agreement are hereby ratified and confirmed, and shall remain in
  full force and effect.

| Advertiser. | Initial |
|-------------|---------|



Owner Initial

### PAYMENT SCHEDULE

Agreement year 2022-2023: For services rendered August 1, 2022 to July 31, 2023.

Please select a renewal option:



### **Huron School District #2-2**

### Policies and Regulations

Code: DDA Gifts and Donations to District

### Gifts and Donations to District

The District may accept, own, manage, and dispose of any grant, gift, devise, bequest of money, real property or personal property. The Board may enter into an agreement for the receipt thereof upon such terms as shall be to the best interest of the District, and make such rules and regulations as it may deem best for the ownership, management, and control of the property.

The Board may authorize one or more private, nonprofit benevolent organizations, independently of the control of the school district, to accept and solicit donations, gifts, grants, or other private financial resources for the ultimate benefit of the educational or interscholastic activities of the school district. Funds collected pursuant to this authorization shall be offered to the District, and if accepted by the Board, the funds shall be placed in the Huron School District Foundation —a hub of the Huron Community Foundation and used for:

- · expansion and enhancement of academic curricula,
- · support of interscholastic activities,
- initiation or support of musical, forensic, civic, technologic, or similar extracurricular instruction or programs,
- · special trips or activities supporting core school activities,
- · other programs or activities of substantial benefit to the educational environment of the school district.

None of the funds received from gifts and donations may be used for debt retirement or ordinary expenditures or expenses.

The private, nonprofit benevolent organization may propose, advocate, or suggest to the Board that the funds offered by the organization be utilized with regard to various curricula, activities, or other programs. However, the Board is not obligated to accept or expend any funds, nor is the organization obligated to provide any funds, unless both mutually agree as to terms and purposes.

During each school year the Board may authorize any portion of the admission fees collected at a single event in each activity recognized by the SD High School Activities Association to be used by a local organization that supports local interscholastic activities. The portion of the admission fees provided to the local organization shall be used for a local endowment fund that is dedicated to assisting the District with its educational needs. The local organization shall maintain complete records of the admission fees it receives, and shall annually provide a report to the Board detailing the specific use of the admission fees.



### Huron School District #2-2

### Policies and Regulations

Code: JHCDD - Administration of Opioid Antagonists

#### Opioid Antagonists Administration Plan

The Board may acquire opioid antagonists and make them available to personnel who are trained by the SD Department of Health (SD DOH) or equivalent to possess and administer the medication for opioid overdose situation in accordance with state law and administrative rules.

Before school personnel may administer an opioid antagonist in the event of a suspected opioid overdose, training must be provided by an individual qualified to do so.

### The training must include:

- 1. Signs and symptoms of an opiate overdose;
- 2. Protocols and procedures for administration of an opioid antagonist;
- 3. Signs and symptoms of adverse responses to an opioid antagonist;
- 4. Protocols and procedures to stabilize the patient if an adverse response occurs;
- 5. Procedures for transporting, storing, and securing an opioid antagonist.
- 6. Opioid antagonist duration;
- The protocols and procedures for monitoring the suspected opioid overdose victim and re-administration of opioid antagonist if necessary for the safety and security of the suspected overdose victim; and
- 8. The method of opioid antagonist administration being taught.

Any school personnel who will have access to the medication and who may administer the medication must receive the required training. Training provided by the SD DOH is at no cost to the District. Training is not required for school personnel who will not have access to the medication or who will not potentially be administering the medication.

Naloxone is an opioid antagonist that comes in either an injectable form or a nasal spray. The medication provided to the schools through the SD DOH will be the nasal spray, and therefore, the SD DOH training will focus exclusively on the nasal spray. The cost of the medication provided by or through the SD DOH will be at no cost to the District. All opioid antagonists must be stored in a locked medicine cabinet, managed by the school nurse or school personnel trained in administration of opioid antagonists or administration of medication.

A standing order by a physician is required for the District to possess the medication. The SD DOH has identified a contract physician who will provide the standing order for the District.

Because opioid antagonists are used in opioid overdose emergency situations, prior parental consent is not required before administration of an opioid antagonist.\* Emergency medical services and the parents or guardians will be contacted immediately following the administration of an opioid antagonist.

The District will report naloxone use to the SD DOH on a form developed by SD DOH.

Pursuant to state law, no school district, administrator, school board member, school nurse, or designated school personnel possessing or making available opioid antagonists in accordance with state law, and no health care professional providing training in relation thereto, may be held liable for any injury or related damage that results from the administration of, the self-administration of, or the failure to administer an opioid antagonist, if such action or inaction constitutes, ordinary negligence. This immunity does not apply to an act or omission constituting gross, willful, or wanton negligence. The administration of an opioid antagonist does not constitute the practice of medicine. The immunity provided pursuant to SDCL 13-34A-24 is in addition to, and not in lieu of, any other immunity provided by law.



### Standard Form of Agreement Between Owner and Architect

AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two (In words, indicate day, month and year.)

BETWEEN the Architect's client identified as the Owner: (Name, legal status, address and other information)

Huron Public School District 150 5th St. SW Huron, SD 57350

and the Architect: (Name, legal status, address and other information)

JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104

for the following Project: (Name, location and detailed description)

21392.01 – Huron School District – Tennis Courts
Replacing fencing around the expanded courts, addition of four tennis courts (two on the west, two on the east), relocation of lighting if needed, and design of five shelters.

The Owner and Architect agree as follows.

#### ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

#### TABLE OF ARTICLES

- 1 INITIAL INFORMATION
- 2 ARCHITECT'S RESPONSIBILITIES
- 3 SCOPE OF ARCHITECT'S BASIC SERVICES
- 4 SUPPLEMENTAL AND ADDITIONAL SERVICES
- 5 OWNER'S RESPONSIBILITIES
- 6 COST OF THE WORK
- 7 COPYRIGHTS AND LICENSES
- 8 CLAIMS AND DISPUTES
- 9 TERMINATION OR SUSPENSION
- 10 MISCELLANEOUS PROVISIONS
- 11 COMPENSATION
- 12 SPECIAL TERMS AND CONDITIONS
- 13 SCOPE OF THE AGREEMENT

#### ARTICLE 1 INITIAL INFORMATION

§ 1.1 This Agreement is based on the Initial Information set forth in this Section 1.1.

(For each item in this section, insert the information or a statement such as "not applicable" or "unknown at time of execution.")

#### § 1.1.1 The Owner's program for the Project:

(Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.)

Replacing fencing around the expanded courts, addition of four tennis courts (two on the west, two on the east), relocation of lighting if needed, and design of five shelters

#### § 1.1.2 The Project's physical characteristics:

(Identify or describe pertinent information about the Project's physical characteristics, such as size; location; dimensions; geotechnical reports; site boundaries; topographic surveys; traffic and utility studies; availability of public and private utilities and services; legal description of the site, etc.)

Outdoor tennis courts located on the school property.

§ 1.1.3 The Owner's budget for the Cost of the Work, as defined in Section 6.1: (Provide total and, if known, a line item breakdown.)

\$1,100,000

- § 1.1.4 The Owner's anticipated design and construction milestone dates:
  - .1 Design phase milestone dates, if any:

Bidding: Late fall 2022

Construction to Commence spring/summer 2023

.2 Construction commencement date:

N/A

.3 Substantial Completion date or dates:

N/A

.4 Other milestone dates:

§ 1.1.5 The Owner intends the following procurement and delivery method for the Project: (Identify method such as competitive bid or negotiated contract, as well as any requirements for accelerated or fasttrack design and construction, multiple bid packages, or phased construction.)

Design/Bid/Build

§ 1.1.6 The Owner's anticipated Sustainable Objective for the Project: (Identify and describe the Owner's Sustainable Objective for the Project, if any.)

N/A

(Paragraph Deleted)

§ 1.1.6.1 If the Owner identifies a Sustainable Objective, the Owner and Architect shall complete and incorporate AIA Document E204TM-2017, Sustainable Projects Exhibit, into this Agreement to define the terms, conditions and services related to the Owner's Sustainable Objective. If E204-2017 is incorporated into this agreement, the Owner and Architect shall incorporate the completed E204-2017 into the agreements with the consultants and contractors performing services or Work in any way associated with the Sustainable Objective.

§ 1.1.7 The Owner identifies the following representative in accordance with Section 5.3: (List name, address, and other contact information.)

Kelly Christopheron Huron Public School District 150 5th St. SW Huron, SD 57350

Telephone:

Email: kelly.christopherson@k12.sd.us

§ 1.1.8 The persons or entities, in addition to the Owner's representative, who are required to review the Architect's submittals to the Owner are as follows:

(List name, address, and other contact information.)

N/A

§ 1.1.9 The Owner shall retain the following consultants and contractors: (List name, legal status, address, and other contact information.)

AIA Document B101" - 2017, Copyright @ 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved, The 'American "AIA." the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:09:09 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes: (3B9ADA4B)

.1 Geotechnical Engineer:

N/A

- .2 Civil Engineer:
- .3 Other, if any:
  (List any other consultants and contractors retained by the Owner.)

N/A

§ 1.1.10 The Architect identifies the following representative in accordance with Section 2.3: (List name, address, and other contact information.)

Catherine Dekkenga JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104 Telephone: 605-271-2531

Email: cdekkenga@jlgarchitects.com

§ 1.1.11 The Architect shall retain the consultants identified in Sections 1.1.11.1 and 1.1.11.2: (List name, legal status, address, and other contact information.)

§ 1.1.11.1 Consultants retained under Basic Services:

.1 Structural Engineer:

SEA, Inc., Mark Smith 401 E. 8th St. Suite 201 Sioux Falls, SD 57103 Telephone: 605-334-0188

.2 Mechanical Engineer:

N/A

Init.

#### .3 Electrical Engineer:

Associated Consulting Engineers, Inc., Brad Shoup 340 S. Phillips Ave. Sioux Falls, SD 57104 Telephone: 605-335-3720

#### .4 Civil Engineer:

Stockwell Engineers, Joe Brown 801 N. Phillips Ave. Suite 100 Sioux Falls, SD 57104 Telephone:

605-338-6668

§ 1.1.11.2 Consultants retained under Supplemental Services:

N/A

§ 1.1.12 Other Initial Information on which the Agreement is based:

N/A

- § 1.2 The Owner and Architect may rely on the Initial Information. Both parties, however, recognize that the Initial Information may materially change and, in that event, the Owner and the Architect shall appropriately adjust the Architect's services, schedule for the Architect's services, and the Architect's compensation. The Owner shall adjust the Owner's budget for the Cost of the Work and the Owner's anticipated design and construction milestones, as necessary, to accommodate material changes in the Initial Information.
- § 1.3 The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. The parties will use AIA Document E203™-2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.
- § 1.3.1 Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in AIA Document E203<sup>TM</sup>\_2013, Building Information Modeling and Digital Data Exhibit, and the requisite AIA Document G202<sup>TM</sup>\_2013, Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

#### ARTICLE 2 ARCHITECT'S RESPONSIBILITIES

§ 2.1 The Architect shall provide professional services as set forth in this Agreement. The Architect represents that it is properly licensed in the jurisdiction where the Project is located to provide the services required by this Agreement, or shall cause such services to be performed by appropriately licensed design professionals.

- § 2.2 The Architect shall perform its services consistent with the professional skill and care ordinarily provided by architects practicing in the same or similar locality under the same or similar circumstances. The Architect shall perform its services as expeditiously as is consistent with such professional skill and care and the orderly progress of the Project.
- § 2.3 The Architect shall identify a representative authorized to act on behalf of the Architect with respect to the Project.
- § 2.4 Except with the Owner's knowledge and consent, the Architect shall not engage in any activity, or accept any employment, interest or contribution that would reasonably appear to compromise the Architect's professional judgment with respect to this Project.
- § 2.5 The Architect shall maintain the following insurance until termination of this Agreement. If any of the requirements set forth below are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect as set forth in Section 11.9.
- § 2.5.1 Commercial General Liability with policy limits of One Million Dollars and Zero Cents (\$ 1,000,000.00 ) for each occurrence and Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) in the aggregate for bodily injury and property damage.
- § 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of One Million Dollars and Zero Cents (\$ 1,000,000.00 ) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.
- § 2.5.3 The Architect may achieve the required limits and coverage for Commercial General Liability and Automobile Liability through a combination of primary and excess or umbrella liability insurance, provided such primary and excess or umbrella liability insurance policies result in the same or greater coverage as the coverages required under Sections 2.5.1 and 2.5.2, and in no event shall any excess or umbrella liability insurance provide narrower coverage than the primary policy. The excess policy shall not require the exhaustion of the underlying limits only through the actual payment by the underlying insurers.
- § 2.5.4 Workers' Compensation at statutory limits.
- § 2.5.5 Employers' Liability with policy limits Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00) each accident, Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000.00) policy limit.
- § 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of Five Million Dollars and Zero Cents (\$ 5,000,000.00 ) per claim and Five Million Dollars and Zero Cents (\$ 5,000,000.00 ) in the aggregate.
- § 2.5.7 Additional Insured Obligations. To the fullest extent permitted by law, the Architect shall cause the primary and excess or umbrella polices for Commercial General Liability and Automobile Liability to include the Owner as an additional insured for claims caused in whole or in part by the Architect's negligent acts or omissions. The additional insured coverage shall be primary and non-contributory to any of the Owner's insurance policies and shall apply to both ongoing and completed operations.
- § 2.5.8 The Architect shall provide certificates of insurance to the Owner that evidence compliance with the requirements in this Section 2.5.

#### ARTICLE 3 SCOPE OF ARCHITECT'S BASIC SERVICES

- § 3.1 The Architect's Basic Services consist of those described in this Article 3 and include structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.
- § 3.1.1 The Architect shall manage the Architect's services, research applicable design criteria, attend Project meetings, communicate with members of the Project team, and report progress to the Owner.

- § 3.1.2 The Architect shall coordinate its services with those services provided by the Owner and the Owner's consultants. The Architect shall be entitled to rely on, and shall not be responsible for, the accuracy, completeness, and timeliness of, services and information furnished by the Owner and the Owner's consultants. The Architect shall provide prompt written notice to the Owner if the Architect becomes aware of any error, omission, or inconsistency in such services or information.
- § 3.1.3 As soon as practicable after the date of this Agreement, the Architect shall submit for the Owner's approval a schedule for the performance of the Architect's services. The schedule initially shall include anticipated dates for the commencement of construction and for Substantial Completion of the Work as set forth in the Initial Information. The schedule shall include allowances for periods of time required for the Owner's review, for the performance of the Owner's consultants, and for approval of submissions by authorities having jurisdiction over the Project. Once approved by the Owner, time limits established by the schedule shall not, except for reasonable cause, be exceeded by the Architect or Owner. With the Owner's approval, the Architect shall adjust the schedule, if necessary, as the Project proceeds until the commencement of construction.
- § 3.1.4 The Architect shall not be responsible for an Owner's directive or substitution, or for the Owner's acceptance of non-conforming Work, made or given without the Architect's written approval.
- § 3.1.5 The Architect shall contact governmental authorities required to approve the Construction Documents and entities providing utility services to the Project. The Architect shall respond to applicable design requirements imposed by those authorities and entities.
- § 3.1.6 The Architect shall assist the Owner in connection with the Owner's responsibility for filing documents required for the approval of governmental authorities having jurisdiction over the Project.

#### § 3.2 Schematic Design Phase Services

Init.

- § 3.2.1 The Architect shall review the program and other information furnished by the Owner, and shall review laws, codes, and regulations applicable to the Architect's services.
- § 3.2.2 The Architect shall prepare a preliminary evaluation of the Owner's program, schedule, budget for the Cost of the Work, Project site, the proposed procurement and delivery method, and other Initial Information, each in terms of the other, to ascertain the requirements of the Project. The Architect shall notify the Owner of (1) any inconsistencies discovered in the information, and (2) other information or consulting services that may be reasonably needed for the Project.
- § 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner approach(es) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.
- § 3.2.4 Based on the Project requirements agreed upon with the Owner, the Architect shall prepare and present, for the Owner's approval, a preliminary design illustrating the scale and relationship of the Project components.
- § 3.2.5 Based on the Owner's approval of the preliminary design, the Architect shall prepare Schematic Design Documents for the Owner's approval. The Schematic Design Documents shall consist of drawings and other documents including a site plan, if appropriate, and preliminary building plans, sections and elevations; and may include some combination of study models, perspective sketches, or digital representations. Preliminary selections of major building systems and construction materials shall be noted on the drawings or described in writing.
- § 3.2.5.1 The Architect shall consider sustainable design alternatives, such as material choices and building orientation, together with other considerations based on program and aesthetics, in developing a design that is consistent with the Owner's program, schedule and budget for the Cost of the Work. The Owner may obtain more advanced sustainable design services as a Supplemental Service under Section 4.1.1.
- § 3.2.5.2 The Architect shall consider the value of materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.

- § 3.2.6 The Architect shall submit to the Owner an estimate of the Cost of the Work prepared in accordance with Section 6.3.
- § 3.2.7 The Architect shall submit the Schematic Design Documents to the Owner, and request the Owner's approval.

#### § 3.3 Design Development Phase Services

- § 3.3.1 Based on the Owner's approval of the Schematic Design Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Design Development Documents for the Owner's approval. The Design Development Documents shall illustrate and describe the development of the approved Schematic Design Documents and shall consist of drawings and other documents including plans, sections, elevations, typical construction details, and diagrammatic layouts of building systems to fix and describe the size and character of the Project as to architectural, structural, mechanical and electrical systems, and other appropriate elements. The Design Development Documents shall also include outline specifications that identify major materials and systems and establish, in general, their quality levels.
- § 3.3.2 The Architect shall update the estimate of the Cost of the Work prepared in accordance with Section 6.3.
- § 3.3.3 The Architect shall submit the Design Development Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, and request the Owner's approval.

#### § 3.4 Construction Documents Phase Services

- § 3.4.1 Based on the Owner's approval of the Design Development Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Construction Documents for the Owner's approval. The Construction Documents shall illustrate and describe the further development of the approved Design Development Documents and shall consist of Drawings and Specifications setting forth in detail the quality levels and performance criteria of materials and systems and other requirements for the construction of the Work. The Owner and Architect acknowledge that, in order to perform the Work, the Contractor will provide additional information, including Shop Drawings, Product Data, Samples and other similar submittals, which the Architect shall review in accordance with Section 3.6.4.
- § 3.4.2 The Architect shall incorporate the design requirements of governmental authorities having jurisdiction over the Project into the Construction Documents.
- § 3.4.3 During the development of the Construction Documents, the Architect shall assist the Owner in the development and preparation of (1) procurement information that describes the time, place, and conditions of bidding, including bidding or proposal forms; (2) the form of agreement between the Owner and Contractor; and (3) the Conditions of the Contract for Construction (General, Supplementary and other Conditions). The Architect shall also compile a project manual that includes the Conditions of the Contract for Construction and Specifications, and may include bidding requirements and sample forms.
- § 3.4.4 The Architect shall update the estimate for the Cost of the Work prepared in accordance with Section 6.3.
- § 3.4.5 The Architect shall submit the Construction Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, take any action required under Section 6.5, and request the Owner's approval.

#### § 3.5 Procurement Phase Services

#### § 3.5.1 General

The Architect shall assist the Owner in establishing a list of prospective contractors. Following the Owner's approval of the Construction Documents, the Architect shall assist the Owner in (1) obtaining either competitive bids or negotiated proposals; (2) confirming responsiveness of bids or proposals; (3) determining the successful bid or proposal, if any; and, (4) awarding and preparing contracts for construction.

#### § 3.5.2 Competitive Bidding

§ 3.5.2.1 Bidding Documents shall consist of bidding requirements and proposed Contract Documents.

- § 3.5.2.2 The Architect shall assist the Owner in bidding the Project by:
  - .1 facilitating the distribution of Bidding Documents to prospective bidders;
  - .2 organizing and conducting a pre-bid conference for prospective bidders;
  - .3 preparing responses to questions from prospective bidders and providing clarifications and interpretations of the Bidding Documents to the prospective bidders in the form of addenda; and,
  - .4 organizing and conducting the opening of the bids, and subsequently documenting and distributing the bidding results, as directed by the Owner.
- § 3.5.2.3 If the Bidding Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective bidders.

#### § 3.5.3 Negotiated Proposals

- § 3.5.3.1 Proposal Documents shall consist of proposal requirements and proposed Contract Documents.
- § 3.5.3.2 The Architect shall assist the Owner in obtaining proposals by:
  - .1 facilitating the distribution of Proposal Documents for distribution to prospective contractors and requesting their return upon completion of the negotiation process;
  - .2 organizing and participating in selection interviews with prospective contractors;
  - .3 preparing responses to questions from prospective contractors and providing clarifications and interpretations of the Proposal Documents to the prospective contractors in the form of addenda; and,
  - .4 participating in negotiations with prospective contractors, and subsequently preparing a summary report of the negotiation results, as directed by the Owner.
- § 3.5.3.3 If the Proposal Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective contractors.

#### § 3.6 Construction Phase Services

#### § 3.6.1 General

- § 3.6.1.1 The Architect shall provide administration of the Contract between the Owner and the Contractor as set forth below and in AIA Document A201<sup>TM</sup>—2017, General Conditions of the Contract for Construction. If the Owner and Contractor modify AIA Document A201–2017, those modifications shall not affect the Architect's services under this Agreement unless the Owner and the Architect amend this Agreement.
- § 3.6.1.2 The Architect shall advise and consult with the Owner during the Construction Phase Services. The Architect shall have authority to act on behalf of the Owner only to the extent provided in this Agreement. The Architect shall not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, nor shall the Architect be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect shall be responsible for the Architect's negligent acts or omissions, but shall not have control over or charge of, and shall not be responsible for, acts or omissions of the Contractor or of any other persons or entities performing portions of the Work.
- § 3.6.1.3 Subject to Section 4.2 and except as provided in Section 3.6.6.5, the Architect's responsibility to provide Construction Phase Services commences with the award of the Contract for Construction and terminates on the date the Architect issues the final Certificate for Payment.

#### § 3.6.2 Evaluations of the Work

§ 3.6.2.1 The Architect shall visit the site at points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1)

known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.

- § 3.6.2.2 The Architect has the authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect shall have the authority to require inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not the Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, suppliers, their agents or employees, or other persons or entities performing portions of the Work.
- § 3.6.2.3 The Architect shall interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests shall be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- § 3.6.2.4 Interpretations and decisions of the Architect shall be consistent with the intent of, and reasonably inferable from, the Contract Documents and shall be in writing or in the form of drawings. When making such interpretations and decisions, the Architect shall endeavor to secure faithful performance by both Owner and Contractor, shall not show partiality to either, and shall not be liable for results of interpretations or decisions rendered in good faith. The Architect's decisions on matters relating to aesthetic effect shall be final if consistent with the intent expressed in the Contract Documents.
- § 3.6.2.5 Unless the Owner and Contractor designate another person to serve as an Initial Decision Maker, as that term is defined in AIA Document A201–2017, the Architect shall render initial decisions on Claims between the Owner and Contractor as provided in the Contract Documents.

#### § 3.6.3 Certificates for Payment to Contractor

- § 3.6.3.1 The Architect shall review and certify the amounts due the Contractor and shall issue certificates in such amounts. The Architect's certification for payment shall constitute a representation to the Owner, based on the Architect's evaluation of the Work as provided in Section 3.6.2 and on the data comprising the Contractor's Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified. The foregoing representations are subject to (1) an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, (2) results of subsequent tests and inspections, (3) correction of minor deviations from the Contract Documents prior to completion, and (4) specific qualifications expressed by the Architect.
- § 3.6.3.2 The issuance of a Certificate for Payment shall not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) ascertained how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.
- § 3.6.3.3 The Architect shall maintain a record of the Applications and Certificates for Payment.

#### § 3.6.4 Submittals

- § 3.6.4.1 The Architect shall review the Contractor's submittal schedule and shall not unreasonably delay or withhold approval of the schedule. The Architect's action in reviewing submittals shall be taken in accordance with the approved submittal schedule or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time, in the Architect's professional judgment, to permit adequate review.
- § 3.6.4.2 The Architect shall review, or take other appropriate action per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The

Init.

Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

- § 3.6.4.3 If the Contract Documents specifically require the Contractor to provide professional design services or certifications by a design professional related to systems, materials, or equipment, the Architect shall specify the appropriate performance and design criteria that such services must satisfy. The Architect shall review and take appropriate action on Shop Drawings and other submittals related to the Work designed or certified by the Contractor's design professional, provided the submittals bear such professional's seal and signature when submitted to the Architect. The Architect's review shall be for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect shall be entitled to rely upon, and shall not be responsible for, the adequacy and accuracy of the services, certifications, and approvals performed or provided by such design professionals.
- § 3.6.4.4 Subject to Section 4.2, the Architect shall review and respond to requests for information about the Contract Documents. The Architect shall set forth, in the Contract Documents, the requirements for requests for information. Requests for information shall include, at a minimum, a detailed written statement that indicates the specific Drawings or Specifications in need of clarification and the nature of the clarification requested. The Architect's response to such requests shall be made in writing within any time limits agreed upon, or otherwise with reasonable promptness. If appropriate, the Architect shall prepare and issue supplemental Drawings and Specifications in response to the requests for information.
- § 3.6.4.5 The Architect shall maintain a record of submittals and copies of submittals supplied by the Contractor in accordance with the requirements of the Contract Documents.

#### § 3.6.5 Changes in the Work

§ 3.6.5.1 The Architect may order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. Subject to Section 4.2, the Architect shall prepare Change Orders and Construction Change Directives for the Owner's approval and execution in accordance with the Contract Documents.

§ 3.6.5.2 The Architect shall maintain records relative to changes in the Work.

#### § 3.6.6 Project Completion

#### § 3.6.6.1 The Architect shall:

- .1 conduct inspections to determine the date or dates of Substantial Completion and the date of final completion;
- .2 issue Certificates of Substantial Completion;
- .3 forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract Documents and received from the Contractor; and,
- .4 issue a final Certificate for Payment based upon a final inspection indicating that, to the best of the Architect's knowledge, information, and belief, the Work complies with the requirements of the Contract Documents.
- § 3.6.6.2 The Architect's inspections shall be conducted with the Owner to check conformance of the Work with the requirements of the Contract Documents and to verify the accuracy and completeness of the list submitted by the Contractor of Work to be completed or corrected.
- § 3.6.6.3 When Substantial Completion has been achieved, the Architect shall inform the Owner about the balance of the Contract Sum remaining to be paid the Contractor, including the amount to be retained from the Contract Sum, if any, for final completion or correction of the Work.
- § 3.6.4 The Architect shall forward to the Owner the following information received from the Contractor: (1) consent of surety or sureties, if any, to reduction in or partial release of retainage or the making of final payment; (2) affidavits, receipts, releases and waivers of liens, or bonds indemnifying the Owner against liens; and (3) any other documentation required of the Contractor under the Contract Documents.

§ 3.6.6.5 Upon request of the Owner, and prior to the expiration of one year from the date of Substantial Completion, the Architect shall, without additional compensation, conduct a meeting with the Owner to review the facility operations and performance.

### § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- The Parties agree and understand that, during the course of the construction of the Work, the Owner,
  Contractor, or other Project participants may choose to, at their own expense and initiative, make use of
  supplemental video imagery and information sourced from devices such as project webcams, Unmanned
  Aerial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain
  visual information and documentation regarding the state of the progress and/or quality of the Work
  (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly
  agree and acknowledge the following:
  - Since Supplementary Video Information encompasses only certain limited visual information
    and provides neither complete nor comprehensive visual information/measurements/dimensions,
    nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video
    Information shall not be used as a substitute for or required supplement to Architect's Construction
    Phase services, herein, but rather may, to the extent provided and made available to Architect in a
    reasonably timely, legible, and complete fashion, supplement the other data and information
    obtained by Architect pursuant to Architect's Construction Phase Services, herein.
  - Architect shall have no obligation to conduct a comprehensive or continuous review of any Supplementary Video Information, but rather may, at its sole discretion, choose to review such Supplementary Video Information, or any portion thereof, in connection with the performance of its Construction Phase Services on the Project.
- 3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part thereof.

#### ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

#### § 4.1 Supplemental Services

§ 4.1.1 The services listed below are not included in Basic Services but may be required for the Project. The Architect shall provide the listed Supplemental Services only if specifically designated in the table below as the Architect's responsibility, and the Owner shall compensate the Architect as provided in Section 11.2. Unless otherwise specifically addressed in this Agreement, if neither the Owner nor the Architect is designated, the parties agree that the listed Supplemental Service is not being provided for the Project.

(Designate the Architect's Supplemental Services and the Owner's Supplemental Services required for the Project by indicating whether the Architect or Owner shall be responsible for providing the identified Supplemental Service. Insert a description of the Supplemental Services in Section 4.1.2 below or attach the description of services as an exhibit to this Agreement.)

| Supplemental Services  | Responsibility<br>(Architect, Owner, or not provided) |
|--|---|
| § 4.1.1.1 Programming  | Not Provided  |
| § 4.1.1.2 Multiple preliminary designs   | Not Provided  |
| § 4.1.1.3 Measured drawings  | Not Provided  |
| § 4.1.1.4 Existing facilities surveys  | Not Provided  |
| § 4.1.1.5 Site evaluation and planning   | Architect   |
| § 4.1.1.6 Building Information Model management responsibilities               | Not Provided  |
| § 4.1.1.7 Development of Building Information Models for post construction use | Not Provided  |

| § 4.1.1.8 Civil engineering   | Architect    |
|---|--------------|
| § 4.1.1.9 Landscape design  | Architect    |
| § 4.1.1.10 Architectural interior design                                | Not Provided |
| § 4.1.1.11 Value analysis   | Not Provided |
| § 4.1.1.12 Detailed cost estimating beyond that required in Section 6.3 | Not Provided |
| § 4.1.1.13 On-site project representation                               | Not Provided |
| § 4.1.1.14 Conformed documents for construction                         | Not Provided |
| § 4.1.1.15 As-designed record drawings                                  | Not Provided |
| § 4.1.1.16 As-constructed record drawings                               | Not Provided |
| § 4.1.1.17 Post-occupancy evaluation                                    | Not Provided |
| § 4.1.1.18 Facility support services                                    | Not Provided |
| § 4.1.1.19 Tenant-related services                                      | Not Provided |
| § 4.1.1.20 Architect's coordination of the Owner's consultants          | Not Provided |
| § 4.1.1.21 Telecommunications/data design                               | Not Provided |
| § 4.1.1.22 Security evaluation and planning                             | Not Provided |
| § 4.1.1.23 Commissioning  | Not Provided |
| § 4.1.1.24 Sustainable Project Services pursuant to Section 4.1.3       | Not Provided |
| § 4.1.1.25 Fast-track design services                                   | Not Provided |
| § 4.1.1.26 Multiple bid packages  | Not Provided |
| § 4.1.1.27 Historic preservation  | Not Provided |
| § 4.1.1.28 Furniture, furnishings, and equipment design                 | Not Provided |
| § 4.1.1.29 Other services provided by specialty Consultants             | Not Provided |
| § 4.1.1.30 Other Supplemental Services                                  | Not Provided |

#### § 4.1.2 Description of Supplemental Services

§ 4.1.2.1 A description of each Supplemental Service identified in Section 4.1.1 as the Architect's responsibility is provided below.

(Describe in detail the Architect's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit. The AIA publishes a number of Standard Form of Architect's Services documents that can be included as an exhibit to describe the Architect's Supplemental Services.)

#### SS 4.1.1.5 Site evaluation and planning

SS 4.1.1.8 Civil Engineering: Limited to usual and customary civil engineering services

SS 4.1.1.9 Landscape design: Limited to Schematic design phase services to indicate design intent and provide basis for cost allowance and bid alternates; based on accepted scope/alternates, remaining documentation will be provided design/build by landscaping subcontractor (Architect to review and approve final drawings for conformance with design intent and review shop drawings)

§ 4.1.2.2 A description of each Supplemental Service identified in Section 4.1.1 as the Owner's responsibility is provided below.

(Describe in detail the Owner's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit.)

#### N/A

§ 4.1.3 If the Owner identified a Sustainable Objective in Article 1, the Architect shall provide, as a Supplemental Service, the Sustainability Services required in AIA Document E204TM\_2017, Sustainable Projects Exhibit, attached to this Agreement. The Owner shall compensate the Architect as provided in Section 11.2.

#### § 4.2 Architect's Additional Services

The Architect may provide Additional Services after execution of this Agreement without invalidating the Agreement. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 4.2 shall entitle the Architect to compensation pursuant to Section 11.3 and an appropriate adjustment in the Architect's schedule.

§ 4.2.1 Upon recognizing the need to perform the following Additional Services, the Architect shall notify the Owner with reasonable promptness and explain the facts and circumstances giving rise to the need. The Architect shall not proceed to provide the following Additional Services until the Architect receives the Owner's written authorization:

- .1 Services necessitated by a change in the Initial Information, previous instructions or approvals given by the Owner, or a material change in the Project including size, quality, complexity, the Owner's schedule or budget for Cost of the Work, or procurement or delivery method;
- .2 Services necessitated by the enactment or revision of codes, laws, or regulations, including changing or editing previously prepared Instruments of Service;
- .3 Changing or editing previously prepared Instruments of Service necessitated by official interpretations of applicable codes, laws or regulations that are either (a) contrary to specific interpretations by the applicable authorities having jurisdiction made prior to the issuance of the building permit, or (b) contrary to requirements of the Instruments of Service when those Instruments of Service were prepared in accordance with the applicable standard of care;
- .4 Services necessitated by decisions of the Owner not rendered in a timely manner or any other failure of performance on the part of the Owner or the Owner's consultants or contractors;
- .5 Preparing digital models or other design documentation for transmission to the Owner's consultants and contractors, or to other Owner-authorized recipients;
- .6 Preparation of design and documentation for alternate bid or proposal requests proposed by the Owner;
- .7 Preparation for, and attendance at, a public presentation, meeting or hearing;
- .8 Preparation for, and attendance at, a dispute resolution proceeding or legal proceeding, except where the Architect is party thereto;
- .9 Evaluation of the qualifications of entities providing bids or proposals;
- .10 Consultation concerning replacement of Work resulting from fire or other cause during construction; or,
- .11 Assistance to the Initial Decision Maker, if other than the Architect.
- § 4.2.2 To avoid delay in the Construction Phase, the Architect shall provide the following Additional Services, notify the Owner with reasonable promptness, and explain the facts and circumstances giving rise to the need. If, upon receipt of the Architect's notice, the Owner determines that all or parts of the services are not required, the Owner shall give prompt written notice to the Architect of the Owner's determination. The Owner shall compensate the Architect for the services provided prior to the Architect's receipt of the Owner's notice.
  - Reviewing a Contractor's submittal out of sequence from the submittal schedule approved by the Architect;
  - .2 Responding to the Contractor's requests for information that are not prepared in accordance with the Contract Documents or where such information is available to the Contractor from a careful study and comparison of the Contract Documents, field conditions, other Owner-provided information, Contractor-prepared coordination drawings, or prior Project correspondence or documentation;
  - .3 Preparing Change Orders and Construction Change Directives that require evaluation of Contractor's proposals and supporting data, or the preparation or revision of Instruments of Service;
  - .4 Evaluating an extensive number of Claims as the Initial Decision Maker; or,
  - .5 Evaluating substitutions proposed by the Owner or Contractor and making subsequent revisions to Instruments of Service resulting therefrom.
- § 4.2.3 The Architect shall provide Construction Phase Services exceeding the limits set forth below as Additional Services. When the limits below are reached, the Architect shall notify the Owner:
  - .1 Two (2) reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor
  - .2 ( ) visits to the site by the Architect during construction
  - .3 One (1) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents
  - .4 One (1) inspections for any portion of the Work to determine final completion.

- § 4.2.4 Except for services required under Section 3.6.6.5 and those services that do not exceed the limits set forth in Section 4.2.3, Construction Phase Services provided more than 60 days after (1) the date of Substantial Completion of the Work or (2) the initial date of Substantial Completion identified in the agreement between the Owner and Contractor, whichever is earlier, shall be compensated as Additional Services to the extent the Architect incurs additional cost in providing those Construction Phase Services.
- § 4.2.5 If the services covered by this Agreement have not been completed within Twenty Four (24) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

#### ARTICLE 5 OWNER'S RESPONSIBILITIES

- § 5.1 Unless otherwise provided for under this Agreement, the Owner shall provide information in a timely manner regarding requirements for and limitations on the Project, including a written program, which shall set forth the Owner's objectives; schedule; constraints and criteria, including space requirements and relationships; flexibility; expandability; special equipment; systems; and site requirements.
- § 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes: (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.
- § 5.3 The Owner shall identify a representative authorized to act on the Owner's behalf with respect to the Project. The Owner shall render decisions and approve the Architect's submittals in a timely manner in order to avoid unreasonable delay in the orderly and sequential progress of the Architect's services.
- § 5.4 The Owner shall furnish surveys to describe physical characteristics, legal limitations and utility locations for the site of the Project, and a written legal description of the site. The surveys and legal information shall include, as applicable, grades and lines of streets, alleys, pavements and adjoining property and structures; designated wetlands; adjacent drainage; rights-of-way, restrictions, easements, encroachments, zoning, deed restrictions, boundaries and contours of the site; locations, dimensions, and other necessary data with respect to existing buildings, other improvements and trees; and information concerning available utility services and lines, both public and private, above and below grade, including inverts and depths. All the information on the survey shall be referenced to a Project benchmark.
- § 5.5 The Owner shall furnish services of geotechnical engineers, which may include test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.
- § 5.6 The Owner shall provide the Supplemental Services designated as the Owner's responsibility in Section 4.1.1.
- § 5.7 If the Owner identified a Sustainable Objective in Article 1, the Owner shall fulfill its responsibilities as required in AIA Document E204™\_2017, Sustainable Projects Exhibit, attached to this Agreement.
- § 5.8 The Owner shall coordinate the services of its own consultants with those services provided by the Architect. Upon the Architect's request, the Owner shall furnish copies of the scope of services in the contracts between the Owner and the Owner's consultants. The Owner shall furnish the services of consultants other than those designated as the responsibility of the Architect in this Agreement, or authorize the Architect to furnish them as an Additional Service, when the Architect requests such services and demonstrates that they are reasonably required by the scope of the Project. The Owner shall require that its consultants and contractors maintain insurance, including professional liability insurance, as appropriate to the services or work provided.

- § 5.9 The Owner shall furnish tests, inspections and reports required by law or the Contract Documents, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials.
- § 5.10 The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.
- § 5.11 The Owner shall provide prompt written notice to the Architect if the Owner becomes aware of any fault or defect in the Project, including errors, omissions or inconsistencies in the Architect's Instruments of Service.
- § 5.12 The Owner shall include the Architect in all communications with the Contractor that relate to or affect the Architect's services or professional responsibilities. The Owner shall promptly notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect.
- § 5.13 Before executing the Contract for Construction, the Owner shall coordinate the Architect's duties and responsibilities set forth in the Contract for Construction with the Architect's services set forth in this Agreement. The Owner shall provide the Architect a copy of the executed agreement between the Owner and Contractor, including the General Conditions of the Contract for Construction.
- § 5.14 The Owner shall provide the Architect access to the Project site prior to commencement of the Work and shall obligate the Contractor to provide the Architect access to the Work wherever it is in preparation or progress.
- § 5.15 Within 15 days after receipt of a written request from the Architect, the Owner shall furnish the requested information as necessary and relevant for the Architect to evaluate, give notice of, or enforce lien rights.

#### ARTICLE 6 COST OF THE WORK

- § 6.1 For purposes of this Agreement, the Cost of the Work shall be the total cost to the Owner to construct all elements of the Project designed or specified by the Architect and shall include contractors' general conditions costs, overhead and profit. The Cost of the Work also includes the reasonable value of labor, materials, and equipment, donated to, or otherwise furnished by, the Owner. The Cost of the Work does not include the compensation of the Architect; the costs of the land, rights-of-way, financing, or contingencies for changes in the Work; or other costs that are the responsibility of the Owner.
- § 6.2 The Owner's budget for the Cost of the Work is provided in Initial Information, and shall be adjusted throughout the Project as required under Sections 5.2, 6.4 and 6.5. Evaluations of the Owner's budget for the Cost of the Work, and the preliminary estimate of the Cost of the Work and updated estimates of the Cost of the Work, prepared by the Architect, represent the Architect's judgment as a design professional. It is recognized, however, that neither the Architect nor the Owner has control over the cost of labor, materials, or equipment; the Contractor's methods of determining bid prices; or competitive bidding, market, or negotiating conditions. Accordingly, the Architect cannot and does not warrant or represent that bids or negotiated prices will not vary from the Owner's budget for the Cost of the Work, or from any estimate of the Cost of the Work, or evaluation, prepared or agreed to by the Architect.
- § 6.3 In preparing estimates of the Cost of Work, the Architect shall be permitted to include contingencies for design, bidding, and price escalation; to determine what materials, equipment, component systems, and types of construction are to be included in the Contract Documents; to recommend reasonable adjustments in the program and scope of the Project; and to include design alternates as may be necessary to adjust the estimated Cost of the Work to meet the Owner's budget. The Architect's estimate of the Cost of the Work shall be based on current area, volume or similar conceptual estimating techniques. If the Owner requires a detailed estimate of the Cost of the Work, the Architect shall provide such an estimate, if identified as the Architect's responsibility in Section 4.1.1, as a Supplemental Service.
- § 6.4 If, through no fault of the Architect, the Procurement Phase has not commenced within 90 days after the Architect submits the Construction Documents to the Owner, the Owner's budget for the Cost of the Work shall be adjusted to reflect changes in the general level of prices in the applicable construction market.

§ 6.5 If at any time the Architect's estimate of the Cost of the Work exceeds the Owner's budget for the Cost of the Work, the Architect shall make appropriate recommendations to the Owner to adjust the Project's size, quality, or budget for the Cost of the Work, and the Owner shall cooperate with the Architect in making such adjustments.

§ 6.6 If the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services is exceeded by the lowest bona fide bid or negotiated proposal, the Owner shall

- .1 give written approval of an increase in the budget for the Cost of the Work;
- .2 authorize rebidding or renegotiating of the Project within a reasonable time;
- .3 terminate in accordance with Section 9.5;
- .4 in consultation with the Architect, revise the Project program, scope, or quality as required to reduce the Cost of the Work; or,
- .5 implement any other mutually acceptable alternative.

§ 6.7 If the Owner chooses to proceed under Section 6.6.4, the Architect shall modify the Construction Documents as necessary to comply with the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services, or the budget as adjusted under Section 6.6.1. If the Owner requires the Architect to modify the Construction Documents because the lowest bona fide bid or negotiated proposal exceeds the Owner's budget for the Cost of the Work due to market conditions the Architect could not reasonably anticipate, the Owner shall compensate the Architect for the modifications as an Additional Service pursuant to Section 11.3; otherwise the Architect's services for modifying the Construction Documents shall be without additional compensation. In any event, the Architect's modification of the Construction Documents shall be the limit of the Architect's responsibility under this Article 6.

#### ARTICLE 7 COPYRIGHTS AND LICENSES

§ 7.1 The Architect and the Owner warrant that in transmitting Instruments of Service, or any other information, the transmitting party is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project.

- § 7.2 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements or for similar purposes in connection with the Project is not to be construed as publication in derogation of the reserved rights of the Architect and the Architect's consultants.
- § 7.3 The Architect grants to the Owner a nonexclusive license to use the Architect's Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations under this Agreement, including prompt payment of all sums due pursuant to Article 9 and Article 11. The Architect shall obtain similar nonexclusive licenses from the Architect's consultants consistent with this Agreement. The license granted under this section permits the Owner to authorize the Contractor, Subcontractors, Sub-subcontractors, and suppliers, as well as the Owner's consultants and separate contractors, to reproduce applicable portions of the Instruments of Service, subject to any protocols established pursuant to Section 1.3, solely and exclusively for use in performing services or construction for the Project. If the Architect rightfully terminates this Agreement for cause as provided in Section 9.4, the license granted in this Section 7.3 shall terminate.
- § 7.3.1 In the event the Owner uses the Instruments of Service without retaining the authors of the Instruments of Service, the Owner releases the Architect and Architect's consultant(s) from all claims and causes of action arising from such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the Architect and its consultants from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's use of the Instruments of Service under this Section 7.3.1. The terms of this Section 7.3.1 shall not apply if the Owner rightfully terminates this Agreement for cause under Section 9.4.
- § 7.4 Except for the licenses granted in this Article 7, no other license or right shall be deemed granted or implied under this Agreement. The Owner shall not assign, delegate, sublicense, pledge or otherwise transfer any license granted herein to another party without the prior written agreement of the Architect. Any unauthorized use of the

Instruments of Service shall be at the Owner's sole risk and without liability to the Architect and the Architect's consultants.

§ 7.5 Except as otherwise stated in Section 7.3, the provisions of this Article 7 shall survive the termination of this Agreement.

#### ARTICLE 8 CLAIMS AND DISPUTES

#### § 8.1 General

- § 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.
- § 8.1.2 To the extent damages are covered by property insurance, the Owner and Architect waive all rights against each other and against the contractors, consultants, agents, and employees of the other for damages, except such rights as they may have to the proceeds of such insurance as set forth in AIA Document A201–2017, General Conditions of the Contract for Construction. The Owner or the Architect, as appropriate, shall require of the contractors, consultants, agents, and employees of any of them, similar waivers in favor of the other parties enumerated herein.
- § 8.1.3 The Architect and Owner waive consequential damages for claims, disputes, or other matters in question, arising out of or relating to this Agreement. This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination of this Agreement, except as specifically provided in Section 9.7.

#### § 8.2 Mediation

- § 8.2.1 Any claim, dispute or other matter in question arising out of or related to this Agreement shall be subject to mediation as a condition precedent to binding dispute resolution. If such matter relates to or is the subject of a lien arising out of the Architect's services, the Architect may proceed in accordance with applicable law to comply with the lien notice or filing deadlines prior to resolution of the matter by mediation or by binding dispute resolution.
- § 8.2.2 The Owner and Architect shall endeavor to resolve claims, disputes and other matters in question between them by mediation, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of this Agreement. A request for mediation shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of a complaint or other appropriate demand for binding dispute resolution but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration proceeding is stayed pursuant to this section, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.
- § 8.2.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.
- § 8.2.4 If the parties do not resolve a dispute through mediation pursuant to this Section 8.2, the method of binding dispute resolution shall be the following: (Check the appropriate box.)

| [ ] | Arbitration pursuant to Section 8.3 of this Agreement |
|-----|---|
| [X] | Litigation in a court of competent jurisdiction       |
| []  | Other: (Specify)                                      |

If the Owner and Architect do not select a method of binding dispute resolution, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, the dispute will be resolved in a court of competent jurisdiction.

#### § 8.3 Arbitration

- § 8.3.1 If the parties have selected arbitration as the method for binding dispute resolution in this Agreement, any claim, dispute or other matter in question arising out of or related to this Agreement subject to, but not resolved by, mediation shall be subject to arbitration, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of this Agreement. A demand for arbitration shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the arbitration.
- § 8.3.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the claim, dispute or other matter in question would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the claim, dispute or other matter in question.
- § 8.3.2 The foregoing agreement to arbitrate, and other agreements to arbitrate with an additional person or entity duly consented to by parties to this Agreement, shall be specifically enforceable in accordance with applicable law in any court having jurisdiction thereof.
- § 8.3.3 The award rendered by the arbitrator(s) shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

#### § 8.3.4 Consolidation or Joinder

- § 8.3.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation; (2) the arbitrations to be consolidated substantially involve common questions of law or fact; and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).
- § 8.3.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.
- § 8.3.4.3 The Owner and Architect grant to any person or entity made a party to an arbitration conducted under this Section 8.3, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Architect under this Agreement.
- § 8.4 The provisions of this Article 8 shall survive the termination of this Agreement.

#### ARTICLE 9 TERMINATION OR SUSPENSION

§ 9.1 If the Owner fails to make payments to the Architect in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Architect's option, cause for suspension of performance of services under this Agreement. If the Architect elects to suspend services, the Architect shall give seven days' written notice to the Owner before suspending services. In the event of a suspension of services, the

Architect shall have no liability to the Owner for delay or damage caused the Owner because of such suspension of services. Before resuming services, the Owner shall pay the Architect all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

- § 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.3 If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Architect, the Architect may terminate this Agreement by giving not less than seven days' written notice.
- § 9.4 Either party may terminate this Agreement upon seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.
- § 9.5 The Owner may terminate this Agreement upon seven days' written notice to the Architect for the Owner's convenience and without cause.
- § 9.6 If the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall compensate the Architect for services performed prior to termination, Reimbursable Expenses incurred, and costs attributable to termination, including the costs attributable to the Architect's termination of consultant agreements.
- § 9.7 In addition to any amounts paid under Section 9.6, if the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall pay to the Architect the following fees:

(Set forth below the amount of any termination or licensing fee, or the method for determining any termination or licensing fee.)

.1 Termination Fee:

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

.2 Licensing Fee if the Owner intends to continue using the Architect's Instruments of Service:

As mutually agreed

- § 9.8 Except as otherwise expressly provided herein, this Agreement shall terminate one year from the date of Substantial Completion.
- § 9.9 The Owner's rights to use the Architect's Instruments of Service in the event of a termination of this Agreement are set forth in Article 7 and Section 9.7.

#### ARTICLE 10 MISCELLANEOUS PROVISIONS

- § 10.1 This Agreement shall be governed by the law of the place where the Project is located, excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 8.3.
- § 10.2 Terms in this Agreement shall have the same meaning as those in AIA Document A201–2017, General Conditions of the Contract for Construction.
- § 10.3 The Owner and Architect, respectively, bind themselves, their agents, successors, assigns, and legal representatives to this Agreement. Neither the Owner nor the Architect shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for

the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement, including any payments due to the Architect by the Owner prior to the assignment.

- § 10.4 If the Owner requests the Architect to execute certificates, the proposed language of such certificates shall be submitted to the Architect for review at least 14 days prior to the requested dates of execution. If the Owner requests the Architect to execute consents reasonably required to facilitate assignment to a lender, the Architect shall execute all such consents that are consistent with this Agreement, provided the proposed consent is submitted to the Architect for review at least 14 days prior to execution. The Architect shall not be required to execute certificates or consents that would require knowledge, services, or responsibilities beyond the scope of this Agreement.
- § 10.5 Nothing contained in this Agreement shall create a contractual relationship with, or a cause of action in favor of, a third party against either the Owner or Architect.
- § 10.6 Unless otherwise required in this Agreement, the Architect shall have no responsibility for the discovery, presence, handling, removal or disposal of, or exposure of persons to, hazardous materials or toxic substances in any form at the Project site.
- § 10.7 The Architect shall have the right to include photographic or artistic representations of the design of the Project among the Architect's promotional and professional materials. The Architect shall be given reasonable access to the completed Project to make such representations. However, the Architect's materials shall not include the Owner's confidential or proprietary information if the Owner has previously advised the Architect in writing of the specific information considered by the Owner to be confidential or proprietary. The Owner shall provide professional credit for the Architect in the Owner's promotional materials for the Project. This Section 10.7 shall survive the termination of this Agreement unless the Owner terminates this Agreement for cause pursuant to Section 9.4.
- § 10.8 If the Architect or Owner receives information specifically designated as "confidential" or "business proprietary," the receiving party shall keep such information strictly confidential and shall not disclose it to any other person except as set forth in Section 10.8.1. This Section 10.8 shall survive the termination of this Agreement.
- § 10.8.1 The receiving party may disclose "confidential" or "business proprietary" information after 7 days' notice to the other party, when required by law, arbitrator's order, or court order, including a subpoena or other form of compulsory legal process issued by a court or governmental entity, or to the extent such information is reasonably necessary for the receiving party to defend itself in any dispute. The receiving party may also disclose such information to its employees, consultants, or contractors in order to perform services or work solely and exclusively for the Project, provided those employees, consultants and contractors are subject to the restrictions on the disclosure and use of such information as set forth in this Section 10.8.
- § 10.9 The invalidity of any provision of the Agreement shall not invalidate the Agreement or its remaining provisions. If it is determined that any provision of the Agreement violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Agreement shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Agreement.

#### ARTICLE 11 COMPENSATION

§ 11.1 For the Architect's Basic Services described under Article 3, the Owner shall compensate the Architect as follows:

 Stipulated Sum (Insert amount)

Compensation in the amount of Eighty Nine Thousand Dollars and Zero Cents (\$89,000.00)

 Percentage Basis (Insert percentage value)

( ) % of the Owner's budget for the Cost of the Work, as calculated in accordance with Section 11.6.

.3 Other (Describe the method of compensation)

§ 11.2 For the Architect's Supplemental Services designated in Section 4.1.1 and for any Sustainability Services required pursuant to Section 4.1.3, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation. If necessary, list specific services to which particular methods of compensation apply.)

Hourly or as mutually agreed upon in advance of services

§ 11.3 For Additional Services that may arise during the course of the Project, including those under Section 4.2, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation.)

Hourly or as mutually agreed upon in advance of services

§ 11.4 Compensation for Supplemental and Additional Services of the Architect's consultants when not included in Section 11.2 or 11.3, shall be the amount invoiced to the Architect plus percent (%), or as follows:

(Insert amount of, or basis for computing, Architect's consultants' compensation for Supplemental or Additional Services.)

As mutually agreed by Owner and Architect in advance of services

§ 11.5 When compensation for Basic Services is based on a stipulated sum or a percentage basis, the proportion of compensation for each phase of services shall be as follows:

| Total Basic Compensation | one hundred | percent ( | 100 | %)   |
|--------------------------|-------------|-----------|-----|------|
| Construction Phase       | Twenty Five | percent ( | 25  | %)   |
| Procurement Phase        | Five        | percent ( | 5   | %)   |
| Phase                    |             | 0.5       |     | - 50 |
| Construction Documents   | Twenty Five | percent ( | 25  | %)   |
| Design Development Phase | Twenty Five | percent ( | 25  | %)   |
| Schematic Design Phase   | Twenty      | percent ( | 20  | %)   |
|                          |             |           |     |      |

- § 11.6 When compensation identified in Section 11.1 is on a percentage basis, progress payments for each phase of Basic Services shall be calculated by multiplying the percentages identified in this Article by the Owner's most recent budget for the Cost of the Work. Compensation paid in previous progress payments shall not be adjusted based on subsequent updates to the Owner's budget for the Cost of the Work.
- § 11.6.1 When compensation is on a percentage basis and any portions of the Project are deleted or otherwise not constructed, compensation for those portions of the Project shall be payable to the extent services are performed on those portions. The Architect shall be entitled to compensation in accordance with this Agreement for all services performed whether or not the Construction Phase is commenced.
- § 11.7 The hourly billing rates for services of the Architect and the Architect's consultants are set forth below. The rates shall be adjusted in accordance with the Architect's and Architect's consultants' normal review practices. (If applicable, attach an exhibit of hourly billing rates or insert them below.)

See Exhibit C - Hourly Compensation and Reimbursable Expenses

**Employee or Category** 

Rate (\$0.00)

#### § 11.8 Compensation for Reimbursable Expenses

§ 11.8.1 Reimbursable Expenses are in addition to compensation for Basic, Supplemental, and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:

- .1 Transportation and authorized out-of-town travel and subsistence;
- .2 Long distance services, dedicated data and communication services, teleconferences, Project web sites, and extranets;
- .3 Permitting and other fees required by authorities having jurisdiction over the Project;
- .4 Printing, reproductions, plots, and standard form documents;
- .5 Postage, handling, and delivery;
- .6 Expense of overtime work requiring higher than regular rates, if authorized in advance by the Owner;
- .7 Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner or required for the Project;
- .8 If required by the Owner, and with the Owner's prior written approval, the Architect's consultants' expenses of professional liability insurance dedicated exclusively to this Project, or the expense of additional insurance coverage or limits in excess of that normally maintained by the Architect's consultants;
- All taxes levied on professional services and on reimbursable expenses;
- .10 Site office expenses;
- .11 Registration fees and any other fees charged by the Certifying Authority or by other entities as necessary to achieve the Sustainable Objective; and,
- .12 Other similar Project-related expenditures.

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus Ten percent (10 %) of the expenses incurred.

§ 11.9 Architect's Insurance. If the types and limits of coverage required in Section 2.5 are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect for the additional costs incurred by the Architect for the additional coverages as set forth below:

(Insert the additional coverages the Architect is required to obtain in order to satisfy the requirements set forth in Section 2.5, and for which the Owner shall reimburse the Architect.)

N/A

#### § 11.10 Payments to the Architect

#### § 11.10.1 Initial Payments

§ 11.10.1.1 An initial payment of Zero (\$ 0 ) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of Zero (\$0) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred.

#### § 11.10.2 Progress Payments

§ 11.10.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid Forty-Five (45) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect. (Insert rate of monthly or annual interest agreed upon.)

#### 1.5 % Monthly

§ 11.10.2.2 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work, unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.

§ 11.10.2.3 Records of Reimbursable Expenses, expenses pertaining to Supplemental and Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

§ 11.10.2.4 Owner Agrees to receive invoices by email, which will be sent to the following individual and email address:

Full Name: Kelly Christopherson

Email Address: kelly.christopherson@k12.sd.us

#### ARTICLE 12 SPECIAL TERMS AND CONDITIONS

Special terms and conditions that modify this Agreement are as follows: (Include other terms and conditions applicable to this Agreement.)

None

#### ARTICLE 13 SCOPE OF THE AGREEMENT

§ 13.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both the Owner and Architect.

§ 13.2 This Agreement is comprised of the following documents identified below:

- .1 AIA Document B101TM-2017, Standard Form Agreement Between Owner and Architect
- .2 AIA Document E203<sup>TM</sup>—2013, Building Information Modeling and Digital Data Exhibit, dated as indicated below:

(Insert the date of the E203-2013 incorporated into this agreement.)

.3 Exhibits:

(Check the appropriate box for any exhibits incorporated into this Agreement.)

[ ] AIA Document E204<sup>TM</sup>–2017, Sustainable Projects Exhibit, dated as indicated below: (Insert the date of the E204-2017 incorporated into this agreement.)

[ ] Other Exhibits incorporated into this Agreement:

(Clearly identify any other exhibits incorporated into this Agreement, including any exhibits and scopes of services identified as exhibits in Section 4.1.2.)

Exhibit A: Project Initial Information

Exhibit B: JLG Image Release

Exhibit C: Schedule of Hourly Compensation & Reimbursable Expenses

.4 Other documents:

(List other documents, if any, forming part of the Agreement,)

This Agreement entered into as of the day and year first written above.

| OWNER (Signature)          | ARCHITECT (Signature)                                  |
|----------------------------|--|
| Garret Bischoff, President | Catherine Dekkenga, AIA, Principal                     |
| (Printed name and title)   | (Printed name, title, and license number, if required) |

1

### Additions and Deletions Report for

AIA® Document B101™ - 2017

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 16:09:09 ET on 10/05/2022.

#### PAGE 1

AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two

Huron Public School District 150 5th St. SW Huron, SD 57350

JLG Architects
232 S. Main Ave.
Sioux Falls, SD 57104

21392.01 - Huron School District - Tennis Courts

Replacing fencing around the expanded courts, addition of four tennis courts (two on the west, two on the east), relocation of lighting if needed, and design of five shelters.

#### PAGE 2

#### **TABLE OF ARTICLES**

(Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.)

Replacing fencing around the expanded courts, addition of four tennis courts (two on the west, two on the east), relocation of lighting if needed, and design of five shelters

Outdoor tennis courts located on the school property. \$1,100,000 PAGE 3 Bidding: Late fall 2022 Construction to Commence spring/summer 2023 N/A N/A Design/Bid/Build N/A Kelly Christopheron Huron Public School District 150 5th St. SW Huron, SD 57350 Telephone: Email: kelly.christopherson@k12.sd.us \*\*\* N/A PAGE 4 N/A

N/A

\*\*\*

Catherine Dekkenga
JLG Architects
232 S. Main Ave.
Sioux Falls, SD 57104
Telephone: 605-271-2531
Email: cdekkenga@jlgarchitects.com

•••

SEA, Inc., Mark Smith

•••

401 E. 8th St.

\*\*\*

Suite 201

\*\*\*

Sioux Falls, SD 57103

•••

Telephone: 605-334-0188

...

N/A

PAGE 5

Associated Consulting Engineers, Inc., Brad Shoup

\*\*\*

340 S. Phillips Ave.

•••

Sioux Falls, SD 57104

\*\*\*

Telephone: 605-335-3720

| ***                                    |   |
|--|---|
| <u>.4 Civ</u>                          | il Engineer:  |
|  |   |
| <u>St</u>                              | tockwell Engineers, Joe Brown   |
|  |   |
| 80                                     | 01 N. Phillips Ave.   |
|  |   |
| St                                     | nite 100  |
| 111                                    |   |
| Si                                     | ioux Falls, SD 57104  |
|  |   |
| <u>To</u>                              | elephone:   |
|  |   |
| <u>\$-605-338-6668</u>                 |   |
|  |   |
|  |   |
| § 1.1.11.2 Consult                     | tants retained under Supplemental Services:   |
| ************************************** |   |
| N/A_                                   |   |
|  |   |
| N/A                                    |   |
| PAGE 6                                 |   |
| Cents (\$ 1,000,0                      | ial General Liability with policy limits of not less than (\$\text{One Million Dollars and Zero} \) 000.00 ) for each occurrence and \(\text{Two Million Dollars and Zero Cents}\) (\$\text{2,000,000.00}\) ) in the lily injury and property damage. |

§ 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of not less than (\$\frac{1}{2}\) One Million Dollars and Zero Cents (\$\frac{1}{2}\) (\$\frac{1}{2}\) (\$\frac{1}{2}\) (000,000.00 ) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.

§ 2.5.5 Employers' Liability with policy limits not less than (\$ ) each accident, (\$ Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each accident, Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000.00 ) policy limit.

§ 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of not less than (\$-Five Million Dollars and Zero Cents (\$ 5,000,000.00 ) per claim and Five Million Dollars and Zero Cents (\$ 5,000,000.00 ) in the aggregate.

§ 3.1 The Architect's Basic Services consist of those described in this Article 3 and include usual and customary structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.

#### PAGE 7

§ 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner alternative approaches approach(cs) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.

...

§ 3.2.5.2 The Architect shall consider the value of alternative materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.

#### PAGE 10

§ 3.6.2.1 The Architect shall visit the site at intervals appropriate to the stage points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.

# PAGE 11

§ 3.6.4.2 The Architect shall review and approve, review, or take other appropriate action upon, per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

#### PAGE 12

# § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- 1. The Parties agree and understand that, during the course of the construction of the Work, the Owner, Contractor, or other Project participants may choose to, at their own expense and initiative, make use of supplemental video imagery and information sourced from devices such as project webcams. Unmanned Aerial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain visual information and documentation regarding the state of the progress and/or quality of the Work (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly agree and acknowledge the following:
  - 1. Since Supplementary Video Information encompasses only certain limited visual information and provides neither complete nor comprehensive visual information/measurements/dimensions, nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video Information shall not be used as a substitute for or required supplement to Architect's Construction Phase services, herein, but rather may, to the extent provided and made available to Architect in a reasonably timely, legible, and complete fashion, supplement the other data and information obtained by Architect pursuant to Architect's Construction Phase Services, herein.
  - 2. Architect shall have no obligation to conduct a comprehensive or continuous review of any Supplementary Video Information, but rather may, at its sole discretion, choose to review such Supplementary Video Information, or any portion thereof, in connection with the performance of its Construction Phase Services on the Project.

3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part

# ARTICLE thereof.

# ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

#### PAGE 13

| § 4.1.1.1 Programming  | Not Provided |
|--|--------------|
| § 4.1.1.2 Multiple preliminary designs   | Not Provided |
| § 4.1.1.3 Measured drawings  | Not Provided |
| § 4.1.1.4 Existing facilities surveys  | Not Provided |
| § 4.1.1.5 Site evaluation and planning   | Architect    |
| § 4.1.1.6 Building Information Model management responsibilities               | Not Provided |
| § 4.1.1.7 Development of Building Information Models for post construction use | Not Provided |
| § 4.1.1.8 Civil engineering  | Architect    |
| § 4.1.1.9 Landscape design   | Architect    |
| § 4.1.1.10 Architectural interior design                                       | Not Provided |
| § 4.1.1.11 Value analysis  | Not Provided |

Additions and Deletions Report for AIA Document B101\* - 2017, Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects, All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:09:09 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

| § 4.1.1.12 | Detailed cost estimating beyond that required in Section 6.3 | Not Provided |
|------------|--|--------------|
| § 4.1.1.13 | On-site project representation                               | Not Provided |
| § 4.1.1.14 | Conformed documents for construction                         | Not Provided |
| § 4.1.1.15 | As-designed record drawings                                  | Not Provided |
| § 4.1.1.16 | As-constructed record drawings                               | Not Provided |
| § 4.1.1.17 | Post-occupancy evaluation                                    | Not Provided |
| § 4.1.1.18 | Facility support services                                    | Not Provided |
| § 4.1.1.19 | Tenant-related services                                      | Not Provided |
| § 4.1.1.20 | Architect's coordination of the Owner's consultants          | Not Provided |
| § 4.1.1.21 | Telecommunications/data design                               | Not Provided |
| § 4.1.1.22 | Security evaluation and planning                             | Not Provided |
| § 4.1.1.23 | Commissioning  | Not Provided |
| § 4.1.1.24 | Sustainable Project Services pursuant to Section 4.1.3       | Not Provided |
| § 4.1.1.25 | Fast-track design services                                   | Not Provided |
| § 4.1.1.26 | Multiple bid packages  | Not Provided |
| § 4.1.1.27 | Historic preservation  | Not Provided |
| § 4.1.1.28 | Furniture, furnishings, and equipment design                 | Not Provided |
| 4.1.1.29   | Other services provided by specialty Consultants             | Not Provided |
| § 4.1.1.30 | Other Supplemental Services                                  | Not Provided |

# SS 4.1.1.5 Site evaluation and planning

SS 4.1.1.8 Civil Engineering: Limited to usual and customary civil engineering services SS 4.1.1.9 Landscape design: Limited to Schematic design phase services to indicate design intent and provide basis for cost allowance and bid alternates; based on accepted scope/alternates, remaining documentation will be provided design/build by landscaping subcontractor (Architect to review and approve final drawings for conformance with design intent and review shop drawings)

#### N/A

...

#### PAGE 14

- .1 Two (2) reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor
- .3 Onc (1) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents
- .4 One (1 ) inspections for any portion of the Work to determine final completion.

# PAGE 15

§ 4.2.5 If the services covered by this Agreement have not been completed within <a href="Twenty Four">Twenty Four</a> (<a href="24">24</a>) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

...

§ 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1; 5.1; (2) the Owner's other costs, and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes: (1) To account for coordination issues in the Construction Documents, including errors and omissions. (2) to modify or change the scope of the project. (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.

#### PAGE 18

§ 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.

#### PAGE 19

[X] Litigation in a court of competent jurisdiction

# PAGE 20

§ 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

...

§ 9.4 Either party may terminate this Agreement upon not less than seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.

•••

§ 9.5 The Owner may terminate this Agreement upon not less than seven days' written notice to the Architect for the Owner's convenience and without cause.

.

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

As mutually agreed

#### PAGE 21

Compensation in the amount of Eighty Nine Thousand Dollars and Zero Cents (\$89,000.00)

#### PAGE 22

Hourly or as mutually agreed upon in advance of services

Hourly or as mutually agreed upon in advance of services

As mutually agreed by Owner and Architect in advance of services

| Schematic Design Phase   | Twenty      | percent ( | 20      | %) |
|--------------------------|-------------|-----------|---------|----|
| Design Development Phase | Twenty Five | percent ( | 25      | %) |
| Construction Documents   | Twenty Five | percent ( | 25      | %) |
| Phase                    | S           | 28 28     | 12. 12. | 8  |
| Procurement Phase        | Five        | percent ( | 5       | %) |
| Construction Phase       | Twenty Five | percent ( | 25      | %) |

See Exhibit C - Hourly Compensation and Reimbursable Expenses

#### PAGE 23

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus Ten\_percent (10\_%) of the expenses incurred.

N/A

§ 11.10.1.1 An initial payment of Zero\_(\$ 0\_) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of <a href="Zero">Zero</a> (\$ 0 ) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred.

Additions and Deletions Report for AIA Document B101" – 2017, Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 16:09:09 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

User Notes:

| performed. Payments are due and payable upon pre                              | services shall be made monthly in proportion to services esentation of the Architect's invoice. Amounts unpaid Forty-Five at at the rate entered below, or in the absence thereof at the legal ace of business of the Architect. |
|---|--|
| •••   |  |
| 1.5 % Monthly   |  |
| PAGE 24   |  |
| § 11.10.2.4 Owner Agrees to receive invoices by emaddress:                    | aail, which will be sent to the following individual and email   |
| ***   |  |
| Full Name: Kelly Christopherson Email Address: kelly.christopherson@k12.sd.us |  |
|   |  |
| None  |  |
|   |  |
| Exhibit A: Project Initial Information  |  |
| <del></del>   |  |
| Exhibit B: JLG Image Release  |  |
|   |  |
| Exhibit C: Schedule of Hourly Compensation & Re                               | imbursable Expenses  |
| PAGE 25   |  |
| Garret Bischoff, President  | Catherine Dekkenga, AIA. Principal   |

# Certification of Document's Authenticity

AIA® Document D401™ - 2003

I, Catherine Dekkenga, AIA, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 16:09:09 ET on 10/05/2022 under Order No. 2114257989 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA<sup>®</sup> Document B101<sup>™</sup> - 2017, Standard Form of Agreement Between Owner and Architect, as published by the AIA in its software, other than those additions and deletions shown in the associated Additions and Deletions Report.

| (Signed) |  |  |  |
|----------|--|--|--|
|          |  |  |  |
|          |  |  |  |
| Title)   |  |  |  |
|          |  |  |  |
|          |  |  |  |
| Dated)   |  |  |  |



# Standard Form of Agreement Between Owner and Architect

AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two (In words, indicate day, month and year.)

BETWEEN the Architect's client identified as the Owner: (Name, legal status, address and other information)

Huron Public School District 150 5th St. SW Huron, SD 57350

and the Architect: (Name, legal status, address and other information)

JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104

for the following Project: (Name, location and detailed description)

21392.01 – Huron School District – BMW Building Controls The project consists of updating control systems in the Buchanan, Madison and Washington buildings.

The Owner and Architect agree as follows.

#### ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

#### **TABLE OF ARTICLES**

- 1 INITIAL INFORMATION
- 2 ARCHITECT'S RESPONSIBILITIES
- 3 SCOPE OF ARCHITECT'S BASIC SERVICES
- 4 SUPPLEMENTAL AND ADDITIONAL SERVICES
- 5 OWNER'S RESPONSIBILITIES
- 6 COST OF THE WORK
- 7 COPYRIGHTS AND LICENSES
- 8 CLAIMS AND DISPUTES
- 9 TERMINATION OR SUSPENSION
- 10 MISCELLANEOUS PROVISIONS
- 11 COMPENSATION
- 12 SPECIAL TERMS AND CONDITIONS
- 13 SCOPE OF THE AGREEMENT

#### ARTICLE 1 INITIAL INFORMATION

§ 1.1 This Agreement is based on the Initial Information set forth in this Section 1.1.

(For each item in this section, insert the information or a statement such as "not applicable" or "unknown at time of execution.")

# § 1.1.1 The Owner's program for the Project:

(Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.)

The project consists of updating control systems in the Buchanan, Madison and Washington buildings.

# § 1.1.2 The Project's physical characteristics:

(Identify or describe pertinent information about the Project's physical characteristics, such as size; location; dimensions; geotechnical reports; site boundaries; topographic surveys; traffic and utility studies; availability of public and private utilities and services; legal description of the site, etc.)

Existing activity building Owned and operated by the School District

§ 1.1.3 The Owner's budget for the Cost of the Work, as defined in Section 6.1: (Provide total and, if known, a line item breakdown.)

\$700,000.00

Init.

- § 1.1.4 The Owner's anticipated design and construction milestone dates:
  - .1 Design phase milestone dates, if any:

Bidding: Late Fall 2022

Construction to commence spring/summer of 2023

.2 Construction commencement date:

N/A

.3 Substantial Completion date or dates:

N/A

.4 Other milestone dates:

N/A

§ 1.1.5 The Owner intends the following procurement and delivery method for the Project: (Identify method such as competitive bid or negotiated contract, as well as any requirements for accelerated or fast-track design and construction, multiple bid packages, or phased construction.)

Design/Bid/Build

§ 1.1.6 The Owner's anticipated Sustainable Objective for the Project: (Identify and describe the Owner's Sustainable Objective for the Project, if any.)

N/A

(Paragraph Deleted)

§ 1.1.6.1 If the Owner identifies a Sustainable Objective, the Owner and Architect shall complete and incorporate AIA Document E204<sup>TM</sup>–2017, Sustainable Projects Exhibit, into this Agreement to define the terms, conditions and services related to the Owner's Sustainable Objective. If E204–2017 is incorporated into this agreement, the Owner and Architect shall incorporate the completed E204–2017 into the agreements with the consultants and contractors performing services or Work in any way associated with the Sustainable Objective.

§ 1.1.7 The Owner identifies the following representative in accordance with Section 5.3: (List name, address, and other contact information.)

Kelly Christopherson Huron Public School District 150 5th St. SW Huron, SD 57350

Telephone:

Email: kelly.christopherson@k12.sd.us

§ 1.1.8 The persons or entities, in addition to the Owner's representative, who are required to review the Architect's submittals to the Owner are as follows:

(List name, address, and other contact information.)

N/A

§ 1.1.9 The Owner shall retain the following consultants and contractors: (List name, legal status, address, and other contact information.)

.1 Geotechnical Engineer:

N/A

Init.

.2 Civil Engineer:

N/A

.3 Other, if any:

(List any other consultants and contractors retained by the Owner.)

N/A

§ 1.1.10 The Architect identifies the following representative in accordance with Section 2.3: (List name, address, and other contact information.)

Catherine Dekkenga JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104

Telephone: 605-271-2531

Email: cdekkenga@jlgarchitects.com

§ 1.1.11 The Architect shall retain the consultants identified in Sections 1.1.11.1 and 1.1.11.2: (List name, legal status, address, and other contact information.)

- § 1.1.11.1 Consultants retained under Basic Services:
  - .1 Structural Engineer:

N/A

.2 Mechanical Engineer:

Associated Consulting Engineers, Inc., Jordan deWit 340 S. Phillips Ave. Sioux Falls, SD 57104 Telephone: 605-335-3720

.3 Electrical Engineer:

Associated Consulting Engineers, Inc., Brad Shoup 340 S. Phillips Ave. Sioux Falls, SD 57104

AIA Document B101" – 2017. Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 12:23:55 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

User Notes:

Telephone: 605-335-3720

§ 1.1.11.2 Consultants retained under Supplemental Services:

N/A

§ 1.1.12 Other Initial Information on which the Agreement is based:

N/A

- § 1.2 The Owner and Architect may rely on the Initial Information. Both parties, however, recognize that the Initial Information may materially change and, in that event, the Owner and the Architect shall appropriately adjust the Architect's services, schedule for the Architect's services, and the Architect's compensation. The Owner shall adjust the Owner's budget for the Cost of the Work and the Owner's anticipated design and construction milestones, as necessary, to accommodate material changes in the Initial Information.
- § 1.3 The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. The parties will use AIA Document E203<sup>TM</sup>–2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.
- § 1.3.1 Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in AIA Document E203<sup>TM</sup>\_2013, Building Information Modeling and Digital Data Exhibit, and the requisite AIA Document G202<sup>TM</sup>\_2013, Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

#### ARTICLE 2 ARCHITECT'S RESPONSIBILITIES

- § 2.1 The Architect shall provide professional services as set forth in this Agreement. The Architect represents that it is properly licensed in the jurisdiction where the Project is located to provide the services required by this Agreement, or shall cause such services to be performed by appropriately licensed design professionals.
- § 2.2 The Architect shall perform its services consistent with the professional skill and care ordinarily provided by architects practicing in the same or similar locality under the same or similar circumstances. The Architect shall perform its services as expeditiously as is consistent with such professional skill and care and the orderly progress of the Project.
- § 2.3 The Architect shall identify a representative authorized to act on behalf of the Architect with respect to the Project.
- § 2.4 Except with the Owner's knowledge and consent, the Architect shall not engage in any activity, or accept any employment, interest or contribution that would reasonably appear to compromise the Architect's professional judgment with respect to this Project.
- § 2.5 The Architect shall maintain the following insurance until termination of this Agreement. If any of the requirements set forth below are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect as set forth in Section 11.9.
- § 2.5.1 Commercial General Liability with policy limits of One Million Dollars and Zero Cents (\$ 1,000,000.00 ) for each occurrence and Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) in the aggregate for bodily injury and property damage.

- § 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of One Million Dollars and Zero Cents (\$ 1,000,000.00 ) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.
- § 2.5.3 The Architect may achieve the required limits and coverage for Commercial General Liability and Automobile Liability through a combination of primary and excess or umbrella liability insurance, provided such primary and excess or umbrella liability insurance policies result in the same or greater coverage as the coverages required under Sections 2.5.1 and 2.5.2, and in no event shall any excess or umbrella liability insurance provide narrower coverage than the primary policy. The excess policy shall not require the exhaustion of the underlying limits only through the actual payment by the underlying insurers.
- § 2.5.4 Workers' Compensation at statutory limits.
- § 2.5.5 Employers' Liability with policy limits Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each accident, Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000.00 ) policy limit.
- § 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) per claim and Four Million Dollars and Zero Cents (\$ 4,000,000.00 ) in the aggregate.
- § 2.5.7 Additional Insured Obligations. To the fullest extent permitted by law, the Architect shall cause the primary and excess or umbrella polices for Commercial General Liability and Automobile Liability to include the Owner as an additional insured for claims caused in whole or in part by the Architect's negligent acts or omissions. The additional insured coverage shall be primary and non-contributory to any of the Owner's insurance policies and shall apply to both ongoing and completed operations.
- § 2.5.8 The Architect shall provide certificates of insurance to the Owner that evidence compliance with the requirements in this Section 2.5.

# ARTICLE 3 SCOPE OF ARCHITECT'S BASIC SERVICES

- § 3.1 The Architect's Basic Services consist of those described in this Article 3 and include structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.
- § 3.1.1 The Architect shall manage the Architect's services, research applicable design criteria, attend Project meetings, communicate with members of the Project team, and report progress to the Owner.
- § 3.1.2 The Architect shall coordinate its services with those services provided by the Owner and the Owner's consultants. The Architect shall be entitled to rely on, and shall not be responsible for, the accuracy, completeness, and timeliness of, services and information furnished by the Owner and the Owner's consultants. The Architect shall provide prompt written notice to the Owner if the Architect becomes aware of any error, omission, or inconsistency in such services or information.
- § 3.1.3 As soon as practicable after the date of this Agreement, the Architect shall submit for the Owner's approval a schedule for the performance of the Architect's services. The schedule initially shall include anticipated dates for the commencement of construction and for Substantial Completion of the Work as set forth in the Initial Information. The schedule shall include allowances for periods of time required for the Owner's review, for the performance of the Owner's consultants, and for approval of submissions by authorities having jurisdiction over the Project. Once approved by the Owner, time limits established by the schedule shall not, except for reasonable cause, be exceeded by the Architect or Owner. With the Owner's approval, the Architect shall adjust the schedule, if necessary, as the Project proceeds until the commencement of construction.
- § 3.1.4 The Architect shall not be responsible for an Owner's directive or substitution, or for the Owner's acceptance of non-conforming Work, made or given without the Architect's written approval.

- § 3.1.5 The Architect shall contact governmental authorities required to approve the Construction Documents and entities providing utility services to the Project. The Architect shall respond to applicable design requirements imposed by those authorities and entities.
- § 3.1.6 The Architect shall assist the Owner in connection with the Owner's responsibility for filing documents required for the approval of governmental authorities having jurisdiction over the Project.

#### § 3.2 Schematic Design Phase Services

- § 3.2.1 The Architect shall review the program and other information furnished by the Owner, and shall review laws, codes, and regulations applicable to the Architect's services.
- § 3.2.2 The Architect shall prepare a preliminary evaluation of the Owner's program, schedule, budget for the Cost of the Work, Project site, the proposed procurement and delivery method, and other Initial Information, each in terms of the other, to ascertain the requirements of the Project. The Architect shall notify the Owner of (1) any inconsistencies discovered in the information, and (2) other information or consulting services that may be reasonably needed for the Project.
- § 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner approach(es) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.
- § 3.2.4 Based on the Project requirements agreed upon with the Owner, the Architect shall prepare and present, for the Owner's approval, a preliminary design illustrating the scale and relationship of the Project components.
- § 3.2.5 Based on the Owner's approval of the preliminary design, the Architect shall prepare Schematic Design Documents for the Owner's approval. The Schematic Design Documents shall consist of drawings and other documents including a site plan, if appropriate, and preliminary building plans, sections and elevations; and may include some combination of study models, perspective sketches, or digital representations. Preliminary selections of major building systems and construction materials shall be noted on the drawings or described in writing.
- § 3.2.5.1 The Architect shall consider sustainable design alternatives, such as material choices and building orientation, together with other considerations based on program and aesthetics, in developing a design that is consistent with the Owner's program, schedule and budget for the Cost of the Work. The Owner may obtain more advanced sustainable design services as a Supplemental Service under Section 4.1.1.
- § 3.2.5.2 The Architect shall consider the value of materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.
- § 3.2.6 The Architect shall submit to the Owner an estimate of the Cost of the Work prepared in accordance with Section 6.3.
- § 3.2.7 The Architect shall submit the Schematic Design Documents to the Owner, and request the Owner's approval.

# § 3.3 Design Development Phase Services

- § 3.3.1 Based on the Owner's approval of the Schematic Design Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Design Development Documents for the Owner's approval. The Design Development Documents shall illustrate and describe the development of the approved Schematic Design Documents and shall consist of drawings and other documents including plans, sections, elevations, typical construction details, and diagrammatic layouts of building systems to fix and describe the size and character of the Project as to architectural, structural, mechanical and electrical systems, and other appropriate elements. The Design Development Documents shall also include outline specifications that identify major materials and systems and establish, in general, their quality levels.
- § 3.3.2 The Architect shall update the estimate of the Cost of the Work prepared in accordance with Section 6.3.

§ 3.3.3 The Architect shall submit the Design Development Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, and request the Owner's approval.

#### § 3.4 Construction Documents Phase Services

- § 3.4.1 Based on the Owner's approval of the Design Development Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Construction Documents for the Owner's approval. The Construction Documents shall illustrate and describe the further development of the approved Design Development Documents and shall consist of Drawings and Specifications setting forth in detail the quality levels and performance criteria of materials and systems and other requirements for the construction of the Work. The Owner and Architect acknowledge that, in order to perform the Work, the Contractor will provide additional information, including Shop Drawings, Product Data, Samples and other similar submittals, which the Architect shall review in accordance with Section 3.6.4.
- § 3.4.2 The Architect shall incorporate the design requirements of governmental authorities having jurisdiction over the Project into the Construction Documents.
- § 3.4.3 During the development of the Construction Documents, the Architect shall assist the Owner in the development and preparation of (1) procurement information that describes the time, place, and conditions of bidding, including bidding or proposal forms; (2) the form of agreement between the Owner and Contractor; and (3) the Conditions of the Contract for Construction (General, Supplementary and other Conditions). The Architect shall also compile a project manual that includes the Conditions of the Contract for Construction and Specifications, and may include bidding requirements and sample forms.
- § 3.4.4 The Architect shall update the estimate for the Cost of the Work prepared in accordance with Section 6.3.
- § 3.4.5 The Architect shall submit the Construction Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, take any action required under Section 6.5, and request the Owner's approval.

#### § 3.5 Procurement Phase Services

# § 3.5.1 General

The Architect shall assist the Owner in establishing a list of prospective contractors. Following the Owner's approval of the Construction Documents, the Architect shall assist the Owner in (1) obtaining either competitive bids or negotiated proposals; (2) confirming responsiveness of bids or proposals; (3) determining the successful bid or proposal, if any; and, (4) awarding and preparing contracts for construction.

#### § 3.5.2 Competitive Bidding

- § 3.5.2.1 Bidding Documents shall consist of bidding requirements and proposed Contract Documents.
- § 3.5.2.2 The Architect shall assist the Owner in bidding the Project by:
  - .1 facilitating the distribution of Bidding Documents to prospective bidders;
  - .2 organizing and conducting a pre-bid conference for prospective bidders;
  - .3 preparing responses to questions from prospective bidders and providing clarifications and interpretations of the Bidding Documents to the prospective bidders in the form of addenda; and,
  - .4 organizing and conducting the opening of the bids, and subsequently documenting and distributing the bidding results, as directed by the Owner.
- § 3.5.2.3 If the Bidding Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective bidders.

#### § 3.5.3 Negotiated Proposals

- § 3.5.3.1 Proposal Documents shall consist of proposal requirements and proposed Contract Documents.
- § 3.5.3.2 The Architect shall assist the Owner in obtaining proposals by:
  - .1 facilitating the distribution of Proposal Documents for distribution to prospective contractors and requesting their return upon completion of the negotiation process;

- .2 organizing and participating in selection interviews with prospective contractors;
- .3 preparing responses to questions from prospective contractors and providing clarifications and interpretations of the Proposal Documents to the prospective contractors in the form of addenda; and,
- .4 participating in negotiations with prospective contractors, and subsequently preparing a summary report of the negotiation results, as directed by the Owner.
- § 3.5.3.3 If the Proposal Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective contractors.

#### § 3.6 Construction Phase Services

#### § 3.6.1 General

- § 3.6.1.1 The Architect shall provide administration of the Contract between the Owner and the Contractor as set forth below and in AIA Document A201™-2017, General Conditions of the Contract for Construction. If the Owner and Contractor modify AIA Document A201-2017, those modifications shall not affect the Architect's services under this Agreement unless the Owner and the Architect amend this Agreement.
- § 3.6.1.2 The Architect shall advise and consult with the Owner during the Construction Phase Services. The Architect shall have authority to act on behalf of the Owner only to the extent provided in this Agreement. The Architect shall not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, nor shall the Architect be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect shall be responsible for the Architect's negligent acts or omissions, but shall not have control over or charge of, and shall not be responsible for, acts or omissions of the Contractor or of any other persons or entities performing portions of the Work.
- § 3.6.1.3 Subject to Section 4.2 and except as provided in Section 3.6.6.5, the Architect's responsibility to provide Construction Phase Services commences with the award of the Contract for Construction and terminates on the date the Architect issues the final Certificate for Payment.

#### § 3.6.2 Evaluations of the Work

- § 3.6.2.1 The Architect shall visit the site at points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.
- § 3.6.2.2 The Architect has the authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect shall have the authority to require inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not the Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, suppliers, their agents or employees, or other persons or entities performing portions of the Work.
- § 3.6.2.3 The Architect shall interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests shall be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- § 3.6.2.4 Interpretations and decisions of the Architect shall be consistent with the intent of, and reasonably inferable from, the Contract Documents and shall be in writing or in the form of drawings. When making such interpretations and decisions, the Architect shall endeavor to secure faithful performance by both Owner and Contractor, shall not show partiality to either, and shall not be liable for results of interpretations or decisions rendered in good faith. The

Architect's decisions on matters relating to aesthetic effect shall be final if consistent with the intent expressed in the Contract Documents.

§ 3.6.2.5 Unless the Owner and Contractor designate another person to serve as an Initial Decision Maker, as that term is defined in AIA Document A201–2017, the Architect shall render initial decisions on Claims between the Owner and Contractor as provided in the Contract Documents.

# § 3.6.3 Certificates for Payment to Contractor

§ 3.6.3.1 The Architect shall review and certify the amounts due the Contractor and shall issue certificates in such amounts. The Architect's certification for payment shall constitute a representation to the Owner, based on the Architect's evaluation of the Work as provided in Section 3.6.2 and on the data comprising the Contractor's Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified. The foregoing representations are subject to (1) an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, (2) results of subsequent tests and inspections, (3) correction of minor deviations from the Contract Documents prior to completion, and (4) specific qualifications expressed by the Architect.

§ 3.6.3.2 The issuance of a Certificate for Payment shall not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) ascertained how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

§ 3.6.3.3 The Architect shall maintain a record of the Applications and Certificates for Payment.

#### § 3.6.4 Submittals

§ 3.6.4.1 The Architect shall review the Contractor's submittal schedule and shall not unreasonably delay or withhold approval of the schedule. The Architect's action in reviewing submittals shall be taken in accordance with the approved submittal schedule or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time, in the Architect's professional judgment, to permit adequate review.

§ 3.6.4.2 The Architect shall review, or take other appropriate action per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

§ 3.6.4.3 If the Contract Documents specifically require the Contractor to provide professional design services or certifications by a design professional related to systems, materials, or equipment, the Architect shall specify the appropriate performance and design criteria that such services must satisfy. The Architect shall review and take appropriate action on Shop Drawings and other submittals related to the Work designed or certified by the Contractor's design professional, provided the submittals bear such professional's seal and signature when submitted to the Architect. The Architect's review shall be for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect shall be entitled to rely upon, and shall not be responsible for, the adequacy and accuracy of the services, certifications, and approvals performed or provided by such design professionals.

§ 3.6.4.4 Subject to Section 4.2, the Architect shall review and respond to requests for information about the Contract Documents. The Architect shall set forth, in the Contract Documents, the requirements for requests for information. Requests for information shall include, at a minimum, a detailed written statement that indicates the specific Drawings or Specifications in need of clarification and the nature of the clarification requested. The Architect's response to such requests shall be made in writing within any time limits agreed upon, or otherwise with reasonable

promptness. If appropriate, the Architect shall prepare and issue supplemental Drawings and Specifications in response to the requests for information.

§ 3.6.4.5 The Architect shall maintain a record of submittals and copies of submittals supplied by the Contractor in accordance with the requirements of the Contract Documents.

#### § 3.6.5 Changes in the Work

§ 3.6.5.1 The Architect may order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. Subject to Section 4.2, the Architect shall prepare Change Orders and Construction Change Directives for the Owner's approval and execution in accordance with the Contract Documents.

§ 3.6.5.2 The Architect shall maintain records relative to changes in the Work.

# § 3.6.6 Project Completion

# § 3.6.6.1 The Architect shall:

- .1 conduct inspections to determine the date or dates of Substantial Completion and the date of final completion;
- .2 issue Certificates of Substantial Completion;
- .3 forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract Documents and received from the Contractor; and,
- .4 issue a final Certificate for Payment based upon a final inspection indicating that, to the best of the Architect's knowledge, information, and belief, the Work complies with the requirements of the Contract Documents.
- § 3.6.6.2 The Architect's inspections shall be conducted with the Owner to check conformance of the Work with the requirements of the Contract Documents and to verify the accuracy and completeness of the list submitted by the Contractor of Work to be completed or corrected.
- § 3.6.6.3 When Substantial Completion has been achieved, the Architect shall inform the Owner about the balance of the Contract Sum remaining to be paid the Contractor, including the amount to be retained from the Contract Sum, if any, for final completion or correction of the Work.
- § 3.6.6.4 The Architect shall forward to the Owner the following information received from the Contractor: (1) consent of surety or sureties, if any, to reduction in or partial release of retainage or the making of final payment; (2) affidavits, receipts, releases and waivers of liens, or bonds indemnifying the Owner against liens; and (3) any other documentation required of the Contractor under the Contract Documents.
- § 3.6.6.5 Upon request of the Owner, and prior to the expiration of one year from the date of Substantial Completion, the Architect shall, without additional compensation, conduct a meeting with the Owner to review the facility operations and performance.

# § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- 1. The Parties agree and understand that, during the course of the construction of the Work, the Owner, Contractor, or other Project participants may choose to, at their own expense and initiative, make use of supplemental video imagery and information sourced from devices such as project webcams, Unmanned Aerial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain visual information and documentation regarding the state of the progress and/or quality of the Work (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly agree and acknowledge the following:
  - Since Supplementary Video Information encompasses only certain limited visual information
    and provides neither complete nor comprehensive visual information/measurements/dimensions,
    nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video
    Information shall not be used as a substitute for or required supplement to Architect's Construction
    Phase services, herein, but rather may, to the extent provided and made available to Architect in a
    reasonably timely, legible, and complete fashion, supplement the other data and information
    obtained by Architect pursuant to Architect's Construction Phase Services, herein.

- Architect shall have no obligation to conduct a comprehensive or continuous review of any Supplementary Video Information, but rather may, at its sole discretion, choose to review such Supplementary Video Information, or any portion thereof, in connection with the performance of its Construction Phase Services on the Project.
- 3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part thereof.

#### ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

# § 4.1 Supplemental Services

§ 4.1.1 The services listed below are not included in Basic Services but may be required for the Project. The Architect shall provide the listed Supplemental Services only if specifically designated in the table below as the Architect's responsibility, and the Owner shall compensate the Architect as provided in Section 11.2. Unless otherwise specifically addressed in this Agreement, if neither the Owner nor the Architect is designated, the parties agree that the listed Supplemental Service is not being provided for the Project.

(Designate the Architect's Supplemental Services and the Owner's Supplemental Services required for the Project by indicating whether the Architect or Owner shall be responsible for providing the identified Supplemental Service. Insert a description of the Supplemental Services in Section 4.1.2 below or attach the description of services as an exhibit to this Agreement.)

| Suppleme   | ntal Services  | Responsibility<br>(Architect, Owner, or not<br>provided) |
|------------|--|--|
| § 4.1.1.1  | Programming  | Not Provided   |
| § 4.1.1.2  | Multiple preliminary designs   | Not Provided   |
| § 4.1.1.3  | Measured drawings  | Not Provided   |
| § 4.1.1.4  | Existing facilities surveys  | Not Provided   |
| § 4.1.1.5  | Site evaluation and planning   | Not Provided   |
| § 4.1.1.6  | Building Information Model management responsibilities               | Not Provided   |
| § 4.1.1.7  | Development of Building Information Models for post construction use | Not Provided   |
| § 4.1.1.8  | Civil engineering  | Not Provided   |
| § 4.1.1.9  | Landscape design   | Not Provided   |
| § 4.1.1.10 | Architectural interior design  | Not Provided   |
| § 4.1.1.11 | Value analysis   | Not Provided   |
| § 4.1.1.12 | Detailed cost estimating beyond that required in Section 6.3         | Not Provided   |
| § 4.1.1.13 | On-site project representation                                       | Not Provided   |
| § 4.1.1.14 | Conformed documents for construction                                 | Not Provided   |
| § 4.1.1.15 | As-designed record drawings  | Not Provided   |
| § 4.1.1.16 | As-constructed record drawings                                       | Not Provided   |
| § 4.1.1.17 | Post-occupancy evaluation  | Not Provided   |
| § 4.1.1.18 | Facility support services  | Not Provided   |
| § 4.1.1.19 | Tenant-related services  | Not Provided   |
| § 4.1.1.20 | Architect's coordination of the Owner's consultants                  | Not Provided   |
| § 4.1.1.21 | Telecommunications/data design                                       | Not Provided   |
| § 4.1.1.22 | Security evaluation and planning                                     | Not Provided   |

| § 4.1.1.25 | Fast-track design services                       | Not Provided |
|------------|--|--------------|
| § 4.1.1.26 | Multiple bid packages                            | Not Provided |
| § 4.1.1.27 | Historic preservation                            | Not Provided |
| § 4.1.1.28 | Furniture, furnishings, and equipment design     | Not Provided |
| § 4.1.1.29 | Other services provided by specialty Consultants | Not Provided |
| § 4.1.1.30 | Other Supplemental Services                      | Not Provided |

#### § 4.1.2 Description of Supplemental Services

§ 4.1.2.1 A description of each Supplemental Service identified in Section 4.1.1 as the Architect's responsibility is provided below.

(Describe in detail the Architect's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit. The AIA publishes a number of Standard Form of Architect's Services documents that can be included as an exhibit to describe the Architect's Supplemental Services.)

N/A

§ 4.1.2.2 A description of each Supplemental Service identified in Section 4.1.1 as the Owner's responsibility is provided below.

(Describe in detail the Owner's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit.)

N/A

§ 4.1.3 If the Owner identified a Sustainable Objective in Article 1, the Architect shall provide, as a Supplemental Service, the Sustainability Services required in AIA Document E204<sup>TM</sup>–2017, Sustainable Projects Exhibit, attached to this Agreement. The Owner shall compensate the Architect as provided in Section 11.2.

#### § 4.2 Architect's Additional Services

The Architect may provide Additional Services after execution of this Agreement without invalidating the Agreement. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 4.2 shall entitle the Architect to compensation pursuant to Section 11.3 and an appropriate adjustment in the Architect's schedule.

- § 4.2.1 Upon recognizing the need to perform the following Additional Services, the Architect shall notify the Owner with reasonable promptness and explain the facts and circumstances giving rise to the need. The Architect shall not proceed to provide the following Additional Services until the Architect receives the Owner's written authorization:
  - .1 Services necessitated by a change in the Initial Information, previous instructions or approvals given by the Owner, or a material change in the Project including size, quality, complexity, the Owner's schedule or budget for Cost of the Work, or procurement or delivery method;
  - .2 Services necessitated by the enactment or revision of codes, laws, or regulations, including changing or editing previously prepared Instruments of Service;
  - .3 Changing or editing previously prepared Instruments of Service necessitated by official interpretations of applicable codes, laws or regulations that are either (a) contrary to specific interpretations by the applicable authorities having jurisdiction made prior to the issuance of the building permit, or (b) contrary to requirements of the Instruments of Service when those Instruments of Service were prepared in accordance with the applicable standard of care;
  - .4 Services necessitated by decisions of the Owner not rendered in a timely manner or any other failure of performance on the part of the Owner or the Owner's consultants or contractors;
  - .5 Preparing digital models or other design documentation for transmission to the Owner's consultants and contractors, or to other Owner-authorized recipients;
  - .6 Preparation of design and documentation for alternate bid or proposal requests proposed by the Owner;
  - .7 Preparation for, and attendance at, a public presentation, meeting or hearing;

- .8 Preparation for, and attendance at, a dispute resolution proceeding or legal proceeding, except where the Architect is party thereto;
- .9 Evaluation of the qualifications of entities providing bids or proposals;
- .10 Consultation concerning replacement of Work resulting from fire or other cause during construction; or.
- .11 Assistance to the Initial Decision Maker, if other than the Architect.
- § 4.2.2 To avoid delay in the Construction Phase, the Architect shall provide the following Additional Services, notify the Owner with reasonable promptness, and explain the facts and circumstances giving rise to the need. If, upon receipt of the Architect's notice, the Owner determines that all or parts of the services are not required, the Owner shall give prompt written notice to the Architect of the Owner's determination. The Owner shall compensate the Architect for the services provided prior to the Architect's receipt of the Owner's notice.
  - .1 Reviewing a Contractor's submittal out of sequence from the submittal schedule approved by the Architect;
  - .2 Responding to the Contractor's requests for information that are not prepared in accordance with the Contract Documents or where such information is available to the Contractor from a careful study and comparison of the Contract Documents, field conditions, other Owner-provided information, Contractor-prepared coordination drawings, or prior Project correspondence or documentation;
  - .3 Preparing Change Orders and Construction Change Directives that require evaluation of Contractor's proposals and supporting data, or the preparation or revision of Instruments of Service;
  - .4 Evaluating an extensive number of Claims as the Initial Decision Maker; or,
  - .5 Evaluating substitutions proposed by the Owner or Contractor and making subsequent revisions to Instruments of Service resulting therefrom.
- § 4.2.3 The Architect shall provide Construction Phase Services exceeding the limits set forth below as Additional Services. When the limits below are reached, the Architect shall notify the Owner:
  - .1 Two (2) reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor
  - .2 Bi-weekly visits to the site by the Architect during construction
  - .3 One (1) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents
  - .4 One (1) inspections for any portion of the Work to determine final completion.
- § 4.2.4 Except for services required under Section 3.6.6.5 and those services that do not exceed the limits set forth in Section 4.2.3, Construction Phase Services provided more than 60 days after (1) the date of Substantial Completion of the Work or (2) the initial date of Substantial Completion identified in the agreement between the Owner and Contractor, whichever is earlier, shall be compensated as Additional Services to the extent the Architect incurs additional cost in providing those Construction Phase Services.
- § 4.2.5 If the services covered by this Agreement have not been completed within Twenty Four (24) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

# ARTICLE 5 OWNER'S RESPONSIBILITIES

- § 5.1 Unless otherwise provided for under this Agreement, the Owner shall provide information in a timely manner regarding requirements for and limitations on the Project, including a written program, which shall set forth the Owner's objectives; schedule; constraints and criteria, including space requirements and relationships; flexibility; expandability; special equipment; systems; and site requirements.
- § 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes; (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the

Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.

- § 5.3 The Owner shall identify a representative authorized to act on the Owner's behalf with respect to the Project. The Owner shall render decisions and approve the Architect's submittals in a timely manner in order to avoid unreasonable delay in the orderly and sequential progress of the Architect's services.
- § 5.4 The Owner shall furnish surveys to describe physical characteristics, legal limitations and utility locations for the site of the Project, and a written legal description of the site. The surveys and legal information shall include, as applicable, grades and lines of streets, alleys, pavements and adjoining property and structures; designated wetlands; adjacent drainage; rights-of-way, restrictions, casements, encroachments, zoning, deed restrictions, boundaries and contours of the site; locations, dimensions, and other necessary data with respect to existing buildings, other improvements and trees; and information concerning available utility services and lines, both public and private, above and below grade, including inverts and depths. All the information on the survey shall be referenced to a Project benchmark.
- § 5.5 The Owner shall furnish services of geotechnical engineers, which may include test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.
- § 5.6 The Owner shall provide the Supplemental Services designated as the Owner's responsibility in Section 4.1.1.
- § 5.7 If the Owner identified a Sustainable Objective in Article 1, the Owner shall fulfill its responsibilities as required in AIA Document E204™–2017, Sustainable Projects Exhibit, attached to this Agreement.
- § 5.8 The Owner shall coordinate the services of its own consultants with those services provided by the Architect. Upon the Architect's request, the Owner shall furnish copies of the scope of services in the contracts between the Owner and the Owner's consultants. The Owner shall furnish the services of consultants other than those designated as the responsibility of the Architect in this Agreement, or authorize the Architect to furnish them as an Additional Service, when the Architect requests such services and demonstrates that they are reasonably required by the scope of the Project. The Owner shall require that its consultants and contractors maintain insurance, including professional liability insurance, as appropriate to the services or work provided.
- § 5.9 The Owner shall furnish tests, inspections and reports required by law or the Contract Documents, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials.
- § 5.10 The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.
- § 5.11 The Owner shall provide prompt written notice to the Architect if the Owner becomes aware of any fault or defect in the Project, including errors, omissions or inconsistencies in the Architect's Instruments of Service.
- § 5.12 The Owner shall include the Architect in all communications with the Contractor that relate to or affect the Architect's services or professional responsibilities. The Owner shall promptly notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect.
- § 5.13 Before executing the Contract for Construction, the Owner shall coordinate the Architect's duties and responsibilities set forth in the Contract for Construction with the Architect's services set forth in this Agreement. The Owner shall provide the Architect a copy of the executed agreement between the Owner and Contractor, including the General Conditions of the Contract for Construction.
- § 5.14 The Owner shall provide the Architect access to the Project site prior to commencement of the Work and shall obligate the Contractor to provide the Architect access to the Work wherever it is in preparation or progress.

§ 5.15 Within 15 days after receipt of a written request from the Architect, the Owner shall furnish the requested information as necessary and relevant for the Architect to evaluate, give notice of, or enforce lien rights.

#### ARTICLE 6 COST OF THE WORK

- § 6.1 For purposes of this Agreement, the Cost of the Work shall be the total cost to the Owner to construct all elements of the Project designed or specified by the Architect and shall include contractors' general conditions costs, overhead and profit. The Cost of the Work also includes the reasonable value of labor, materials, and equipment, donated to, or otherwise furnished by, the Owner. The Cost of the Work does not include the compensation of the Architect; the costs of the land, rights-of-way, financing, or contingencies for changes in the Work; or other costs that are the responsibility of the Owner.
- § 6.2 The Owner's budget for the Cost of the Work is provided in Initial Information, and shall be adjusted throughout the Project as required under Sections 5.2, 6.4 and 6.5. Evaluations of the Owner's budget for the Cost of the Work, and the preliminary estimate of the Cost of the Work and updated estimates of the Cost of the Work, prepared by the Architect, represent the Architect's judgment as a design professional. It is recognized, however, that neither the Architect nor the Owner has control over the cost of labor, materials, or equipment; the Contractor's methods of determining bid prices; or competitive bidding, market, or negotiating conditions. Accordingly, the Architect cannot and does not warrant or represent that bids or negotiated prices will not vary from the Owner's budget for the Cost of the Work, or from any estimate of the Cost of the Work, or evaluation, prepared or agreed to by the Architect.
- § 6.3 In preparing estimates of the Cost of Work, the Architect shall be permitted to include contingencies for design, bidding, and price escalation; to determine what materials, equipment, component systems, and types of construction are to be included in the Contract Documents; to recommend reasonable adjustments in the program and scope of the Project; and to include design alternates as may be necessary to adjust the estimated Cost of the Work to meet the Owner's budget. The Architect's estimate of the Cost of the Work shall be based on current area, volume or similar conceptual estimating techniques. If the Owner requires a detailed estimate of the Cost of the Work, the Architect shall provide such an estimate, if identified as the Architect's responsibility in Section 4.1.1, as a Supplemental Service.
- § 6.4 If, through no fault of the Architect, the Procurement Phase has not commenced within 90 days after the Architect submits the Construction Documents to the Owner, the Owner's budget for the Cost of the Work shall be adjusted to reflect changes in the general level of prices in the applicable construction market.
- § 6.5 If at any time the Architect's estimate of the Cost of the Work exceeds the Owner's budget for the Cost of the Work, the Architect shall make appropriate recommendations to the Owner to adjust the Project's size, quality, or budget for the Cost of the Work, and the Owner shall cooperate with the Architect in making such adjustments.
- § 6.6 If the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services is exceeded by the lowest bona fide bid or negotiated proposal, the Owner shall
  - .1 give written approval of an increase in the budget for the Cost of the Work;
  - .2 authorize rebidding or renegotiating of the Project within a reasonable time;
  - .3 terminate in accordance with Section 9.5;
  - .4 in consultation with the Architect, revise the Project program, scope, or quality as required to reduce the Cost of the Work; or,
  - .5 implement any other mutually acceptable alternative.
- § 6.7 If the Owner chooses to proceed under Section 6.6.4, the Architect shall modify the Construction Documents as necessary to comply with the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services, or the budget as adjusted under Section 6.6.1. If the Owner requires the Architect to modify the Construction Documents because the lowest bona fide bid or negotiated proposal exceeds the Owner's budget for the Cost of the Work due to market conditions the Architect could not reasonably anticipate, the Owner shall compensate the Architect for the modifications as an Additional Service pursuant to Section 11.3; otherwise the Architect's services for modifying the Construction Documents shall be without additional compensation. In any event, the Architect's modification of the Construction Documents shall be the limit of the Architect's responsibility under this Article 6.

#### ARTICLE 7 COPYRIGHTS AND LICENSES

- § 7.1 The Architect and the Owner warrant that in transmitting Instruments of Service, or any other information, the transmitting party is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project.
- § 7.2 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements or for similar purposes in connection with the Project is not to be construed as publication in derogation of the reserved rights of the Architect and the Architect's consultants.
- § 7.3 The Architect grants to the Owner a nonexclusive license to use the Architect's Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations under this Agreement, including prompt payment of all sums due pursuant to Article 9 and Article 11. The Architect shall obtain similar nonexclusive licenses from the Architect's consultants consistent with this Agreement. The license granted under this section permits the Owner to authorize the Contractor, Subcontractors, Sub-subcontractors, and suppliers, as well as the Owner's consultants and separate contractors, to reproduce applicable portions of the Instruments of Service, subject to any protocols established pursuant to Section 1.3, solely and exclusively for use in performing services or construction for the Project. If the Architect rightfully terminates this Agreement for cause as provided in Section 9.4, the license granted in this Section 7.3 shall terminate.
- § 7.3.1 In the event the Owner uses the Instruments of Service without retaining the authors of the Instruments of Service, the Owner releases the Architect and Architect's consultant(s) from all claims and causes of action arising from such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the Architect and its consultants from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's use of the Instruments of Service under this Section 7.3.1. The terms of this Section 7.3.1 shall not apply if the Owner rightfully terminates this Agreement for cause under Section 9.4.
- § 7.4 Except for the licenses granted in this Article 7, no other license or right shall be deemed granted or implied under this Agreement. The Owner shall not assign, delegate, sublicense, pledge or otherwise transfer any license granted herein to another party without the prior written agreement of the Architect. Any unauthorized use of the Instruments of Service shall be at the Owner's sole risk and without liability to the Architect and the Architect's consultants.
- § 7.5 Except as otherwise stated in Section 7.3, the provisions of this Article 7 shall survive the termination of this Agreement.

# ARTICLE 8 CLAIMS AND DISPUTES

#### § 8.1 General

- § 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.
- § 8.1.2 To the extent damages are covered by property insurance, the Owner and Architect waive all rights against each other and against the contractors, consultants, agents, and employees of the other for damages, except such rights as they may have to the proceeds of such insurance as set forth in AIA Document A201–2017, General

Conditions of the Contract for Construction. The Owner or the Architect, as appropriate, shall require of the contractors, consultants, agents, and employees of any of them, similar waivers in favor of the other parties enumerated herein.

§ 8.1.3 The Architect and Owner waive consequential damages for claims, disputes, or other matters in question, arising out of or relating to this Agreement. This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination of this Agreement, except as specifically provided in Section 9.7.

#### § 8.2 Mediation

- § 8.2.1 Any claim, dispute or other matter in question arising out of or related to this Agreement shall be subject to mediation as a condition precedent to binding dispute resolution. If such matter relates to or is the subject of a lien arising out of the Architect's services, the Architect may proceed in accordance with applicable law to comply with the lien notice or filing deadlines prior to resolution of the matter by mediation or by binding dispute resolution.
- § 8.2.2 The Owner and Architect shall endeavor to resolve claims, disputes and other matters in question between them by mediation, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of this Agreement. A request for mediation shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of a complaint or other appropriate demand for binding dispute resolution but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration proceeding is stayed pursuant to this section, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.
- § 8.2.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

| dis | pul | 4 If the parties do not resolve a dispute through mediation pursuant to this Section 8.2, the method of binding the resolution shall be the following: |
|-----|-----|--|
| I   | ]   | Arbitration pursuant to Section 8.3 of this Agreement  |
| [ X | 1   | Litigation in a court of competent jurisdiction  |

[ ] Other: (Specify)

If the Owner and Architect do not select a method of binding dispute resolution, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, the dispute will be resolved in a court of competent jurisdiction.

#### § 8.3 Arbitration

- § 8.3.1 If the parties have selected arbitration as the method for binding dispute resolution in this Agreement, any claim, dispute or other matter in question arising out of or related to this Agreement subject to, but not resolved by, mediation shall be subject to arbitration, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of this Agreement. A demand for arbitration shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the arbitration.
- § 8.3.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the claim, dispute or other matter in question would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the

arbitration shall constitute the institution of legal or equitable proceedings based on the claim, dispute or other matter in question.

- § 8.3.2 The foregoing agreement to arbitrate, and other agreements to arbitrate with an additional person or entity duly consented to by parties to this Agreement, shall be specifically enforceable in accordance with applicable law in any court having jurisdiction thereof.
- § 8.3.3 The award rendered by the arbitrator(s) shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

#### § 8.3.4 Consolidation or Joinder

- § 8.3.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation; (2) the arbitrations to be consolidated substantially involve common questions of law or fact; and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).
- § 8.3.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.
- § 8.3.4.3 The Owner and Architect grant to any person or entity made a party to an arbitration conducted under this Section 8.3, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Architect under this Agreement.
- § 8.4 The provisions of this Article 8 shall survive the termination of this Agreement.

# ARTICLE 9 TERMINATION OR SUSPENSION

- § 9.1 If the Owner fails to make payments to the Architect in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Architect's option, cause for suspension of performance of services under this Agreement. If the Architect elects to suspend services, the Architect shall give seven days' written notice to the Owner before suspending services. In the event of a suspension of services, the Architect shall have no liability to the Owner for delay or damage caused the Owner because of such suspension of services. Before resuming services, the Owner shall pay the Architect all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.3 If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Architect, the Architect may terminate this Agreement by giving not less than seven days' written notice.
- § 9.4 Either party may terminate this Agreement upon seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.
- § 9.5 The Owner may terminate this Agreement upon seven days' written notice to the Architect for the Owner's convenience and without cause.
- § 9.6 If the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall compensate the Architect for services performed prior to

termination, Reimbursable Expenses incurred, and costs attributable to termination, including the costs attributable to the Architect's termination of consultant agreements.

§ 9.7 In addition to any amounts paid under Section 9.6, if the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall pay to the Architect the following fees:

(Set forth below the amount of any termination or licensing fee, or the method for determining any termination or licensing fee.)

.1 Termination Fee:

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

.2 Licensing Fee if the Owner intends to continue using the Architect's Instruments of Service:

As mutually agreed

- § 9.8 Except as otherwise expressly provided herein, this Agreement shall terminate one year from the date of Substantial Completion.
- § 9.9 The Owner's rights to use the Architect's Instruments of Service in the event of a termination of this Agreement are set forth in Article 7 and Section 9.7.

#### ARTICLE 10 MISCELLANEOUS PROVISIONS

- § 10.1 This Agreement shall be governed by the law of the place where the Project is located, excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 8.3.
- § 10.2 Terms in this Agreement shall have the same meaning as those in AIA Document A201–2017, General Conditions of the Contract for Construction.
- § 10.3 The Owner and Architect, respectively, bind themselves, their agents, successors, assigns, and legal representatives to this Agreement. Neither the Owner nor the Architect shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement, including any payments due to the Architect by the Owner prior to the assignment.
- § 10.4 If the Owner requests the Architect to execute certificates, the proposed language of such certificates shall be submitted to the Architect for review at least 14 days prior to the requested dates of execution. If the Owner requests the Architect to execute consents reasonably required to facilitate assignment to a lender, the Architect shall execute all such consents that are consistent with this Agreement, provided the proposed consent is submitted to the Architect for review at least 14 days prior to execution. The Architect shall not be required to execute certificates or consents that would require knowledge, services, or responsibilities beyond the scope of this Agreement.
- § 10.5 Nothing contained in this Agreement shall create a contractual relationship with, or a cause of action in favor of, a third party against either the Owner or Architect.
- § 10.6 Unless otherwise required in this Agreement, the Architect shall have no responsibility for the discovery, presence, handling, removal or disposal of, or exposure of persons to, hazardous materials or toxic substances in any form at the Project site.
- § 10.7 The Architect shall have the right to include photographic or artistic representations of the design of the Project among the Architect's promotional and professional materials. The Architect shall be given reasonable access to the completed Project to make such representations. However, the Architect's materials shall not include the Owner's confidential or proprietary information if the Owner has previously advised the Architect in writing of the specific information considered by the Owner to be confidential or proprietary. The Owner shall provide professional credit for the Architect in the Owner's promotional materials for the Project. This Section 10.7 shall

survive the termination of this Agreement unless the Owner terminates this Agreement for cause pursuant to Section 9.4.

§ 10.8 If the Architect or Owner receives information specifically designated as "confidential" or "business proprietary," the receiving party shall keep such information strictly confidential and shall not disclose it to any other person except as set forth in Section 10.8.1. This Section 10.8 shall survive the termination of this Agreement.

§ 10.8.1 The receiving party may disclose "confidential" or "business proprietary" information after 7 days' notice to the other party, when required by law, arbitrator's order, or court order, including a subpoena or other form of compulsory legal process issued by a court or governmental entity, or to the extent such information is reasonably necessary for the receiving party to defend itself in any dispute. The receiving party may also disclose such information to its employees, consultants, or contractors in order to perform services or work solely and exclusively for the Project, provided those employees, consultants and contractors are subject to the restrictions on the disclosure and use of such information as set forth in this Section 10.8.

§ 10.9 The invalidity of any provision of the Agreement shall not invalidate the Agreement or its remaining provisions. If it is determined that any provision of the Agreement violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Agreement shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Agreement.

# ARTICLE 11 COMPENSATION

§ 11.1 For the Architect's Basic Services described under Article 3, the Owner shall compensate the Architect as follows:

.1 Stipulated Sum (Insert amount)

Compensation in the amount of Forty Nine Thousand Five Hundred Dollars and Zero Cents (\$49,500.00)

.2 Percentage Basis (Insert percentage value)

( ) % of the Owner's budget for the Cost of the Work, as calculated in accordance with Section 11.6.

Other (Describe the method of compensation)

§ 11.2 For the Architect's Supplemental Services designated in Section 4.1.1 and for any Sustainability Services required pursuant to Section 4.1.3, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation. If necessary, list specific services to which particular methods of compensation apply.)

Hourly or as mutually agreed upon in advance of services

§ 11.3 For Additional Services that may arise during the course of the Project, including those under Section 4.2, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation.)

Hourly or as mutually agreed upon in advance of services

Init.

§ 11.4 Compensation for Supplemental and Additional Services of the Architect's consultants when not included in Section 11.2 or 11.3, shall be the amount invoiced to the Architect plus percent ( %), or as follows: (Insert amount of, or basis for computing, Architect's consultants' compensation for Supplemental or Additional Services.)

As mutually agreed by Owner and Architect in advance of services

§ 11.5 When compensation for Basic Services is based on a stipulated sum or a percentage basis, the proportion of compensation for each phase of services shall be as follows:

| Total Basic Compensation | one hundred | percent ( | 100 | %)    |
|--------------------------|-------------|-----------|-----|-------|
| Construction Phase       | Twenty Five | percent ( | 25  | %)    |
| Procurement Phase        | Five        | percent ( | 5   | %)    |
| Phase                    |             | 100       |     | 0.000 |
| Construction Documents   | Twenty Five | percent ( | 25  | %)    |
| Design Development Phase | Twenty Five | percent ( | 25  | %)    |
| Schematic Design Phase   | Twenty      | percent ( | 20  | %)    |

- § 11.6 When compensation identified in Section 11.1 is on a percentage basis, progress payments for each phase of Basic Services shall be calculated by multiplying the percentages identified in this Article by the Owner's most recent budget for the Cost of the Work. Compensation paid in previous progress payments shall not be adjusted based on subsequent updates to the Owner's budget for the Cost of the Work.
- § 11.6.1 When compensation is on a percentage basis and any portions of the Project are deleted or otherwise not constructed, compensation for those portions of the Project shall be payable to the extent services are performed on those portions. The Architect shall be entitled to compensation in accordance with this Agreement for all services performed whether or not the Construction Phase is commenced.
- § 11.7 The hourly billing rates for services of the Architect and the Architect's consultants are set forth below. The rates shall be adjusted in accordance with the Architect's and Architect's consultants' normal review practices. (If applicable, attach an exhibit of hourly billing rates or insert them below.)

See Exhibit C - Hourly Compensation and Reimbursable Expenses

| Employee or Category | Rate (\$0.00) |  |
|----------------------|---------------|--|
| N/A                  | N/A           |  |

# § 11.8 Compensation for Reimbursable Expenses

- § 11.8.1 Reimbursable Expenses are in addition to compensation for Basic, Supplemental, and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:
  - Transportation and authorized out-of-town travel and subsistence;
  - .2 Long distance services, dedicated data and communication services, teleconferences, Project web sites, and extranets;
  - .3 Permitting and other fees required by authorities having jurisdiction over the Project;
  - .4 Printing, reproductions, plots, and standard form documents;
  - .5 Postage, handling, and delivery;
  - .6 Expense of overtime work requiring higher than regular rates, if authorized in advance by the Owner;
  - .7 Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner or required for the Project;
  - .8 If required by the Owner, and with the Owner's prior written approval, the Architect's consultants' expenses of professional liability insurance dedicated exclusively to this Project, or the expense of additional insurance coverage or limits in excess of that normally maintained by the Architect's consultants;
  - .9 All taxes levied on professional services and on reimbursable expenses;
  - .10 Site office expenses;
  - .11 Registration fees and any other fees charged by the Certifying Authority or by other entities as necessary to achieve the Sustainable Objective; and,
  - .12 Other similar Project-related expenditures.

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus Ten percent (10 %) of the expenses incurred.

§ 11.9 Architect's Insurance. If the types and limits of coverage required in Section 2.5 are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect for the additional costs incurred by the Architect for the additional coverages as set forth below:

(Insert the additional coverages the Architect is required to obtain in order to satisfy the requirements set forth in Section 2.5, and for which the Owner shall reimburse the Architect.)

N/A

# § 11.10 Payments to the Architect

#### § 11.10.1 Initial Payments

§ 11.10.1.1 An initial payment of Zero (\$ 0 ) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of Zero (\$ 0 ) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred.

# § 11.10.2 Progress Payments

§ 11.10.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid Forty-Five (45) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect. (Insert rate of monthly or annual interest agreed upon.)

#### 1.5 % Monthly

§ 11.10.2.2 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work, unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.

§ 11.10.2.3 Records of Reimbursable Expenses, expenses pertaining to Supplemental and Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

§ 11.10.2.4 Owner Agrees to receive invoices by email, which will be sent to the following individual and email address:

Full Name: Kelly Christopherson

Email Address: kelly.christopherson@k12.sd.us

# ARTICLE 12 SPECIAL TERMS AND CONDITIONS

Special terms and conditions that modify this Agreement are as follows: (Include other terms and conditions applicable to this Agreement.)

None

# ARTICLE 13 SCOPE OF THE AGREEMENT

§ 13.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both the Owner and Architect.

§ 13.2 This Agreement is comprised of the following documents identified below:

- .1 AIA Document B101TM-2017, Standard Form Agreement Between Owner and Architect
- .2 AIA Document E203TM-2013, Building Information Modeling and Digital Data Exhibit, dated as indicated below:

(Insert the date of the E203-2013 incorporated into this agreement.)

| OWNER (Signature)   | ARCHITECT (Signature)   |
|---|---|
| This Agreement entered into as of the day and yea   | nr first written above.   |
| .4 Other documents:<br>(List other documents, if any, forming p   | art of the Agreement.)  |
| Exhibit A: Project Initial Information  Exhibit B: JLG Image Release  Exhibit C: Schedule of Hourly Compensation & F                      | Reimbursable Expenses   |
| Other Exhibits incorporated into this Agre<br>(Clearly identify any other exhibits incorporated identified as exhibits in Section 4.1.2.) | eement:<br>into this Agreement, including any exhibits and scopes of services |
| [ ] AIA Document E204TM_2017, Sustainable (Insert the date of the E204-2017 incorporated int  | Projects Exhibit, dated as indicated below:<br>to this agreement.)            |
| .3 Exhibits: (Check the appropriate box for an  | y exhibits incorporated into this Agreement.)                                 |
|   | •   |

Catherine Dekkenga, AIA, Principal

(Printed name, title, and license number, if required)

Garret Bischoff, President

(Printed name and title)

# Additions and Deletions Report for

AIA® Document B101 - 2017

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 12:23:55 ET on 10/05/2022.

#### PAGE 1

AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two

Huron Public School District 150 5th St. SW Huron, SD 57350

JLG Architects
232 S. Main Ave.
Sioux Falls, SD 57104

21392.01 – Huron School District – BMW Building Controls

The project consists of updating control systems in the Buchanan, Madison and Washington buildings.

#### PAGE 2

#### TABLE OF ARTICLES

(Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.)

The project consists of updating control systems in the Buchanan, Madison and Washington buildings,

Existing activity building Owned and operated by the School District

# \$700,000.00 PAGE 3 Bidding: Late Fall 2022 Construction to commence spring/summer of 2023 N/A ... N/A ... N/A Design/Bid/Build N/A Kelly Christopherson Huron Public School District 150 5th St. SW Huron, SD 57350 Telephone: Email: kelly.christopherson@k12.sd.us N/A

PAGE 4

N/A

N/A

\*\*\*

N/A

Catherine Dekkenga JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104 Telephone: 605-271-2531 Email: cdekkenga@jlgarchitects.com N/A \*\*\* Associated Consulting Engineers, Inc., Jordan deWit 340 S. Phillips Ave. Sioux Falls, SD 57104 Telephone: 605-335-3720 -Associated Consulting Engineers, Inc., Brad Shoup 340 S. Phillips Ave. Sioux Falls, SD 57104 PAGE 5 Telephone: 605-335-3720 -N/A

N/A

§ 2.5.1 Commercial General Liability with policy limits of not less than (\$\scrt{One Million Dollars and Zero}\$\) Cents (\$\\$1,000,000.00\$) for each occurrence and \(\frac{\text{Two Million Dollars and Zero Cents}}{\text{Cents}}\) (\$\\$2,000,000.00\$) in the aggregate for bodily injury and property damage.

#### PAGE 6

§ 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of not less than (\$-One Million Dollars and Zero Cents (\$1,000,000.00) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.

§ 2.5.5 Employers' Liability with policy limits not less than (\$ ) each accident, (\$ Five Hundred Thousand Dollars and Zero Cents (\$ 500,000,00 ) each accident, Five Hundred Thousand Dollars and Zero Cents (\$ 500,000,00 ) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000,00 ) policy limit.

§ 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of not less than (\$\frac{1}{2}\text{Two Million Dollars and Zero Cents} (\$\frac{2}{2}\text{000}\text{000}\text{00}) per claim and Four Million Dollars and Zero Cents} (\$\frac{4}{2}\text{000}\text{000}\text{000}\text{00}) in the aggregate.

§ 3.1 The Architect's Basic Services consist of those described in this Article 3 and include usual and eustomary structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.

## PAGE 7

§ 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner alternative approaches approach(es) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.

§ 3.2.5.2 The Architect shall consider the value of alternative materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.

## PAGE 9

§ 3.6.2.1 The Architect shall visit the site at intervals appropriate to the stage-points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.

#### PAGE 10

§ 3.6.4.2 The Architect shall review and approve, review, or take other appropriate action upon, per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

#### PAGE 11

#### § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- The Parties agree and understand that, during the course of the construction of the Work, the Owner,
   Contractor, or other Project participants may choose to, at their own expense and initiative, make use of
   supplemental video imagery and information sourced from devices such as project webcams, Unmanned
   Acrial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain
   visual information and documentation regarding the state of the progress and/or quality of the Work
   (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly
   agree and acknowledge the following:
  - Since Supplementary Video Information encompasses only certain limited visual information
    and provides neither complete nor comprehensive visual information/measurements/dimensions,
    nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video
    Information shall not be used as a substitute for or required supplement to Architect's Construction
    Phase services, herein, but rather may, to the extent provided and made available to Architect in a
    reasonably timely, legible, and complete fashion, supplement the other data and information
    obtained by Architect pursuant to Architect's Construction Phase Services, herein.
  - Architect shall have no obligation to conduct a comprehensive or continuous review of any Supplementary Video Information, but rather may, at its sole discretion, choose to review such Supplementary Video Information, or any portion thereof, in connection with the performance of its Construction Phase Services on the Project.

#### PAGE 12

3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part

ARTICLE thereof.

...

## ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

## PAGE 13

§ 4.1.1.1 Programming Not Provided

| § 4.1.1.2  | Multiple preliminary designs   | Not Provided |
|------------|--|--------------|
| § 4.1.1.3  | Measured drawings  | Not Provided |
| § 4.1.1.4  | Existing facilities surveys  | Not Provided |
| § 4.1.1.5  | Site evaluation and planning   | Not Provided |
| § 4.1.1.6  | Building Information Model management responsibilities               | Not Provided |
| § 4.1.1.7  | Development of Building Information Models for post construction use | Not Provided |
| § 4.1.1.8  | Civil engineering  | Not Provided |
| § 4.1.1.9  | Landscape design   | Not Provided |
| § 4.1.1.10 | Architectural interior design  | Not Provided |
| § 4.1.1.11 | Value analysis   | Not Provided |
| § 4.1.1.12 | Detailed cost estimating beyond that required in Section 6.3         | Not Provided |
| § 4.1.1.13 | On-site project representation                                       | Not Provided |
| § 4.1.1.14 | Conformed documents for construction                                 | Not Provided |
| § 4.1.1.15 | As-designed record drawings  | Not Provided |
| § 4.1.1.16 | As-constructed record drawings                                       | Not Provided |
| § 4.1.1.17 | Post-occupancy evaluation  | Not Provided |
| § 4.1.1.18 | Facility support services  | Not Provided |
| § 4.1.1.19 | Tenant-related services  | Not Provided |
| § 4.1.1.20 | Architect's coordination of the Owner's consultants                  | Not Provided |
| § 4.1.1.21 | Telecommunications/data design                                       | Not Provided |
| § 4.1.1.22 | Security evaluation and planning                                     | Not Provided |
| § 4.1.1.23 | Commissioning  | Not Provided |
| § 4.1.1.24 | Sustainable Project Services pursuant to Section 4.1.3               | Not Provided |
| § 4.1.1.25 | Fast-track design services   | Not Provided |
| § 4.1.1.26 | Multiple bid packages  | Not Provided |
| § 4.1.1.27 | Historic preservation  | Not Provided |
| § 4.1.1.28 | Furniture, furnishings, and equipment design                         | Not Provided |
| § 4.1.1.29 | Other services provided by specialty Consultants                     | Not Provided |
| § 4.1.1.30 | Other Supplemental Services  | Not Provided |

N/A

N/A

## PAGE 14

.1 Two (2) reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor

.2 ()-Bi-weekly visits to the site by the Architect during construction

Additions and Deletions Report for AIA Document B101" – 2017. Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects, All rights reserved. The "American Institute of Architects," AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 12:23:55 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service, To report copyright violations, e-mail copyright@aia.org.

(389ADA44)

- One (1 ) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents
  - .4 One (1 ) inspections for any portion of the Work to determine final completion.

§ 4.2.5 If the services covered by this Agreement have not been completed within <a href="Twenty Four">Twenty Four</a> (<a href="24">24</a>) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

## PAGE 15

§ 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1; 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes; (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.

#### PAGE 17

§ 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.

#### PAGE 18

[X] Litigation in a court of competent jurisdiction

#### PAGE 19

§ 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

§ 9.4 Either party may terminate this Agreement upon not less than seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.

Additions and Deletions Report for AIA Document B101" – 2017. Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects." AIA, "the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 12:23:55 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

User Notes:

§ 9.5 The Owner may terminate this Agreement upon not less than seven days' written notice to the Architect for the Owner's convenience and without cause.

## PAGE 20

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

As mutually agreed

## PAGE 21

Compensation in the amount of Forty Nine Thousand Five Hundred Dollars and Zero Cents (\$49,500.00)

Hourly or as mutually agreed upon in advance of services

Hourly or as mutually agreed upon in advance of services

#### PAGE 22

As mutually agreed by Owner and Architect in advance of services

| Schematic Design Phase   | Twenty                     | percent (       | 20_  | %)    |
|--------------------------|----------------------------|-----------------|--|-------|
| Design Development Phase | Twenty Five                | percent (       | 25   | %)    |
| Construction Documents   | Twenty Five                | percent (       | 25   | %)    |
| Phase                    | San Hilliam Milliam Commen | Angressen at AN | the same to the sa | 35.90 |
| Procurement Phase        | Five                       | percent (       | 5  | %)    |
| Construction Phase       | Twenty Five                | percent (       | 25   | %)    |

See Exhibit C - Hourly Compensation and Reimbursable Expenses

N/A N/A PAGE 23

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus Ten percent ( 10 %) of the expenses incurred.

N/A

§ 11.10.1.1 An initial payment of Zero\_(\$ 0\_) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice. § 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of Zero (\$ 0 ) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred. § 11.10.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid Forty-Five (45) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect. 1.5 % Monthly § 11.10.2.4 Owner Agrees to receive invoices by email, which will be sent to the following individual and email address: Full Name: Kelly Christopherson Email Address: kellv.christopherson@k12.sd.us None PAGE 24 Exhibit A: Project Initial Information Exhibit B: JLG Image Release Exhibit C: Schedule of Hourly Compensation & Reimbursable Expenses Garret Bischoff, President Catherine Dekkenga, AIA. Principal

## Certification of Document's Authenticity

AIA® Document D401™ - 2003

I, Catherine Dekkenga, AIA, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 12:23:55 ET on 10/05/2022 under Order No. 2114257989 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA<sup>®</sup> Document BI01<sup>TM</sup> - 2017, Standard Form of Agreement Between Owner and Architect, as published by the AIA in its software, other than those additions and deletions shown in the associated Additions and Deletions Report.

| (Signed) |  |
|----------|--|
|          |  |
| (Title)  |  |
| (Dated)  |  |



# Standard Form of Agreement Between Owner and Architect

AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two (In words, indicate day, month and year.)

BETWEEN the Architect's client identified as the Owner: (Name, legal status, address and other information)

Huron Public School District 150 5th St. SW Huron, SD 57350

and the Architect: (Name, legal status, address and other information)

JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104

for the following Project: (Name, location and detailed description)

21392.01 - Huron School District - TAC A/C Adding A/C coil units in the existing TAC building

The Owner and Architect agree as follows.

## ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences, Consultation with an attorney is encouraged with respect to its completion or modification.

#### TABLE OF ARTICLES

- 1 INITIAL INFORMATION
- 2 ARCHITECT'S RESPONSIBILITIES
- 3 SCOPE OF ARCHITECT'S BASIC SERVICES
- 4 SUPPLEMENTAL AND ADDITIONAL SERVICES
- 5 OWNER'S RESPONSIBILITIES
- 6 COST OF THE WORK
- 7 COPYRIGHTS AND LICENSES
- 8 CLAIMS AND DISPUTES
- 9 TERMINATION OR SUSPENSION
- 10 MISCELLANEOUS PROVISIONS
- 11 COMPENSATION
- 12 SPECIAL TERMS AND CONDITIONS
- 13 SCOPE OF THE AGREEMENT

#### ARTICLE 1 INITIAL INFORMATION

§ 1.1 This Agreement is based on the Initial Information set forth in this Section 1.1.

(For each item in this section, insert the information or a statement such as "not applicable" or "unknown at time of execution.")

## § 1.1.1 The Owner's program for the Project:

(Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.)

Adding A/C coils unit in the existing TAC building

## § 1.1.2 The Project's physical characteristics:

(Identify or describe pertinent information about the Project's physical characteristics, such as size; location; dimensions; geotechnical reports; site boundaries; topographic surveys; traffic and utility studies; availability of public and private utilities and services; legal description of the site, etc.)

Existing activity building Owned and operated by the School District

§ 1.1.3 The Owner's budget for the Cost of the Work, as defined in Section 6.1: (Provide total and, if known, a line item breakdown.)

\$260,000.00

- § 1.1.4 The Owner's anticipated design and construction milestone dates:
  - .1 Design phase milestone dates, if any:

1

Bidding: Fall 2022

Construction to commence spring summer of 2023

.2 Construction commencement date:

N/A

.3 Substantial Completion date or dates:

N/A

.4 Other milestone dates:

§ 1.1.5 The Owner intends the following procurement and delivery method for the Project: (Identify method such as competitive bid or negotiated contract, as well as any requirements for accelerated or fast-track design and construction, multiple bid packages, or phased construction.)

Design/Bid/Build

§ 1.1.6 The Owner's anticipated Sustainable Objective for the Project: (Identify and describe the Owner's Sustainable Objective for the Project, if any.)

N/A

(Paragraph Deleted)

§ 1.1.6.1 If the Owner identifies a Sustainable Objective, the Owner and Architect shall complete and incorporate AIA Document E204<sup>TM</sup>—2017, Sustainable Projects Exhibit, into this Agreement to define the terms, conditions and services related to the Owner's Sustainable Objective. If E204—2017 is incorporated into this agreement, the Owner and Architect shall incorporate the completed E204—2017 into the agreements with the consultants and contractors performing services or Work in any way associated with the Sustainable Objective.

§ 1.1.7 The Owner identifies the following representative in accordance with Section 5.3: (List name, address, and other contact information.)

Kelly Christopherson Huron Public School District 150 5th St. SW Huron, SD 57350

Telephone:

Email: kelly.christopherson@k12.sd.us

§ 1.1.8 The persons or entities, in addition to the Owner's representative, who are required to review the Architect's submittals to the Owner are as follows:

(List name, address, and other contact information.)

N/A

§ 1.1.9 The Owner shall retain the following consultants and contractors: (List name, legal status, address, and other contact information.)

.1 Geotechnical Engineer:

N/A

.2 Civil Engineer:

N/A

.3 Other, if any:

(List any other consultants and contractors retained by the Owner.)

N/A

§ 1.1.10 The Architect identifies the following representative in accordance with Section 2.3: (List name, address, and other contact information.)

Catherine Dekkenga JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104

Telephone: 605-271-2531

Email: cdekkenga@jlgarchitects.com

§ 1.1.11 The Architect shall retain the consultants identified in Sections 1.1.11.1 and 1.1.11.2: (List name, legal status, address, and other contact information.)

- § 1.1.11.1 Consultants retained under Basic Services:
  - .1 Structural Engineer:

N/A

.2 Mechanical Engineer:

Associated Consulting Engineers, Inc., Jordan deWit 340 S. Phillips Avc. Sioux Falls, SD 57104 Telephone: 605-335-3720

.3 Electrical Engineer:

Associated Consulting Engineers, Inc., Brad Shoup 340 S. Phillips Avc. Sioux Falls, SD 57104 Telephone: 605-335-3720

§ 1.1.11.2 Consultants retained under Supplemental Services:

N/A

§ 1.1.12 Other Initial Information on which the Agreement is based:

N/A

- § 1.2 The Owner and Architect may rely on the Initial Information. Both parties, however, recognize that the Initial Information may materially change and, in that event, the Owner and the Architect shall appropriately adjust the Architect's services, schedule for the Architect's services, and the Architect's compensation. The Owner shall adjust the Owner's budget for the Cost of the Work and the Owner's anticipated design and construction milestones, as necessary, to accommodate material changes in the Initial Information.
- § 1.3 The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. The parties will use AIA Document E203™−2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.
- § 1.3.1 Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in AIA Document E203<sup>TM</sup>-2013, Building Information Modeling and Digital Data Exhibit, and the requisite AIA Document G202<sup>TM</sup>-2013, Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

## ARTICLE 2 ARCHITECT'S RESPONSIBILITIES

- § 2.1 The Architect shall provide professional services as set forth in this Agreement. The Architect represents that it is properly licensed in the jurisdiction where the Project is located to provide the services required by this Agreement, or shall cause such services to be performed by appropriately licensed design professionals.
- § 2.2 The Architect shall perform its services consistent with the professional skill and care ordinarily provided by architects practicing in the same or similar locality under the same or similar circumstances. The Architect shall perform its services as expeditiously as is consistent with such professional skill and care and the orderly progress of the Project.
- § 2.3 The Architect shall identify a representative authorized to act on behalf of the Architect with respect to the Project.
- § 2.4 Except with the Owner's knowledge and consent, the Architect shall not engage in any activity, or accept any employment, interest or contribution that would reasonably appear to compromise the Architect's professional judgment with respect to this Project.
- § 2.5 The Architect shall maintain the following insurance until termination of this Agreement. If any of the requirements set forth below are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect as set forth in Section 11.9.
- § 2.5.1 Commercial General Liability with policy limits of One Million Dollars and Zero Cents (\$ 1,000,000.00) for each occurrence and Two Million Dollars and Zero Cents (\$ 2,000,000.00) in the aggregate for bodily injury and property damage.

- § 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of One Million Dollars and Zero Cents (\$1,000,000.00) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.
- § 2.5.3 The Architect may achieve the required limits and coverage for Commercial General Liability and Automobile Liability through a combination of primary and excess or umbrella liability insurance, provided such primary and excess or umbrella liability insurance policies result in the same or greater coverage as the coverages required under Sections 2.5.1 and 2.5.2, and in no event shall any excess or umbrella liability insurance provide narrower coverage than the primary policy. The excess policy shall not require the exhaustion of the underlying limits only through the actual payment by the underlying insurers.
- § 2.5.4 Workers' Compensation at statutory limits.
- § 2.5.5 Employers' Liability with policy limits Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each accident, Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000.00 ) policy limit.
- § 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) per claim and Four Million Dollars and Zero Cents (\$ 4,000,000.00 ) in the aggregate.
- § 2.5.7 Additional Insured Obligations. To the fullest extent permitted by law, the Architect shall cause the primary and excess or umbrella polices for Commercial General Liability and Automobile Liability to include the Owner as an additional insured for claims caused in whole or in part by the Architect's negligent acts or omissions. The additional insured coverage shall be primary and non-contributory to any of the Owner's insurance policies and shall apply to both ongoing and completed operations.
- § 2.5.8 The Architect shall provide certificates of insurance to the Owner that evidence compliance with the requirements in this Section 2.5.

## ARTICLE 3 SCOPE OF ARCHITECT'S BASIC SERVICES

- § 3.1 The Architect's Basic Services consist of those described in this Article 3 and include structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.
- § 3.1.1 The Architect shall manage the Architect's services, research applicable design criteria, attend Project meetings, communicate with members of the Project team, and report progress to the Owner.
- § 3.1.2 The Architect shall coordinate its services with those services provided by the Owner and the Owner's consultants. The Architect shall be entitled to rely on, and shall not be responsible for, the accuracy, completeness, and timeliness of, services and information furnished by the Owner and the Owner's consultants. The Architect shall provide prompt written notice to the Owner if the Architect becomes aware of any error, omission, or inconsistency in such services or information.
- § 3.1.3 As soon as practicable after the date of this Agreement, the Architect shall submit for the Owner's approval a schedule for the performance of the Architect's services. The schedule initially shall include anticipated dates for the commencement of construction and for Substantial Completion of the Work as set forth in the Initial Information. The schedule shall include allowances for periods of time required for the Owner's review, for the performance of the Owner's consultants, and for approval of submissions by authorities having jurisdiction over the Project. Once approved by the Owner, time limits established by the schedule shall not, except for reasonable cause, be exceeded by the Architect or Owner. With the Owner's approval, the Architect shall adjust the schedule, if necessary, as the Project proceeds until the commencement of construction.
- § 3.1.4 The Architect shall not be responsible for an Owner's directive or substitution, or for the Owner's acceptance of non-conforming Work, made or given without the Architect's written approval.

- § 3.1.5 The Architect shall contact governmental authorities required to approve the Construction Documents and entities providing utility services to the Project. The Architect shall respond to applicable design requirements imposed by those authorities and entities.
- § 3.1.6 The Architect shall assist the Owner in connection with the Owner's responsibility for filing documents required for the approval of governmental authorities having jurisdiction over the Project.

## § 3.2 Schematic Design Phase Services

- § 3.2.1 The Architect shall review the program and other information furnished by the Owner, and shall review laws, codes, and regulations applicable to the Architect's services.
- § 3.2.2 The Architect shall prepare a preliminary evaluation of the Owner's program, schedule, budget for the Cost of the Work, Project site, the proposed procurement and delivery method, and other Initial Information, each in terms of the other, to ascertain the requirements of the Project. The Architect shall notify the Owner of (1) any inconsistencies discovered in the information, and (2) other information or consulting services that may be reasonably needed for the Project.
- § 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner approach(es) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.
- § 3.2.4 Based on the Project requirements agreed upon with the Owner, the Architect shall prepare and present, for the Owner's approval, a preliminary design illustrating the scale and relationship of the Project components.
- § 3.2.5 Based on the Owner's approval of the preliminary design, the Architect shall prepare Schematic Design Documents for the Owner's approval. The Schematic Design Documents shall consist of drawings and other documents including a site plan, if appropriate, and preliminary building plans, sections and elevations; and may include some combination of study models, perspective sketches, or digital representations. Preliminary selections of major building systems and construction materials shall be noted on the drawings or described in writing.
- § 3.2.5.1 The Architect shall consider sustainable design alternatives, such as material choices and building orientation, together with other considerations based on program and aesthetics, in developing a design that is consistent with the Owner's program, schedule and budget for the Cost of the Work. The Owner may obtain more advanced sustainable design services as a Supplemental Service under Section 4.1.1.
- § 3.2.5.2 The Architect shall consider the value of materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.
- § 3.2.6 The Architect shall submit to the Owner an estimate of the Cost of the Work prepared in accordance with Section 6.3.
- § 3.2.7 The Architect shall submit the Schematic Design Documents to the Owner, and request the Owner's approval.

## § 3.3 Design Development Phase Services

- § 3.3.1 Based on the Owner's approval of the Schematic Design Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Design Development Documents for the Owner's approval. The Design Development Documents shall illustrate and describe the development of the approved Schematic Design Documents and shall consist of drawings and other documents including plans, sections, elevations, typical construction details, and diagrammatic layouts of building systems to fix and describe the size and character of the Project as to architectural, structural, mechanical and electrical systems, and other appropriate elements. The Design Development Documents shall also include outline specifications that identify major materials and systems and establish, in general, their quality levels.
- § 3.3.2 The Architect shall update the estimate of the Cost of the Work prepared in accordance with Section 6.3.

§ 3.3.3 The Architect shall submit the Design Development Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, and request the Owner's approval.

#### § 3.4 Construction Documents Phase Services

- § 3.4.1 Based on the Owner's approval of the Design Development Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Construction Documents for the Owner's approval. The Construction Documents shall illustrate and describe the further development of the approved Design Development Documents and shall consist of Drawings and Specifications setting forth in detail the quality levels and performance criteria of materials and systems and other requirements for the construction of the Work. The Owner and Architect acknowledge that, in order to perform the Work, the Contractor will provide additional information, including Shop Drawings, Product Data, Samples and other similar submittals, which the Architect shall review in accordance with Section 3.6.4.
- § 3.4.2 The Architect shall incorporate the design requirements of governmental authorities having jurisdiction over the Project into the Construction Documents.
- § 3.4.3 During the development of the Construction Documents, the Architect shall assist the Owner in the development and preparation of (1) procurement information that describes the time, place, and conditions of bidding, including bidding or proposal forms; (2) the form of agreement between the Owner and Contractor; and (3) the Conditions of the Contract for Construction (General, Supplementary and other Conditions). The Architect shall also compile a project manual that includes the Conditions of the Contract for Construction and Specifications, and may include bidding requirements and sample forms.
- § 3.4.4 The Architect shall update the estimate for the Cost of the Work prepared in accordance with Section 6.3.
- § 3.4.5 The Architect shall submit the Construction Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, take any action required under Section 6.5, and request the Owner's approval.

#### § 3.5 Procurement Phase Services

## § 3.5.1 General

The Architect shall assist the Owner in establishing a list of prospective contractors. Following the Owner's approval of the Construction Documents, the Architect shall assist the Owner in (1) obtaining either competitive bids or negotiated proposals; (2) confirming responsiveness of bids or proposals; (3) determining the successful bid or proposal, if any; and, (4) awarding and preparing contracts for construction.

#### § 3.5.2 Competitive Bidding

- § 3.5.2.1 Bidding Documents shall consist of bidding requirements and proposed Contract Documents.
- § 3.5.2.2 The Architect shall assist the Owner in bidding the Project by:
  - .1 facilitating the distribution of Bidding Documents to prospective bidders;
  - organizing and conducting a pre-bid conference for prospective bidders;
  - .3 preparing responses to questions from prospective bidders and providing clarifications and interpretations of the Bidding Documents to the prospective bidders in the form of addenda; and,
  - .4 organizing and conducting the opening of the bids, and subsequently documenting and distributing the bidding results, as directed by the Owner.
- § 3.5.2.3 If the Bidding Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective bidders.

#### § 3.5.3 Negotiated Proposals

- § 3.5.3.1 Proposal Documents shall consist of proposal requirements and proposed Contract Documents.
- § 3.5.3.2 The Architect shall assist the Owner in obtaining proposals by:
  - .1 facilitating the distribution of Proposal Documents for distribution to prospective contractors and requesting their return upon completion of the negotiation process;

- .2 organizing and participating in selection interviews with prospective contractors:
- .3 preparing responses to questions from prospective contractors and providing clarifications and interpretations of the Proposal Documents to the prospective contractors in the form of addenda; and,
- .4 participating in negotiations with prospective contractors, and subsequently preparing a summary report of the negotiation results, as directed by the Owner.

§ 3.5.3.3 If the Proposal Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective contractors.

## § 3.6 Construction Phase Services

## § 3.6.1 General

- § 3.6.1.1 The Architect shall provide administration of the Contract between the Owner and the Contractor as set forth below and in AIA Document A201™–2017, General Conditions of the Contract for Construction. If the Owner and Contractor modify AIA Document A201–2017, those modifications shall not affect the Architect's services under this Agreement unless the Owner and the Architect amend this Agreement.
- § 3.6.1.2 The Architect shall advise and consult with the Owner during the Construction Phase Services. The Architect shall have authority to act on behalf of the Owner only to the extent provided in this Agreement. The Architect shall not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, nor shall the Architect be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect shall be responsible for the Architect's negligent acts or omissions, but shall not have control over or charge of, and shall not be responsible for, acts or omissions of the Contractor or of any other persons or entities performing portions of the Work.
- § 3.6.1.3 Subject to Section 4.2 and except as provided in Section 3.6.6.5, the Architect's responsibility to provide Construction Phase Services commences with the award of the Contract for Construction and terminates on the date the Architect issues the final Certificate for Payment.

#### § 3.6.2 Evaluations of the Work

- § 3.6.2.1 The Architect shall visit the site at points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.
- § 3.6.2.2 The Architect has the authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect shall have the authority to require inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not the Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, suppliers, their agents or employees, or other persons or entities performing portions of the Work.
- § 3.6.2.3 The Architect shall interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests shall be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- § 3.6.2.4 Interpretations and decisions of the Architect shall be consistent with the intent of, and reasonably inferable from, the Contract Documents and shall be in writing or in the form of drawings. When making such interpretations and decisions, the Architect shall endeavor to secure faithful performance by both Owner and Contractor, shall not show partiality to either, and shall not be liable for results of interpretations or decisions rendered in good faith. The

Architect's decisions on matters relating to aesthetic effect shall be final if consistent with the intent expressed in the Contract Documents.

§ 3.6.2.5 Unless the Owner and Contractor designate another person to serve as an Initial Decision Maker, as that term is defined in AIA Document A201–2017, the Architect shall render initial decisions on Claims between the Owner and Contractor as provided in the Contract Documents.

## § 3.6.3 Certificates for Payment to Contractor

§ 3.6.3.1 The Architect shall review and certify the amounts due the Contractor and shall issue certificates in such amounts. The Architect's certification for payment shall constitute a representation to the Owner, based on the Architect's evaluation of the Work as provided in Section 3.6.2 and on the data comprising the Contractor's Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified. The foregoing representations are subject to (1) an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, (2) results of subsequent tests and inspections, (3) correction of minor deviations from the Contract Documents prior to completion, and (4) specific qualifications expressed by the Architect.

§ 3.6.3.2 The issuance of a Certificate for Payment shall not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) ascertained how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

§ 3.6.3.3 The Architect shall maintain a record of the Applications and Certificates for Payment.

#### § 3.6.4 Submittals

§ 3.6.4.1 The Architect shall review the Contractor's submittal schedule and shall not unreasonably delay or withhold approval of the schedule. The Architect's action in reviewing submittals shall be taken in accordance with the approved submittal schedule or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time, in the Architect's professional judgment, to permit adequate review.

§ 3.6.4.2 The Architect shall review, or take other appropriate action per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

§ 3.6.4.3 If the Contract Documents specifically require the Contractor to provide professional design services or certifications by a design professional related to systems, materials, or equipment, the Architect shall specify the appropriate performance and design criteria that such services must satisfy. The Architect shall review and take appropriate action on Shop Drawings and other submittals related to the Work designed or certified by the Contractor's design professional, provided the submittals bear such professional's seal and signature when submitted to the Architect. The Architect's review shall be for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect shall be entitled to rely upon, and shall not be responsible for, the adequacy and accuracy of the services, certifications, and approvals performed or provided by such design professionals.

§ 3.6.4.4 Subject to Section 4.2, the Architect shall review and respond to requests for information about the Contract Documents. The Architect shall set forth, in the Contract Documents, the requirements for requests for information. Requests for information shall include, at a minimum, a detailed written statement that indicates the specific Drawings or Specifications in need of clarification and the nature of the clarification requested. The Architect's response to such requests shall be made in writing within any time limits agreed upon, or otherwise with reasonable

promptness. If appropriate, the Architect shall prepare and issue supplemental Drawings and Specifications in response to the requests for information.

§ 3.6.4.5 The Architect shall maintain a record of submittals and copies of submittals supplied by the Contractor in accordance with the requirements of the Contract Documents.

#### § 3.6.5 Changes in the Work

§ 3.6.5.1 The Architect may order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. Subject to Section 4.2, the Architect shall prepare Change Orders and Construction Change Directives for the Owner's approval and execution in accordance with the Contract Documents.

§ 3.6.5.2 The Architect shall maintain records relative to changes in the Work.

## § 3.6.6 Project Completion

## § 3.6.6.1 The Architect shall:

- .1 conduct inspections to determine the date or dates of Substantial Completion and the date of final completion;
- .2 issue Certificates of Substantial Completion;
- .3 forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract Documents and received from the Contractor; and,
- .4 issue a final Certificate for Payment based upon a final inspection indicating that, to the best of the Architect's knowledge, information, and belief, the Work complies with the requirements of the Contract Documents.
- § 3.6.6.2 The Architect's inspections shall be conducted with the Owner to check conformance of the Work with the requirements of the Contract Documents and to verify the accuracy and completeness of the list submitted by the Contractor of Work to be completed or corrected.
- § 3.6.6.3 When Substantial Completion has been achieved, the Architect shall inform the Owner about the balance of the Contract Sum remaining to be paid the Contractor, including the amount to be retained from the Contract Sum, if any, for final completion or correction of the Work.
- § 3.6.4 The Architect shall forward to the Owner the following information received from the Contractor: (1) consent of surety or sureties, if any, to reduction in or partial release of retainage or the making of final payment; (2) affidavits, receipts, releases and waivers of liens, or bonds indemnifying the Owner against liens; and (3) any other documentation required of the Contractor under the Contract Documents.
- § 3.6.6.5 Upon request of the Owner, and prior to the expiration of one year from the date of Substantial Completion, the Architect shall, without additional compensation, conduct a meeting with the Owner to review the facility operations and performance.

## § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- 1. The Parties agree and understand that, during the course of the construction of the Work, the Owner, Contractor, or other Project participants may choose to, at their own expense and initiative, make use of supplemental video imagery and information sourced from devices such as project webcams, Unmanned Aerial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain visual information and documentation regarding the state of the progress and/or quality of the Work (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly agree and acknowledge the following:
  - Since Supplementary Video Information encompasses only certain limited visual information
    and provides neither complete nor comprehensive visual information/measurements/dimensions,
    nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video
    Information shall not be used as a substitute for or required supplement to Architect's Construction
    Phase services, herein, but rather may, to the extent provided and made available to Architect in a
    reasonably timely, legible, and complete fashion, supplement the other data and information
    obtained by Architect pursuant to Architect's Construction Phase Services, herein.

- Architect shall have no obligation to conduct a comprehensive or continuous review of any Supplementary Video Information, but rather may, at its sole discretion, choose to review such Supplementary Video Information, or any portion thereof, in connection with the performance of its Construction Phase Services on the Project.
- 3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part thereof.

## ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

## § 4.1 Supplemental Services

§ 4.1.1 The services listed below are not included in Basic Services but may be required for the Project. The Architect shall provide the listed Supplemental Services only if specifically designated in the table below as the Architect's responsibility, and the Owner shall compensate the Architect as provided in Section 11.2. Unless otherwise specifically addressed in this Agreement, if neither the Owner nor the Architect is designated, the parties agree that the listed Supplemental Service is not being provided for the Project.

(Designate the Architect's Supplemental Services and the Owner's Supplemental Services required for the Project by indicating whether the Architect or Owner shall be responsible for providing the identified Supplemental Service. Insert a description of the Supplemental Services in Section 4.1.2 below or attach the description of services as an exhibit to this Agreement.)

| Supplemental Services |  | Responsibility<br>(Architect, Owner, or no.<br>provided) |  |
|-----------------------|--|--|--|
| § 4.1.1.1             | Programming  | Not Provided   |  |
| § 4.1.1.2             | Multiple preliminary designs   | Not Provided   |  |
| § 4.1.1.3             | Measured drawings  | Not Provided   |  |
| § 4.1.1.4             | Existing facilities surveys  | Not Provided   |  |
| § 4.1.1.5             | Site evaluation and planning   | Not Provided   |  |
| § 4.1.1.6             | Building Information Model management responsibilities               | Not Provided   |  |
| § 4.1.1.7             | Development of Building Information Models for post construction use | Not Provided   |  |
| § 4.1.1.8             | Civil engineering  | Not Provided   |  |
| § 4.1.1.9             | Landscape design   | Not Provided   |  |
| § 4.1.1.10            | Architectural interior design  | Not Provided   |  |
| § 4.1.1.11            | Value analysis   | Not Provided   |  |
| § 4.1.1.12            | Detailed cost estimating beyond that required in Section 6.3         | Not Provided   |  |
| § 4.1.1.13            | On-site project representation                                       | Not Provided   |  |
| § 4.1.1.14            | Conformed documents for construction                                 | Not Provided   |  |
| § 4.1.1.15            | As-designed record drawings  | Not Provided   |  |
| § 4.1.1.16            | As-constructed record drawings                                       | Not Provided   |  |
| § 4.1.1.17            | Post-occupancy evaluation  | Not Provided   |  |
| § 4.1.1.18            | Facility support services  | Not Provided   |  |
| § 4.1.1.19            | Tenant-related services  | Not Provided   |  |
| § 4.1.1.20            | Architect's coordination of the Owner's consultants                  | Not Provided   |  |
| § 4.1.1.21            | Telecommunications/data design                                       | Not Provided   |  |
| § 4.1.1.22            | Security evaluation and planning                                     | Not Provided   |  |

| § 4.1.1.23 | Commissioning  | Not Provided |
|------------|--|--------------|
| § 4.1.1.24 | Sustainable Project Services pursuant to Section 4.1.3 | Not Provided |
| § 4.1.1.25 | Fast-track design services                             | Not Provided |
| § 4.1.1.26 | Multiple bid packages                                  | Not Provided |
| § 4.1.1.27 | Historic preservation                                  | Not Provided |
| § 4.1.1.28 | Furniture, furnishings, and equipment design           | Not Provided |
| § 4.1.1.29 | Other services provided by specialty Consultants       | Not Provided |
| § 4.1.1.30 | Other Supplemental Services                            | Not Provided |

## § 4.1.2 Description of Supplemental Services

§ 4.1.2.1 A description of each Supplemental Service identified in Section 4.1.1 as the Architect's responsibility is provided below.

(Describe in detail the Architect's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit. The AIA publishes a number of Standard Form of Architect's Services documents that can be included as an exhibit to describe the Architect's Supplemental Services.)

N/A

§ 4.1.2.2 A description of each Supplemental Service identified in Section 4.1.1 as the Owner's responsibility is provided below.

(Describe in detail the Owner's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit.)

N/A

§ 4.1.3 If the Owner identified a Sustainable Objective in Article 1, the Architect shall provide, as a Supplemental Service, the Sustainability Services required in AIA Document E204<sup>TM</sup>–2017, Sustainable Projects Exhibit, attached to this Agreement. The Owner shall compensate the Architect as provided in Section 11.2.

#### § 4.2 Architect's Additional Services

The Architect may provide Additional Services after execution of this Agreement without invalidating the Agreement. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 4.2 shall entitle the Architect to compensation pursuant to Section 11.3 and an appropriate adjustment in the Architect's schedule.

- § 4.2.1 Upon recognizing the need to perform the following Additional Services, the Architect shall notify the Owner with reasonable promptness and explain the facts and circumstances giving rise to the need. The Architect shall not proceed to provide the following Additional Services until the Architect receives the Owner's written authorization:
  - .1 Services necessitated by a change in the Initial Information, previous instructions or approvals given by the Owner, or a material change in the Project including size, quality, complexity, the Owner's schedule or budget for Cost of the Work, or procurement or delivery method;
  - .2 Services necessitated by the enactment or revision of codes, laws, or regulations, including changing or editing previously prepared Instruments of Service;
  - .3 Changing or editing previously prepared Instruments of Service necessitated by official interpretations of applicable codes, laws or regulations that are either (a) contrary to specific interpretations by the applicable authorities having jurisdiction made prior to the issuance of the building permit, or (b) contrary to requirements of the Instruments of Service when those Instruments of Service were prepared in accordance with the applicable standard of care;
  - .4 Services necessitated by decisions of the Owner not rendered in a timely manner or any other failure of performance on the part of the Owner or the Owner's consultants or contractors;
  - .5 Preparing digital models or other design documentation for transmission to the Owner's consultants and contractors, or to other Owner-authorized recipients;
  - 6. Preparation of design and documentation for alternate bid or proposal requests proposed by the Owner;
  - .7 Preparation for, and attendance at, a public presentation, meeting or hearing;

- .8 Preparation for, and attendance at, a dispute resolution proceeding or legal proceeding, except where the Architect is party thereto;
- .9 Evaluation of the qualifications of entities providing bids or proposals;
- .10 Consultation concerning replacement of Work resulting from fire or other cause during construction; or,
- .11 Assistance to the Initial Decision Maker, if other than the Architect.
- § 4.2.2 To avoid delay in the Construction Phase, the Architect shall provide the following Additional Services, notify the Owner with reasonable promptness, and explain the facts and circumstances giving rise to the need. If, upon receipt of the Architect's notice, the Owner determines that all or parts of the services are not required, the Owner shall give prompt written notice to the Architect of the Owner's determination. The Owner shall compensate the Architect for the services provided prior to the Architect's receipt of the Owner's notice.
  - Reviewing a Contractor's submittal out of sequence from the submittal schedule approved by the Architect;
  - .2 Responding to the Contractor's requests for information that are not prepared in accordance with the Contract Documents or where such information is available to the Contractor from a careful study and comparison of the Contract Documents, field conditions, other Owner-provided information, Contractor-prepared coordination drawings, or prior Project correspondence or documentation;
  - .3 Preparing Change Orders and Construction Change Directives that require evaluation of Contractor's proposals and supporting data, or the preparation or revision of Instruments of Service;
  - .4 Evaluating an extensive number of Claims as the Initial Decision Maker; or,
  - .5 Evaluating substitutions proposed by the Owner or Contractor and making subsequent revisions to Instruments of Service resulting therefrom.
- § 4.2.3 The Architect shall provide Construction Phase Services exceeding the limits set forth below as Additional Services. When the limits below are reached, the Architect shall notify the Owner:
  - .1 Two (2) reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor
  - .2 Bi-weekly visits to the site by the Architect during construction
  - .3 One (1) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents
  - .4 One (1) inspections for any portion of the Work to determine final completion.
- § 4.2.4 Except for services required under Section 3.6.6.5 and those services that do not exceed the limits set forth in Section 4.2.3, Construction Phase Services provided more than 60 days after (1) the date of Substantial Completion of the Work or (2) the initial date of Substantial Completion identified in the agreement between the Owner and Contractor, whichever is earlier, shall be compensated as Additional Services to the extent the Architect incurs additional cost in providing those Construction Phase Services.
- § 4.2.5 If the services covered by this Agreement have not been completed within Twenty Four (24) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

## ARTICLE 5 OWNER'S RESPONSIBILITIES

- § 5.1 Unless otherwise provided for under this Agreement, the Owner shall provide information in a timely manner regarding requirements for and limitations on the Project, including a written program, which shall set forth the Owner's objectives; schedule; constraints and criteria, including space requirements and relationships; flexibility; expandability; special equipment; systems; and site requirements.
- § 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes; (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the

1

Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.

- § 5.3 The Owner shall identify a representative authorized to act on the Owner's behalf with respect to the Project. The Owner shall render decisions and approve the Architect's submittals in a timely manner in order to avoid unreasonable delay in the orderly and sequential progress of the Architect's services.
- § 5.4 The Owner shall furnish surveys to describe physical characteristics, legal limitations and utility locations for the site of the Project, and a written legal description of the site. The surveys and legal information shall include, as applicable, grades and lines of streets, alleys, pavements and adjoining property and structures; designated wetlands; adjacent drainage; rights-of-way, restrictions, easements, encroachments, zoning, deed restrictions, boundaries and contours of the site; locations, dimensions, and other necessary data with respect to existing buildings, other improvements and trees; and information concerning available utility services and lines, both public and private, above and below grade, including inverts and depths. All the information on the survey shall be referenced to a Project benchmark.
- § 5.5 The Owner shall furnish services of geotechnical engineers, which may include test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.
- § 5.6 The Owner shall provide the Supplemental Services designated as the Owner's responsibility in Section 4.1.1.
- § 5.7 If the Owner identified a Sustainable Objective in Article 1, the Owner shall fulfill its responsibilities as required in AIA Document E204™ 2017, Sustainable Projects Exhibit, attached to this Agreement.
- § 5.8 The Owner shall coordinate the services of its own consultants with those services provided by the Architect. Upon the Architect's request, the Owner shall furnish copies of the scope of services in the contracts between the Owner and the Owner's consultants. The Owner shall furnish the services of consultants other than those designated as the responsibility of the Architect in this Agreement, or authorize the Architect to furnish them as an Additional Service, when the Architect requests such services and demonstrates that they are reasonably required by the scope of the Project. The Owner shall require that its consultants and contractors maintain insurance, including professional liability insurance, as appropriate to the services or work provided.
- § 5.9 The Owner shall furnish tests, inspections and reports required by law or the Contract Documents, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials.
- § 5.10 The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.
- § 5.11 The Owner shall provide prompt written notice to the Architect if the Owner becomes aware of any fault or defect in the Project, including errors, omissions or inconsistencies in the Architect's Instruments of Service.
- § 5.12 The Owner shall include the Architect in all communications with the Contractor that relate to or affect the Architect's services or professional responsibilities. The Owner shall promptly notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect.
- § 5.13 Before executing the Contract for Construction, the Owner shall coordinate the Architect's duties and responsibilities set forth in the Contract for Construction with the Architect's services set forth in this Agreement. The Owner shall provide the Architect a copy of the executed agreement between the Owner and Contractor, including the General Conditions of the Contract for Construction.
- § 5.14 The Owner shall provide the Architect access to the Project site prior to commencement of the Work and shall obligate the Contractor to provide the Architect access to the Work wherever it is in preparation or progress.

§ 5.15 Within 15 days after receipt of a written request from the Architect, the Owner shall furnish the requested information as necessary and relevant for the Architect to evaluate, give notice of, or enforce lien rights.

## ARTICLE 6 COST OF THE WORK

- § 6.1 For purposes of this Agreement, the Cost of the Work shall be the total cost to the Owner to construct all elements of the Project designed or specified by the Architect and shall include contractors' general conditions costs, overhead and profit. The Cost of the Work also includes the reasonable value of labor, materials, and equipment, donated to, or otherwise furnished by, the Owner. The Cost of the Work does not include the compensation of the Architect; the costs of the land, rights-of-way, financing, or contingencies for changes in the Work; or other costs that are the responsibility of the Owner.
- § 6.2 The Owner's budget for the Cost of the Work is provided in Initial Information, and shall be adjusted throughout the Project as required under Sections 5.2, 6.4 and 6.5. Evaluations of the Owner's budget for the Cost of the Work, and the preliminary estimate of the Cost of the Work and updated estimates of the Cost of the Work, prepared by the Architect, represent the Architect's judgment as a design professional. It is recognized, however, that neither the Architect nor the Owner has control over the cost of labor, materials, or equipment; the Contractor's methods of determining bid prices; or competitive bidding, market, or negotiating conditions. Accordingly, the Architect cannot and does not warrant or represent that bids or negotiated prices will not vary from the Owner's budget for the Cost of the Work, or from any estimate of the Cost of the Work, or evaluation, prepared or agreed to by the Architect.
- § 6.3 In preparing estimates of the Cost of Work, the Architect shall be permitted to include contingencies for design, bidding, and price escalation; to determine what materials, equipment, component systems, and types of construction are to be included in the Contract Documents; to recommend reasonable adjustments in the program and scope of the Project; and to include design alternates as may be necessary to adjust the estimated Cost of the Work to meet the Owner's budget. The Architect's estimate of the Cost of the Work shall be based on current area, volume or similar conceptual estimating techniques. If the Owner requires a detailed estimate of the Cost of the Work, the Architect shall provide such an estimate, if identified as the Architect's responsibility in Section 4.1.1, as a Supplemental Service.
- § 6.4 If, through no fault of the Architect, the Procurement Phase has not commenced within 90 days after the Architect submits the Construction Documents to the Owner, the Owner's budget for the Cost of the Work shall be adjusted to reflect changes in the general level of prices in the applicable construction market.
- § 6.5 If at any time the Architect's estimate of the Cost of the Work exceeds the Owner's budget for the Cost of the Work, the Architect shall make appropriate recommendations to the Owner to adjust the Project's size, quality, or budget for the Cost of the Work, and the Owner shall cooperate with the Architect in making such adjustments.
- § 6.6 If the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services is exceeded by the lowest bona fide bid or negotiated proposal, the Owner shall
  - .1 give written approval of an increase in the budget for the Cost of the Work;
  - .2 authorize rebidding or renegotiating of the Project within a reasonable time;
  - .3 terminate in accordance with Section 9.5;
  - .4 in consultation with the Architect, revise the Project program, scope, or quality as required to reduce the Cost of the Work; or,
  - .5 implement any other mutually acceptable alternative.
- § 6.7 If the Owner chooses to proceed under Section 6.6.4, the Architect shall modify the Construction Documents as necessary to comply with the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services, or the budget as adjusted under Section 6.6.1. If the Owner requires the Architect to modify the Construction Documents because the lowest bona fide bid or negotiated proposal exceeds the Owner's budget for the Cost of the Work due to market conditions the Architect could not reasonably anticipate, the Owner shall compensate the Architect for the modifications as an Additional Service pursuant to Section 11.3; otherwise the Architect's services for modifying the Construction Documents shall be without additional compensation. In any event, the Architect's modification of the Construction Documents shall be the limit of the Architect's responsibility under this Article 6.

#### ARTICLE 7 COPYRIGHTS AND LICENSES

- § 7.1 The Architect and the Owner warrant that in transmitting Instruments of Service, or any other information, the transmitting party is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project.
- § 7.2 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements or for similar purposes in connection with the Project is not to be construed as publication in derogation of the reserved rights of the Architect and the Architect's consultants.
- § 7.3 The Architect grants to the Owner a nonexclusive license to use the Architect's Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations under this Agreement, including prompt payment of all sums due pursuant to Article 9 and Article 11. The Architect shall obtain similar nonexclusive licenses from the Architect's consultants consistent with this Agreement. The license granted under this section permits the Owner to authorize the Contractor, Subcontractors, Sub-subcontractors, and suppliers, as well as the Owner's consultants and separate contractors, to reproduce applicable portions of the Instruments of Service, subject to any protocols established pursuant to Section 1.3, solely and exclusively for use in performing services or construction for the Project. If the Architect rightfully terminates this Agreement for cause as provided in Section 9.4, the license granted in this Section 7.3 shall terminate.
- § 7.3.1 In the event the Owner uses the Instruments of Service without retaining the authors of the Instruments of Service, the Owner releases the Architect and Architect's consultant(s) from all claims and causes of action arising from such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the Architect and its consultants from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's use of the Instruments of Service under this Section 7.3.1. The terms of this Section 7.3.1 shall not apply if the Owner rightfully terminates this Agreement for cause under Section 9.4.
- § 7.4 Except for the licenses granted in this Article 7, no other license or right shall be deemed granted or implied under this Agreement. The Owner shall not assign, delegate, sublicense, pledge or otherwise transfer any license granted herein to another party without the prior written agreement of the Architect. Any unauthorized use of the Instruments of Service shall be at the Owner's sole risk and without liability to the Architect and the Architect's consultants.
- § 7.5 Except as otherwise stated in Section 7.3, the provisions of this Article 7 shall survive the termination of this Agreement.

## ARTICLE 8 CLAIMS AND DISPUTES § 8.1 General

- § 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.
- § 8.1.2 To the extent damages are covered by property insurance, the Owner and Architect waive all rights against each other and against the contractors, consultants, agents, and employees of the other for damages, except such rights as they may have to the proceeds of such insurance as set forth in AIA Document A201–2017, General

Conditions of the Contract for Construction. The Owner or the Architect, as appropriate, shall require of the contractors, consultants, agents, and employees of any of them, similar waivers in favor of the other parties enumerated herein.

§ 8.1.3 The Architect and Owner waive consequential damages for claims, disputes, or other matters in question, arising out of or relating to this Agreement. This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination of this Agreement, except as specifically provided in Section 9.7.

#### § 8.2 Mediation

- § 8.2.1 Any claim, dispute or other matter in question arising out of or related to this Agreement shall be subject to mediation as a condition precedent to binding dispute resolution. If such matter relates to or is the subject of a lien arising out of the Architect's services, the Architect may proceed in accordance with applicable law to comply with the lien notice or filing deadlines prior to resolution of the matter by mediation or by binding dispute resolution.
- § 8.2.2 The Owner and Architect shall endeavor to resolve claims, disputes and other matters in question between them by mediation, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of this Agreement. A request for mediation shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of a complaint or other appropriate demand for binding dispute resolution but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration proceeding is stayed pursuant to this section, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.
- § 8.2.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.
- § 8.2.4 If the parties do not resolve a dispute through mediation pursuant to this Section 8.2, the method of binding dispute resolution shall be the following: (Check the appropriate box.)

| ]   | Arbitration pursuant to Section 8.3 of this Agreement |
|-----|---|
| X ] | Litigation in a court of competent jurisdiction       |
| 1   | Other: (Specify)                                      |

If the Owner and Architect do not select a method of binding dispute resolution, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, the dispute will be resolved in a court of competent jurisdiction.

## § 8.3 Arbitration

- § 8.3.1 If the parties have selected arbitration as the method for binding dispute resolution in this Agreement, any claim, dispute or other matter in question arising out of or related to this Agreement subject to, but not resolved by, mediation shall be subject to arbitration, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of this Agreement. A demand for arbitration shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the arbitration.
- § 8.3.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the claim, dispute or other matter in question would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the

arbitration shall constitute the institution of legal or equitable proceedings based on the claim, dispute or other matter in question.

- § 8.3.2 The foregoing agreement to arbitrate, and other agreements to arbitrate with an additional person or entity duly consented to by parties to this Agreement, shall be specifically enforceable in accordance with applicable law in any court having jurisdiction thereof.
- § 8.3.3 The award rendered by the arbitrator(s) shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

## § 8.3.4 Consolidation or Joinder

- § 8.3.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation; (2) the arbitrations to be consolidated substantially involve common questions of law or fact; and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).
- § 8.3.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.
- § 8.3.4.3 The Owner and Architect grant to any person or entity made a party to an arbitration conducted under this Section 8.3, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Architect under this Agreement.
- § 8.4 The provisions of this Article 8 shall survive the termination of this Agreement.

## ARTICLE 9 TERMINATION OR SUSPENSION

- § 9.1 If the Owner fails to make payments to the Architect in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Architect's option, cause for suspension of performance of services under this Agreement. If the Architect elects to suspend services, the Architect shall give seven days' written notice to the Owner before suspending services. In the event of a suspension of services, the Architect shall have no liability to the Owner for delay or damage caused the Owner because of such suspension of services. Before resuming services, the Owner shall pay the Architect all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.3 If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Architect, the Architect may terminate this Agreement by giving not less than seven days' written notice.
- § 9.4 Either party may terminate this Agreement upon seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.
- § 9.5 The Owner may terminate this Agreement upon seven days' written notice to the Architect for the Owner's convenience and without cause.
- § 9.6 If the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall compensate the Architect for services performed prior to

termination, Reimbursable Expenses incurred, and costs attributable to termination, including the costs attributable to the Architect's termination of consultant agreements.

§ 9.7 In addition to any amounts paid under Section 9.6, if the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall pay to the Architect the following fees:

(Set forth below the amount of any termination or licensing fee, or the method for determining any termination or licensing fee.)

.1 Termination Fee:

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

.2 Licensing Fee if the Owner intends to continue using the Architect's Instruments of Service:

As mutually agreed

- § 9.8 Except as otherwise expressly provided herein, this Agreement shall terminate one year from the date of Substantial Completion.
- § 9.9 The Owner's rights to use the Architect's Instruments of Service in the event of a termination of this Agreement are set forth in Article 7 and Section 9.7.

## ARTICLE 10 MISCELLANEOUS PROVISIONS

- § 10.1 This Agreement shall be governed by the law of the place where the Project is located, excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 8.3.
- § 10.2 Terms in this Agreement shall have the same meaning as those in AIA Document A201–2017, General Conditions of the Contract for Construction.
- § 10.3 The Owner and Architect, respectively, bind themselves, their agents, successors, assigns, and legal representatives to this Agreement. Neither the Owner nor the Architect shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement, including any payments due to the Architect by the Owner prior to the assignment.
- § 10.4 If the Owner requests the Architect to execute certificates, the proposed language of such certificates shall be submitted to the Architect for review at least 14 days prior to the requested dates of execution. If the Owner requests the Architect to execute consents reasonably required to facilitate assignment to a lender, the Architect shall execute all such consents that are consistent with this Agreement, provided the proposed consent is submitted to the Architect for review at least 14 days prior to execution. The Architect shall not be required to execute certificates or consents that would require knowledge, services, or responsibilities beyond the scope of this Agreement.
- § 10.5 Nothing contained in this Agreement shall create a contractual relationship with, or a cause of action in favor of, a third party against either the Owner or Architect.
- § 10.6 Unless otherwise required in this Agreement, the Architect shall have no responsibility for the discovery, presence, handling, removal or disposal of, or exposure of persons to, hazardous materials or toxic substances in any form at the Project site.
- § 10.7 The Architect shall have the right to include photographic or artistic representations of the design of the Project among the Architect's promotional and professional materials. The Architect shall be given reasonable access to the completed Project to make such representations. However, the Architect's materials shall not include the Owner's confidential or proprietary information if the Owner has previously advised the Architect in writing of the specific information considered by the Owner to be confidential or proprietary. The Owner shall provide professional credit for the Architect in the Owner's promotional materials for the Project. This Section 10.7 shall

survive the termination of this Agreement unless the Owner terminates this Agreement for cause pursuant to Section 9.4.

- § 10.8 If the Architect or Owner receives information specifically designated as "confidential" or "business proprietary," the receiving party shall keep such information strictly confidential and shall not disclose it to any other person except as set forth in Section 10.8.1. This Section 10.8 shall survive the termination of this Agreement.
- § 10.8.1 The receiving party may disclose "confidential" or "business proprietary" information after 7 days' notice to the other party, when required by law, arbitrator's order, or court order, including a subpoena or other form of compulsory legal process issued by a court or governmental entity, or to the extent such information is reasonably necessary for the receiving party to defend itself in any dispute. The receiving party may also disclose such information to its employees, consultants, or contractors in order to perform services or work solely and exclusively for the Project, provided those employees, consultants and contractors are subject to the restrictions on the disclosure and use of such information as set forth in this Section 10.8.
- § 10.9 The invalidity of any provision of the Agreement shall not invalidate the Agreement or its remaining provisions. If it is determined that any provision of the Agreement violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Agreement shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Agreement.

## ARTICLE 11 COMPENSATION

§ 11.1 For the Architect's Basic Services described under Article 3, the Owner shall compensate the Architect as follows:

.1 Stipulated Sum (Insert amount)

Compensation in the amount of Twenty Eight Thousand Dollars and Zero Cents (\$28,000.00)

.2 Percentage Basis (Insert percentage value)

( )% of the Owner's budget for the Cost of the Work, as calculated in accordance with Section 11.6.

.3 Other (Describe the method of compensation)

§ 11.2 For the Architect's Supplemental Services designated in Section 4.1.1 and for any Sustainability Services required pursuant to Section 4.1.3, the Owner shall compensate the Architect as follows:

(Insert amount of, or basis for, compensation. If necessary, list specific services to which particular methods of compensation apply.)

Hourly or as mutually agreed upon in advance of services

§ 11.3 For Additional Services that may arise during the course of the Project, including those under Section 4.2, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation.)

Hourly or as mutually agreed upon in advance of services

§ 11.4 Compensation for Supplemental and Additional Services of the Architect's consultants when not included in Section 11.2 or 11.3, shall be the amount invoiced to the Architect plus percent (%), or as follows: (Insert amount of, or basis for computing, Architect's consultants' compensation for Supplemental or Additional Services.)

1

As mutually agreed by Owner and Architect in advance of services

§ 11.5 When compensation for Basic Services is based on a stipulated sum or a percentage basis, the proportion of compensation for each phase of services shall be as follows:

| Total Basic Compensation | one hundred | percent ( | 100 | %) |
|--------------------------|-------------|-----------|-----|----|
| Construction Phase       | Twenty Five | percent ( | 25  | %) |
| Procurement Phase        | Five        | percent ( | 5   | %) |
| Phase                    |             |           |     |    |
| Construction Documents   | Twenty Five | percent ( | 25  | %) |
| Design Development Phase | Twenty Five | percent ( | 25  | %) |
| Schematic Design Phase   | Twenty      | percent ( | 20  | %) |

§ 11.6 When compensation identified in Section 11.1 is on a percentage basis, progress payments for each phase of Basic Services shall be calculated by multiplying the percentages identified in this Article by the Owner's most recent budget for the Cost of the Work. Compensation paid in previous progress payments shall not be adjusted based on subsequent updates to the Owner's budget for the Cost of the Work.

§ 11.6.1 When compensation is on a percentage basis and any portions of the Project are deleted or otherwise not constructed, compensation for those portions of the Project shall be payable to the extent services are performed on those portions. The Architect shall be entitled to compensation in accordance with this Agreement for all services performed whether or not the Construction Phase is commenced.

§ 11.7 The hourly billing rates for services of the Architect and the Architect's consultants are set forth below. The rates shall be adjusted in accordance with the Architect's and Architect's consultants' normal review practices. (If applicable, attach an exhibit of hourly billing rates or insert them below.)

See Exhibit C - Hourly Compensation and Reimbursable Expenses

| Employee or Category | Rate (\$0.00) |
|----------------------|---------------|
| N/A                  | N/A           |
|                      |               |

## § 11.8 Compensation for Reimbursable Expenses

§ 11.8.1 Reimbursable Expenses are in addition to compensation for Basic, Supplemental, and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:

- Transportation and authorized out-of-town travel and subsistence;
- .2 Long distance services, dedicated data and communication services, teleconferences, Project web sites, and extranets;
- .3 Permitting and other fees required by authorities having jurisdiction over the Project;
- .4 Printing, reproductions, plots, and standard form documents:
- .5 Postage, handling, and delivery;
- .6 Expense of overtime work requiring higher than regular rates, if authorized in advance by the Owner;
- .7 Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner or required for the Project;
- .8 If required by the Owner, and with the Owner's prior written approval, the Architect's consultants' expenses of professional liability insurance dedicated exclusively to this Project, or the expense of additional insurance coverage or limits in excess of that normally maintained by the Architect's consultants;
- .9 All taxes levied on professional services and on reimbursable expenses;
- .10 Site office expenses:
- .11 Registration fees and any other fees charged by the Certifying Authority or by other entities as necessary to achieve the Sustainable Objective; and,
- .12 Other similar Project-related expenditures.

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus Ten percent (10 %) of the expenses incurred.

§ 11.9 Architect's Insurance. If the types and limits of coverage required in Section 2.5 are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect for the additional costs incurred by the Architect for the additional coverages as set forth below:

(Insert the additional coverages the Architect is required to obtain in order to satisfy the requirements set forth in Section 2.5, and for which the Owner shall reimburse the Architect.)

N/A

## § 11.10 Payments to the Architect

#### § 11.10.1 Initial Payments

§ 11.10.1.1 An initial payment of Zero (\$0) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of Zero (\$ 0 ) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred.

## § 11.10.2 Progress Payments

§ 11.10.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid Forty-Five (45) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect. (Insert rate of monthly or annual interest agreed upon.)

## 1.5 % Monthly

§ 11.10.2.2 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work, unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.

§ 11.10.2.3 Records of Reimbursable Expenses, expenses pertaining to Supplemental and Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

§ 11.10.2.4 Owner Agrees to receive invoices by email, which will be sent to the following individual and email address:

Full Name: Kelly Christopherson

Email Address: kelly.christopherson@k12.sd.us

## ARTICLE 12 SPECIAL TERMS AND CONDITIONS

Special terms and conditions that modify this Agreement are as follows: (Include other terms and conditions applicable to this Agreement.)

None

#### ARTICLE 13 SCOPE OF THE AGREEMENT

§ 13.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both the Owner and Architect.

§ 13.2 This Agreement is comprised of the following documents identified below:

- .1 AIA Document B101TM 2017, Standard Form Agreement Between Owner and Architect
- .2 AIA Document E203<sup>TM</sup>-2013, Building Information Modeling and Digital Data Exhibit, dated as indicated below:

(Insert the date of the E203-2013 incorporated into this agreement.)

| Exhibit A: Project I<br>Exhibit B: JLG Imag |   |
|---|---|
| Exhibit B: JLG Imag                         |   |
| Exhibit B: JLG Imag                         | e Release   |
| Exhibit A: Project I                        | nitial Information  |
|   | s incorporated into this Agreement: other exhibits incorporated into this Agreement, including any exhibits and scopes of services in Section 4.1.2.) |
|   | t E204 <sup>TM</sup> _2017, Sustainable Projects Exhibit, dated as indicated below: E204-2017 incorporated into this agreement.)                      |
|   | the appropriate box for any exhibits incorporated into this Agreement.)   |

Catherine Dekkenga, AIA, Principal

(Printed name, title, and license number, if required)

Garret Bischoff, President

(Printed name and title)

# Additions and Deletions Report for

AIA® Document B101® - 2017

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined, Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 12:22:54 ET on 10/05/2022.

| PAGE 1   |
|--|
| AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two   |
| iii.   |
| Huron Public School District  150 5th St. SW  Huron, SD 57350  |
| ***  |
| JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104  |
| ***  |
| 21392.01 – Huron School District – TAC A/C Adding A/C coil units in the existing TAC building  |
| PAGE 2   |
| TABLE OF ARTICLES  |
| ***  |
| (Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.) |
| m.   |
| Adding A/C coils unit in the existing TAC building   |
| ***  |
| Existing activity building Owned and operated by the School District   |

# \$260,000.00 PAGE 3 Bidding: Fall 2022 Construction to commence spring summer of 2023 ... N/A \*\*\* N/A Design/Bid/Build N/A Kelly Christopherson Huron Public School District 150 5th St. SW Huron, SD 57350 Telephone: Email: kelly.christopherson@k12.sd.us ... N/A N/A PAGE 4 N/A N/A

Catherine Dekkenga

JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104 Telephone: 605-271-2531 Email: cdekkenga@jlgarchitects.com N/A Associated Consulting Engineers, Inc., Jordan deWit 340 S. Phillips Ave. Sioux Falls, SD 57104 Telephone: 605-335-3720 -Associated Consulting Engineers, Inc., Brad Shoup 340 S. Phillips Ave. Sioux Falls, SD 57104 PAGE 5 Telephone: 605-335-3720 -N/A N/A

§ 2.5.1 Commercial General Liability with policy limits of not less than (\$-One Million Dollars and Zero Cents (\$ 1,000,000.00 ) for each occurrence and  $\underline{\text{Two Million Dollars and Zero Cents}}$  (\$ 2,000,000.00 ) in the aggregate for bodily injury and property damage.

Additions and Deletions Report for AIA Document B101" – 2017. Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects." AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 12:22:54 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents\* Terms of Service. To User Notes:

#### PAGE 6

§ 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of not less than (\$-One Million Dollars and Zero Cents (\$1,000,000.00) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.

\*\*\*

§ 2.5.5 Employers' Liability with policy limits not less than (\$ ) each accident, (\$ Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each accident, Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000.00 ) policy limit.

§ 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of not less than (\$ Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) per claim and Four Million Dollars and Zero Cents (\$ 4,000,000.00 ) in the aggregate.

...

§ 3.1 The Architect's Basic Services consist of those described in this Article 3 and include usual and customary structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.

## PAGE 7

§ 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner alternative approaches approach(cs) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.

§ 3.2.5.2 The Architect shall consider the value of alternative materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.

#### PAGE 9

§ 3.6.2.1 The Architect shall visit the site at intervals appropriate to the stage-points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.

## PAGE 10

§ 3.6.4.2 The Architect shall review and approve, review, or take other appropriate action upon, per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

#### PAGE 11

## § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- 1. The Parties agree and understand that, during the course of the construction of the Work, the Owner. Contractor, or other Project participants may choose to, at their own expense and initiative, make use of supplemental video imagery and information sourced from devices such as project webcams, Unmanned Aerial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain visual information and documentation regarding the state of the progress and/or quality of the Work (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly agree and acknowledge the following:
  - Since Supplementary Video Information encompasses only certain limited visual information
    and provides neither complete nor comprehensive visual information/measurements/dimensions,
    nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video
    Information shall not be used as a substitute for or required supplement to Architect's Construction
    Phase services, herein, but rather may, to the extent provided and made available to Architect in a
    reasonably timely, legible, and complete fashion, supplement the other data and information
    obtained by Architect pursuant to Architect's Construction Phase Services, herein.
  - Architect shall have no obligation to conduct a comprehensive or continuous review of any
    Supplementary Video Information, but rather may, at its sole discretion, choose to review such
    Supplementary Video Information, or any portion thereof, in connection with the performance of
    its Construction Phase Services on the Project.

#### PAGE 12

3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part

## ARTICLE thereof.

## ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

## PAGE 13

| § 4.1.1.1 | Programming  | Not Provided |
|-----------|--|--------------|
| § 4.1.1.2 | Multiple preliminary designs                           | Not Provided |
| § 4.1.1.3 | Measured drawings                                      | Not Provided |
| § 4.1.1.4 | Existing facilities surveys                            | Not Provided |
| § 4.1.1.5 | Site evaluation and planning                           | Not Provided |
| § 4.1.1.6 | Building Information Model management responsibilities | Not Provided |

Additions and Deletions Report for AIA Document B101" – 2017. Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects." "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may no be used without permission. This document was produced by AIA software at 12:22:54 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org.

User Notes:

| § 4.1.1.7  | Development of Building Information Models for post construction use | Not Provided |
|------------|--|--------------|
| § 4.1.1.8  | Civil engineering  | Not Provided |
| § 4.1.1.9  | Landscape design   | Not Provided |
| § 4.1.1.10 | Architectural interior design  | Not Provided |
| § 4.1.1.11 | Value analysis   | Not Provided |
| § 4.1.1.12 | Detailed cost estimating beyond that required in Section 6.3         | Not Provided |
| § 4.1.1.13 | On-site project representation                                       | Not Provided |
| § 4.1.1.14 | Conformed documents for construction                                 | Not Provided |
| § 4.1.1.15 | As-designed record drawings  | Not Provided |
| § 4.1.1.16 | As-constructed record drawings                                       | Not Provided |
| § 4.1.1.17 | Post-occupancy evaluation  | Not Provided |
| § 4.1.1.18 | Facility support services  | Not Provided |
| § 4.1.1.19 | Tenant-related services  | Not Provided |
| § 4.1.1.20 | Architect's coordination of the Owner's consultants                  | Not Provided |
| § 4.1.1.21 | Telecommunications/data design                                       | Not Provided |
| § 4.1.1.22 | Security evaluation and planning                                     | Not Provided |
| § 4.1.1.23 | Commissioning  | Not Provided |
| § 4.1.1.24 | Sustainable Project Services pursuant to Section 4.1.3               | Not Provided |
| § 4.1.1.25 | Fast-track design services   | Not Provided |
| § 4.1.1.26 | Multiple bid packages  | Not Provided |
| § 4.1.1.27 | Historic preservation  | Not Provided |
| § 4.1.1.28 | Furniture, furnishings, and equipment design                         | Not Provided |
| § 4.1.1.29 | Other services provided by specialty Consultants                     | Not Provided |
| § 4.1.1.30 | Other Supplemental Services  | Not Provided |

N/A

...

N/A

\*\*\*

\*\*\*

# PAGE 14

.1 <u>Two (2 )</u> reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor

.2 (+)Bi-weekly visits to the site by the Architect during construction

.3 One (1 ) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents

.4 One (1 ) inspections for any portion of the Work to determine final completion.

Additions and Deletions Report for AIA Document B101" – 2017. Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 12:22:54 ET on 10/05/2022 under Order No.2114257999 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service, To report copyright violations, e-mail copyright@aia.org.

(389ADA42)

§ 4.2.5 If the services covered by this Agreement have not been completed within Twenty Four (24) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

#### PAGE 15

§ 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1; 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes: (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.

#### PAGE 17

§ 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.

#### PAGE 18

[X] Litigation in a court of competent jurisdiction

#### PAGE 19

§ 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

§ 9.4 Either party may terminate this Agreement upon not less than seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.

§ 9.5 The Owner may terminate this Agreement upon not less than seven days' written notice to the Architect for the Owner's convenience and without cause.

|    | - |    |    |
|----|---|----|----|
| PΑ |   | _  | 20 |
| PA |   | е. | 11 |

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

As mutually agreed

#### PAGE 21

Compensation in the amount of Twenty Eight Thousand Dollars and Zero Cents (\$28,000.00)

Hourly or as mutually agreed upon in advance of services

Hourly or as mutually agreed upon in advance of services

#### PAGE 22

As mutually agreed by Owner and Architect in advance of services

| Schematic Design Phase   | Twenty      | percent (                              | 20 | %)   |
|--------------------------|-------------|--|----|------|
| Design Development Phase | Twenty Five | percent (                              | 25 | %)   |
| Construction Documents   | Twenty Five | percent (                              | 25 | %)   |
| Phase                    |             | ************************************** |    | 1000 |
| Procurement Phase        | Five        | percent (                              | 5  | %)   |
| Construction Phase       | Twenty Five | percent (                              | 25 | %)   |

See Exhibit C - Hourly Compensation and Reimbursable Expenses

<u>N/A</u> PAGE 23

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus Ten\_percent (10\_%) of the expenses incurred.

N/A

....

§ 11.10.1.1 An initial payment of Zero (\$ 0 ) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

| § 11.10.1.2 If a Sustainability Certification is part of the Sustainable Obj<br>Zero_(\$ 0_) shall be made upon execution of this Agreement for regis<br>Certifying Authority and necessary to achieve the Sustainability Certifi<br>Certifying Authority shall be credited to the Owner's account at the time | tration fees and other fees payable to the<br>cation. The Architect's payments to the     |
|--|---|
| ***  |   |
| § 11.10.2.1 Unless otherwise agreed, payments for services shall be mad performed. Payments are due and payable upon presentation of the Arc (45_) days after the invoice date shall bear interest at the rate entered trate prevailing from time to time at the principal place of business of the            | hitect's invoice. Amounts unpaid Forty-Five below, or in the absence thereof at the legal |
|  |   |
| 1.5 % Monthly  |   |
|  |   |
| § 11.10.2.4 Owner Agrees to receive invoices by email, which will be seaddress;  | ent to the following individual and email   |
|  |   |
| Full Name: Kelly Christopherson<br>Email Address: kelly.christopherson@k12.sd.us   |   |
|  |   |
| None   |   |
| PAGE 24  |   |
| Exhibit A: Project Initial Information   |   |
|  |   |
| Exhibit B: JLG Image Release   |   |
| ***  |   |
| Exhibit C: Schedule of Hourly Compensation & Reimbursable Expense  | es  |
| ***  |   |
| Garret Bischoff, President Catherine   | Dekkenga, AIA, Principal  |
|  |   |

# Certification of Document's Authenticity

AIA® Document D401™ - 2003

I, Catherine Dekkenga, AIA, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 12:22:54 ET on 10/05/2022 under Order No. 2114257989 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA\* Document B101<sup>TM</sup> - 2017, Standard Form of Agreement Between Owner and Architect, as published by the AIA in its software, other than those additions and deletions shown in the associated Additions and Deletions Report.

| (Signed) |  |
|----------|--|
|          |  |
| (Title)  |  |
| (Dated)  |  |



# Standard Form of Agreement Between Owner and Architect

AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two (In words, indicate day, month and year.)

BETWEEN the Architect's client identified as the Owner: (Name, legal status, address and other information)

Huron Public School District 150 5th St. SW Huron, SD 57350

and the Architect: (Name, legal status, address and other information)

JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104

for the following Project: (Name, location and detailed description)

21392.01 – Huron School District – Track Resurfacing Resurfacing existing thrack

The Owner and Architect agree as follows.

#### ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

#### TABLE OF ARTICLES

- 1 INITIAL INFORMATION
- 2 ARCHITECT'S RESPONSIBILITIES
- 3 SCOPE OF ARCHITECT'S BASIC SERVICES
- 4 SUPPLEMENTAL AND ADDITIONAL SERVICES
- 5 OWNER'S RESPONSIBILITIES
- 6 COST OF THE WORK
- 7 COPYRIGHTS AND LICENSES
- 8 CLAIMS AND DISPUTES
- 9 TERMINATION OR SUSPENSION
- 10 MISCELLANEOUS PROVISIONS
- 11 COMPENSATION
- 12 SPECIAL TERMS AND CONDITIONS
- 13 SCOPE OF THE AGREEMENT

#### ARTICLE 1 INITIAL INFORMATION

§ 1.1 This Agreement is based on the Initial Information set forth in this Section 1.1.

(For each item in this section, insert the information or a statement such as "not applicable" or "unknown at time of execution.")

#### § 1.1.1 The Owner's program for the Project:

(Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.)

Resurfacing existing track

# § 1.1.2 The Project's physical characteristics:

(Identify or describe pertinent information about the Project's physical characteristics, such as size; location; dimensions; geotechnical reports; site boundaries; topographic surveys; traffic and utility studies; availability of public and private utilities and services; legal description of the site, etc.)

Existing track owned and operated by the School District

§ 1.1.3 The Owner's budget for the Cost of the Work, as defined in Section 6.1: (Provide total and, if known, a line item breakdown.)

\$250,000.00 - \$525,000.00

- § 1.1.4 The Owner's anticipated design and construction milestone dates:
  - .1 Design phase milestone dates, if any:

Bidding: Late fall 2022 Construction to Commence spring/summer 2023

.2 Construction commencement date:

N/A

.3 Substantial Completion date or dates:

N/A

.4 Other milestone dates:

N/A

§ 1.1.5 The Owner intends the following procurement and delivery method for the Project: (Identify method such as competitive bid or negotiated contract, as well as any requirements for accelerated or fast-track design and construction, multiple bid packages, or phased construction.)

Design/Bid/Build

§ 1.1.6 The Owner's anticipated Sustainable Objective for the Project: (Identify and describe the Owner's Sustainable Objective for the Project, if any.)

N/A (Paragraph Deleted)

§ 1.1.6.1 If the Owner identifies a Sustainable Objective, the Owner and Architect shall complete and incorporate AIA Document E204<sup>TM</sup>\_2017, Sustainable Projects Exhibit, into this Agreement to define the terms, conditions and services related to the Owner's Sustainable Objective. If E204\_2017 is incorporated into this agreement, the Owner and Architect shall incorporate the completed E204\_2017 into the agreements with the consultants and contractors performing services or Work in any way associated with the Sustainable Objective.

§ 1.1.7 The Owner identifies the following representative in accordance with Section 5.3: (List name, address, and other contact information.)

Kelly Christopheron Huron Public School District 150 5th St. SW Huron, SD 57350 Telephone:

Email: kelly.christopherson@k12.sd.us

§ 1.1.8 The persons or entities, in addition to the Owner's representative, who are required to review the Architect's submittals to the Owner are as follows:

(List name, address, and other contact information.)

N/A

Init.

§ 1.1.9 The Owner shall retain the following consultants and contractors: (List name, legal status, address, and other contact information.)

| .1 Geotech   | nical Engineer:   |                   |
|--|---|-------------------|
| N/A  |   |                   |
|  |   |                   |
| Name of the Control o |   |                   |
| (Paragraph Deleted)  |   |                   |
|  |   |                   |
|  |   |                   |
| .3 Other, if (List any of  | any:<br>Cany: any: and contractors retained by the Owner.   | )                 |
| N/A  |   |                   |
|  | t identifies the following representative in accordance wand other contact information.)                  | ith Section 2.3:  |
| Catherine Dekkenga<br>JLG Architects   |   |                   |
| 232 S. Main Ave.   |   |                   |
| Sioux Falls, SD 5710   | 4   |                   |
| Telephone: 605-271-2<br>Email: cdekkenga@jl  |   |                   |
|  | t shall retain the consultants identified in Sections 1.1.11 us, address, and other contact information.) | 1.1 and 1.1.11.2: |
| § 1.1.11.1 Consultants<br>.1 Structura   | retained under Basic Services:<br>al Engineer:  |                   |
| N/A  |   |                   |
|  |   |                   |
| .2 Mechani   | cal Engineer:   |                   |
| N/A  |   |                   |
|  |   |                   |
| .3 Electrica   | l Engineer:   |                   |
| N/A  |   |                   |

#### .2 Civil Engineer:

Stockwell Engineers, Inc. 801 N. Phillips Ave. Suite 100 Sioux Falls, SD 57104 Telephone: 605-338-6668

§ 1.1.11.2 Consultants retained under Supplemental Services:

N/A

§ 1.1.12 Other Initial Information on which the Agreement is based:

N/A

- § 1.2 The Owner and Architect may rely on the Initial Information. Both parties, however, recognize that the Initial Information may materially change and, in that event, the Owner and the Architect shall appropriately adjust the Architect's services, schedule for the Architect's services, and the Architect's compensation. The Owner shall adjust the Owner's budget for the Cost of the Work and the Owner's anticipated design and construction milestones, as necessary, to accommodate material changes in the Initial Information.
- § 1.3 The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. The parties will use AIA Document E203<sup>TM</sup>-2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.
- § 1.3.1 Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in AIA Document E203<sup>TM</sup>–2013, Building Information Modeling and Digital Data Exhibit, and the requisite AIA Document G202<sup>TM</sup>–2013, Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

## ARTICLE 2 ARCHITECT'S RESPONSIBILITIES

- § 2.1 The Architect shall provide professional services as set forth in this Agreement. The Architect represents that it is properly licensed in the jurisdiction where the Project is located to provide the services required by this Agreement, or shall cause such services to be performed by appropriately licensed design professionals.
- § 2.2 The Architect shall perform its services consistent with the professional skill and care ordinarily provided by architects practicing in the same or similar locality under the same or similar circumstances. The Architect shall perform its services as expeditiously as is consistent with such professional skill and care and the orderly progress of the Project.
- § 2.3 The Architect shall identify a representative authorized to act on behalf of the Architect with respect to the Project.
- § 2.4 Except with the Owner's knowledge and consent, the Architect shall not engage in any activity, or accept any employment, interest or contribution that would reasonably appear to compromise the Architect's professional judgment with respect to this Project.

- § 2.5 The Architect shall maintain the following insurance until termination of this Agreement. If any of the requirements set forth below are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect as set forth in Section 11.9.
- § 2.5.1 Commercial General Liability with policy limits of One Million Dollars and Zero Cents (\$ 1,000,000.00 ) for each occurrence and Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) in the aggregate for bodily injury and property damage.
- § 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of One Million Dollars and Zero Cents (\$ 1,000,000.00 ) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.
- § 2.5.3 The Architect may achieve the required limits and coverage for Commercial General Liability and Automobile Liability through a combination of primary and excess or umbrella liability insurance, provided such primary and excess or umbrella liability insurance policies result in the same or greater coverage as the coverages required under Sections 2.5.1 and 2.5.2, and in no event shall any excess or umbrella liability insurance provide narrower coverage than the primary policy. The excess policy shall not require the exhaustion of the underlying limits only through the actual payment by the underlying insurers.
- § 2.5.4 Workers' Compensation at statutory limits.
- § 2.5.5 Employers' Liability with policy limits Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00) each accident, Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000.00) policy limit.
- § 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) per claim and Four Million Dollars and Zero Cents (\$ 4,000,000.00 ) in the aggregate.
- § 2.5.7 Additional Insured Obligations. To the fullest extent permitted by law, the Architect shall cause the primary and excess or umbrella polices for Commercial General Liability and Automobile Liability to include the Owner as an additional insured for claims caused in whole or in part by the Architect's negligent acts or omissions. The additional insured coverage shall be primary and non-contributory to any of the Owner's insurance policies and shall apply to both ongoing and completed operations.
- § 2.5.8 The Architect shall provide certificates of insurance to the Owner that evidence compliance with the requirements in this Section 2.5.

#### ARTICLE 3 SCOPE OF ARCHITECT'S BASIC SERVICES

- § 3.1 The Architect's Basic Services consist of those described in this Article 3 and include structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.
- § 3.1.1 The Architect shall manage the Architect's services, research applicable design criteria, attend Project meetings, communicate with members of the Project team, and report progress to the Owner.
- § 3.1.2 The Architect shall coordinate its services with those services provided by the Owner and the Owner's consultants. The Architect shall be entitled to rely on, and shall not be responsible for, the accuracy, completeness, and timeliness of, services and information furnished by the Owner and the Owner's consultants. The Architect shall provide prompt written notice to the Owner if the Architect becomes aware of any error, omission, or inconsistency in such services or information.
- § 3.1.3 As soon as practicable after the date of this Agreement, the Architect shall submit for the Owner's approval a schedule for the performance of the Architect's services. The schedule initially shall include anticipated dates for the commencement of construction and for Substantial Completion of the Work as set forth in the Initial Information. The schedule shall include allowances for periods of time required for the Owner's review, for the performance of the Owner's consultants, and for approval of submissions by authorities having jurisdiction over the Project. Once

approved by the Owner, time limits established by the schedule shall not, except for reasonable cause, be exceeded by the Architect or Owner. With the Owner's approval, the Architect shall adjust the schedule, if necessary, as the Project proceeds until the commencement of construction.

- § 3.1.4 The Architect shall not be responsible for an Owner's directive or substitution, or for the Owner's acceptance of non-conforming Work, made or given without the Architect's written approval.
- § 3.1.5 The Architect shall contact governmental authorities required to approve the Construction Documents and entities providing utility services to the Project. The Architect shall respond to applicable design requirements imposed by those authorities and entities.
- § 3.1.6 The Architect shall assist the Owner in connection with the Owner's responsibility for filing documents required for the approval of governmental authorities having jurisdiction over the Project.

# § 3.2 Schematic Design Phase Services

- § 3.2.1 The Architect shall review the program and other information furnished by the Owner, and shall review laws, codes, and regulations applicable to the Architect's services.
- § 3.2.2 The Architect shall prepare a preliminary evaluation of the Owner's program, schedule, budget for the Cost of the Work, Project site, the proposed procurement and delivery method, and other Initial Information, each in terms of the other, to ascertain the requirements of the Project. The Architect shall notify the Owner of (1) any inconsistencies discovered in the information, and (2) other information or consulting services that may be reasonably needed for the Project.
- § 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner approach(es) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.
- § 3.2.4 Based on the Project requirements agreed upon with the Owner, the Architect shall prepare and present, for the Owner's approval, a preliminary design illustrating the scale and relationship of the Project components.
- § 3.2.5 Based on the Owner's approval of the preliminary design, the Architect shall prepare Schematic Design Documents for the Owner's approval. The Schematic Design Documents shall consist of drawings and other documents including a site plan, if appropriate, and preliminary building plans, sections and elevations; and may include some combination of study models, perspective sketches, or digital representations. Preliminary selections of major building systems and construction materials shall be noted on the drawings or described in writing.
- § 3.2.5.1 The Architect shall consider sustainable design alternatives, such as material choices and building orientation, together with other considerations based on program and aesthetics, in developing a design that is consistent with the Owner's program, schedule and budget for the Cost of the Work. The Owner may obtain more advanced sustainable design services as a Supplemental Service under Section 4.1.1.
- § 3.2.5.2 The Architect shall consider the value of materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.
- § 3.2.6 The Architect shall submit to the Owner an estimate of the Cost of the Work prepared in accordance with Section 6.3.
- § 3.2.7 The Architect shall submit the Schematic Design Documents to the Owner, and request the Owner's approval.

# § 3.3 Design Development Phase Services

§ 3.3.1 Based on the Owner's approval of the Schematic Design Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Design Development Documents for the Owner's approval. The Design Development Documents shall illustrate and describe the development of the approved Schematic Design Documents and shall consist of drawings and other

documents including plans, sections, elevations, typical construction details, and diagrammatic layouts of building systems to fix and describe the size and character of the Project as to architectural, structural, mechanical and electrical systems, and other appropriate elements. The Design Development Documents shall also include outline specifications that identify major materials and systems and establish, in general, their quality levels.

- § 3.3.2 The Architect shall update the estimate of the Cost of the Work prepared in accordance with Section 6.3.
- § 3.3.3 The Architect shall submit the Design Development Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, and request the Owner's approval.

#### § 3.4 Construction Documents Phase Services

- § 3.4.1 Based on the Owner's approval of the Design Development Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Construction Documents for the Owner's approval. The Construction Documents shall illustrate and describe the further development of the approved Design Development Documents and shall consist of Drawings and Specifications setting forth in detail the quality levels and performance criteria of materials and systems and other requirements for the construction of the Work. The Owner and Architect acknowledge that, in order to perform the Work, the Contractor will provide additional information, including Shop Drawings, Product Data, Samples and other similar submittals, which the Architect shall review in accordance with Section 3.6.4.
- § 3.4.2 The Architect shall incorporate the design requirements of governmental authorities having jurisdiction over the Project into the Construction Documents.
- § 3.4.3 During the development of the Construction Documents, the Architect shall assist the Owner in the development and preparation of (1) procurement information that describes the time, place, and conditions of bidding, including bidding or proposal forms; (2) the form of agreement between the Owner and Contractor; and (3) the Conditions of the Contract for Construction (General, Supplementary and other Conditions). The Architect shall also compile a project manual that includes the Conditions of the Contract for Construction and Specifications, and may include bidding requirements and sample forms.
- § 3.4.4 The Architect shall update the estimate for the Cost of the Work prepared in accordance with Section 6.3.
- § 3.4.5 The Architect shall submit the Construction Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, take any action required under Section 6.5, and request the Owner's approval.

#### § 3.5 Procurement Phase Services

#### § 3.5.1 General

The Architect shall assist the Owner in establishing a list of prospective contractors. Following the Owner's approval of the Construction Documents, the Architect shall assist the Owner in (1) obtaining either competitive bids or negotiated proposals; (2) confirming responsiveness of bids or proposals; (3) determining the successful bid or proposal, if any; and, (4) awarding and preparing contracts for construction.

# § 3.5.2 Competitive Bidding

- § 3.5.2.1 Bidding Documents shall consist of bidding requirements and proposed Contract Documents.
- § 3.5.2.2 The Architect shall assist the Owner in bidding the Project by:
  - .1 facilitating the distribution of Bidding Documents to prospective bidders;
  - .2 organizing and conducting a pre-bid conference for prospective bidders;
  - .3 preparing responses to questions from prospective bidders and providing clarifications and interpretations of the Bidding Documents to the prospective bidders in the form of addenda; and,
  - .4 organizing and conducting the opening of the bids, and subsequently documenting and distributing the bidding results, as directed by the Owner.
- § 3.5.2.3 If the Bidding Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective bidders.

# § 3.5.3 Negotiated Proposals

- § 3.5.3.1 Proposal Documents shall consist of proposal requirements and proposed Contract Documents.
- § 3.5.3.2 The Architect shall assist the Owner in obtaining proposals by:
  - .1 facilitating the distribution of Proposal Documents for distribution to prospective contractors and requesting their return upon completion of the negotiation process;
  - .2 organizing and participating in selection interviews with prospective contractors;
  - .3 preparing responses to questions from prospective contractors and providing clarifications and interpretations of the Proposal Documents to the prospective contractors in the form of addenda; and,
  - .4 participating in negotiations with prospective contractors, and subsequently preparing a summary report of the negotiation results, as directed by the Owner.
- § 3.5.3.3 If the Proposal Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective contractors.

#### § 3.6 Construction Phase Services

#### § 3.6.1 General

- § 3.6.1.1 The Architect shall provide administration of the Contract between the Owner and the Contractor as set forth below and in AIA Document A201<sup>TM</sup>—2017, General Conditions of the Contract for Construction. If the Owner and Contractor modify AIA Document A201–2017, those modifications shall not affect the Architect's services under this Agreement unless the Owner and the Architect amend this Agreement.
- § 3.6.1.2 The Architect shall advise and consult with the Owner during the Construction Phase Services. The Architect shall have authority to act on behalf of the Owner only to the extent provided in this Agreement. The Architect shall not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, nor shall the Architect be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect shall be responsible for the Architect's negligent acts or omissions, but shall not have control over or charge of, and shall not be responsible for, acts or omissions of the Contractor or of any other persons or entities performing portions of the Work.
- § 3.6.1.3 Subject to Section 4.2 and except as provided in Section 3.6.6.5, the Architect's responsibility to provide Construction Phase Services commences with the award of the Contract for Construction and terminates on the date the Architect issues the final Certificate for Payment.

# § 3.6.2 Evaluations of the Work

- § 3.6.2.1 The Architect shall visit the site at points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.
- § 3.6.2.2 The Architect has the authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect shall have the authority to require inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not the Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, suppliers, their agents or employees, or other persons or entities performing portions of the Work.

- § 3.6.2.3 The Architect shall interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests shall be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- § 3.6.2.4 Interpretations and decisions of the Architect shall be consistent with the intent of, and reasonably inferable from, the Contract Documents and shall be in writing or in the form of drawings. When making such interpretations and decisions, the Architect shall endeavor to secure faithful performance by both Owner and Contractor, shall not show partiality to either, and shall not be liable for results of interpretations or decisions rendered in good faith. The Architect's decisions on matters relating to aesthetic effect shall be final if consistent with the intent expressed in the Contract Documents.
- § 3.6.2.5 Unless the Owner and Contractor designate another person to serve as an Initial Decision Maker, as that term is defined in AIA Document A201–2017, the Architect shall render initial decisions on Claims between the Owner and Contractor as provided in the Contract Documents.

# § 3.6.3 Certificates for Payment to Contractor

- § 3.6.3.1 The Architect shall review and certify the amounts due the Contractor and shall issue certificates in such amounts. The Architect's certification for payment shall constitute a representation to the Owner, based on the Architect's evaluation of the Work as provided in Section 3.6.2 and on the data comprising the Contractor's Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified. The foregoing representations are subject to (1) an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, (2) results of subsequent tests and inspections, (3) correction of minor deviations from the Contract Documents prior to completion, and (4) specific qualifications expressed by the Architect.
- § 3.6.3.2 The issuance of a Certificate for Payment shall not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) ascertained how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.
- § 3.6.3.3 The Architect shall maintain a record of the Applications and Certificates for Payment.

#### § 3.6.4 Submittals

- § 3.6.4.1 The Architect shall review the Contractor's submittal schedule and shall not unreasonably delay or withhold approval of the schedule. The Architect's action in reviewing submittals shall be taken in accordance with the approved submittal schedule or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time, in the Architect's professional judgment, to permit adequate review.
- § 3.6.4.2 The Architect shall review, or take other appropriate action per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- § 3.6.4.3 If the Contract Documents specifically require the Contractor to provide professional design services or certifications by a design professional related to systems, materials, or equipment, the Architect shall specify the appropriate performance and design criteria that such services must satisfy. The Architect shall review and take appropriate action on Shop Drawings and other submittals related to the Work designed or certified by the Contractor's design professional, provided the submittals bear such professional's seal and signature when submitted to the Architect. The Architect's review shall be for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect shall be entitled to

rely upon, and shall not be responsible for, the adequacy and accuracy of the services, certifications, and approvals performed or provided by such design professionals.

§ 3.6.4.4 Subject to Section 4.2, the Architect shall review and respond to requests for information about the Contract Documents. The Architect shall set forth, in the Contract Documents, the requirements for requests for information. Requests for information shall include, at a minimum, a detailed written statement that indicates the specific Drawings or Specifications in need of clarification and the nature of the clarification requested. The Architect's response to such requests shall be made in writing within any time limits agreed upon, or otherwise with reasonable promptness. If appropriate, the Architect shall prepare and issue supplemental Drawings and Specifications in response to the requests for information.

§ 3.6.4.5 The Architect shall maintain a record of submittals and copies of submittals supplied by the Contractor in accordance with the requirements of the Contract Documents.

#### § 3.6.5 Changes in the Work

§ 3.6.5.1 The Architect may order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. Subject to Section 4.2, the Architect shall prepare Change Orders and Construction Change Directives for the Owner's approval and execution in accordance with the Contract Documents.

§ 3.6.5.2 The Architect shall maintain records relative to changes in the Work.

#### § 3.6.6 Project Completion

§ 3.6.6.1 The Architect shall:

- .1 conduct inspections to determine the date or dates of Substantial Completion and the date of final completion;
- .2 issue Certificates of Substantial Completion;
- .3 forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract Documents and received from the Contractor; and,
- .4 issue a final Certificate for Payment based upon a final inspection indicating that, to the best of the Architect's knowledge, information, and belief, the Work complies with the requirements of the Contract Documents.
- § 3.6.6.2 The Architect's inspections shall be conducted with the Owner to check conformance of the Work with the requirements of the Contract Documents and to verify the accuracy and completeness of the list submitted by the Contractor of Work to be completed or corrected.
- § 3.6.6.3 When Substantial Completion has been achieved, the Architect shall inform the Owner about the balance of the Contract Sum remaining to be paid the Contractor, including the amount to be retained from the Contract Sum, if any, for final completion or correction of the Work.
- § 3.6.4 The Architect shall forward to the Owner the following information received from the Contractor: (1) consent of surety or sureties, if any, to reduction in or partial release of retainage or the making of final payment; (2) affidavits, receipts, releases and waivers of liens, or bonds indemnifying the Owner against liens; and (3) any other documentation required of the Contractor under the Contract Documents.
- § 3.6.6.5 Upon request of the Owner, and prior to the expiration of one year from the date of Substantial Completion, the Architect shall, without additional compensation, conduct a meeting with the Owner to review the facility operations and performance.

# § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

 The Parties agree and understand that, during the course of the construction of the Work, the Owner, Contractor, or other Project participants may choose to, at their own expense and initiative, make use of supplemental video imagery and information sourced from devices such as project webcams, Unmanned Aerial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain visual information and documentation regarding the state of the progress and/or quality of the Work (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly agree and acknowledge the following:

1. Since Supplementary Video Information encompasses only certain limited visual information and provides neither complete nor comprehensive visual information/measurements/dimensions, nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video Information shall not be used as a substitute for or required supplement to Architect's Construction Phase services, herein, but rather may, to the extent provided and made available to Architect in a reasonably timely, legible, and complete fashion, supplement the other data and information obtained by Architect pursuant to Architect's Construction Phase Services, herein.

 Architect shall have no obligation to conduct a comprehensive or continuous review of any Supplementary Video Information, but rather may, at its sole discretion, choose to review such Supplementary Video Information, or any portion thereof, in connection with the performance of

its Construction Phase Services on the Project.

3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part thereof.

# ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

#### § 4.1 Supplemental Services

§ 4.1.1 The services listed below are not included in Basic Services but may be required for the Project. The Architect shall provide the listed Supplemental Services only if specifically designated in the table below as the Architect's responsibility, and the Owner shall compensate the Architect as provided in Section 11.2. Unless otherwise specifically addressed in this Agreement, if neither the Owner nor the Architect is designated, the parties agree that the listed Supplemental Service is not being provided for the Project.

(Designate the Architect's Supplemental Services and the Owner's Supplemental Services required for the Project by indicating whether the Architect or Owner shall be responsible for providing the identified Supplemental Service. Insert a description of the Supplemental Services in Section 4.1.2 below or attach the description of services as an exhibit to this Agreement.)

|            | ental Services   | Responsibility<br>(Architect, Owner, or not<br>provided) |
|------------|--|--|
| § 4.1.1.1  | Programming  | Not Provided   |
| § 4.1.1.2  | Multiple preliminary designs   | Not Provided   |
| § 4.1.1.3  | Measured drawings  | Not Provided   |
| § 4.1.1.4  | Existing facilities surveys  | Not Provided   |
| § 4.1.1.5  | Site evaluation and planning   | Not Provided   |
| § 4.1.1.6  | Building Information Model management responsibilities               | Not Provided   |
| § 4.1.1.7  | Development of Building Information Models for post construction use | Not Provided   |
|            | Civil engineering  | Architect  |
| § 4.1.1.9  | Landscape design   | Architect  |
| § 4.1.1.10 | Architectural interior design  | Not Provided   |
| § 4.1.1.11 | Value analysis   | Not Provided   |
| § 4.1.1.12 | Detailed cost estimating beyond that required in Section 6.3         | Not Provided   |
| § 4.1.1.13 | On-site project representation                                       | Not Provided   |
| § 4.1.1.14 | Conformed documents for construction                                 | Not Provided   |
| § 4.1.1.15 | As-designed record drawings  | Not Provided   |
| § 4.1.1.16 | As-constructed record drawings                                       | Not Provided Not Provided                                |

| § 4.1.1.17 | Post-occupancy evaluation                              | Not Provided |
|------------|--|--------------|
| § 4.1.1.18 | Facility support services                              | Not Provided |
| § 4.1.1.19 | Tenant-related services                                | Not Provided |
| § 4.1.1.20 | Architect's coordination of the Owner's consultants    | Not Provided |
| § 4.1.1.21 | Telecommunications/data design                         | Not Provided |
| § 4.1.1.22 | Security evaluation and planning                       | Not Provided |
| § 4.1.1.23 | Commissioning  | Not Provided |
| § 4.1.1.24 | Sustainable Project Services pursuant to Section 4.1.3 | Not Provided |
| § 4.1.1.25 | Fast-track design services                             | Not Provided |
| § 4.1.1.26 | Multiple bid packages                                  | Not Provided |
| § 4.1.1.27 | Historic preservation                                  | Not Provided |
| § 4.1.1.28 | Furniture, furnishings, and equipment design           | Not Provided |
| § 4.1.1.29 | Other services provided by specialty Consultants       | Not Provided |
| § 4.1.1.30 | Other Supplemental Services                            | Not Provided |

#### § 4.1.2 Description of Supplemental Services

§ 4.1.2.1 A description of each Supplemental Service identified in Section 4.1.1 as the Architect's responsibility is provided below.

(Describe in detail the Architect's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit. The AIA publishes a number of Standard Form of Architect's Services documents that can be included as an exhibit to describe the Architect's Supplemental Services.)

SS 4.1.1.8 Civil Engineering: Limited to usual and customary civil engineering services

SS 4.1.1.9 Landscape design: Limited to Schematic design phase services to indicate design intent and provide basis for cost allowance and bid alternates; based on accepted scope/alternates, remaining documentation will be provided design/build by landscaping subcontractor (Architect to review and approve final drawings for conformance with design intent and review shop drawings)

§ 4.1.2.2 A description of each Supplemental Service identified in Section 4.1.1 as the Owner's responsibility is provided below.

(Describe in detail the Owner's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit.)

N/A

§ 4.1.3 If the Owner identified a Sustainable Objective in Article 1, the Architect shall provide, as a Supplemental Service, the Sustainability Services required in AIA Document E204™\_2017, Sustainable Projects Exhibit, attached to this Agreement. The Owner shall compensate the Architect as provided in Section 11.2.

#### § 4.2 Architect's Additional Services

The Architect may provide Additional Services after execution of this Agreement without invalidating the Agreement. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 4.2 shall entitle the Architect to compensation pursuant to Section 11.3 and an appropriate adjustment in the Architect's schedule.

- § 4.2.1 Upon recognizing the need to perform the following Additional Services, the Architect shall notify the Owner with reasonable promptness and explain the facts and circumstances giving rise to the need. The Architect shall not proceed to provide the following Additional Services until the Architect receives the Owner's written authorization:
  - 1 Services necessitated by a change in the Initial Information, previous instructions or approvals given by the Owner, or a material change in the Project including size, quality, complexity, the Owner's schedule or budget for Cost of the Work, or procurement or delivery method;

- .2 Services necessitated by the enactment or revision of codes, laws, or regulations, including changing or editing previously prepared Instruments of Service;
- .3 Changing or editing previously prepared Instruments of Service necessitated by official interpretations of applicable codes, laws or regulations that are either (a) contrary to specific interpretations by the applicable authorities having jurisdiction made prior to the issuance of the building permit, or (b) contrary to requirements of the Instruments of Service when those Instruments of Service were prepared in accordance with the applicable standard of care;
- .4 Services necessitated by decisions of the Owner not rendered in a timely manner or any other failure of performance on the part of the Owner or the Owner's consultants or contractors;
- .5 Preparing digital models or other design documentation for transmission to the Owner's consultants and contractors, or to other Owner-authorized recipients;
- .6 Preparation of design and documentation for alternate bid or proposal requests proposed by the Owner;
- .7 Preparation for, and attendance at, a public presentation, meeting or hearing;
- .8 Preparation for, and attendance at, a dispute resolution proceeding or legal proceeding, except where the Architect is party thereto;
- .9 Evaluation of the qualifications of entities providing bids or proposals;
- .10 Consultation concerning replacement of Work resulting from fire or other cause during construction; or.
- .11 Assistance to the Initial Decision Maker, if other than the Architect.
- § 4.2.2 To avoid delay in the Construction Phase, the Architect shall provide the following Additional Services, notify the Owner with reasonable promptness, and explain the facts and circumstances giving rise to the need. If, upon receipt of the Architect's notice, the Owner determines that all or parts of the services are not required, the Owner shall give prompt written notice to the Architect of the Owner's determination. The Owner shall compensate the Architect for the services provided prior to the Architect's receipt of the Owner's notice.
  - Reviewing a Contractor's submittal out of sequence from the submittal schedule approved by the Architect;
  - .2 Responding to the Contractor's requests for information that are not prepared in accordance with the Contract Documents or where such information is available to the Contractor from a careful study and comparison of the Contract Documents, field conditions, other Owner-provided information, Contractor-prepared coordination drawings, or prior Project correspondence or documentation;
  - .3 Preparing Change Orders and Construction Change Directives that require evaluation of Contractor's proposals and supporting data, or the preparation or revision of Instruments of Service;
  - .4 Evaluating an extensive number of Claims as the Initial Decision Maker; or,
  - .5 Evaluating substitutions proposed by the Owner or Contractor and making subsequent revisions to Instruments of Service resulting therefrom.
- § 4.2.3 The Architect shall provide Construction Phase Services exceeding the limits set forth below as Additional Services. When the limits below are reached, the Architect shall notify the Owner:
  - .1 Two (2) reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor
  - .2 Bi-weekly visits to the site by the Architect during construction
  - .3 One (1) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents
  - .4 One (1) inspections for any portion of the Work to determine final completion.
- § 4.2.4 Except for services required under Section 3.6.6.5 and those services that do not exceed the limits set forth in Section 4.2.3, Construction Phase Services provided more than 60 days after (1) the date of Substantial Completion of the Work or (2) the initial date of Substantial Completion identified in the agreement between the Owner and Contractor, whichever is earlier, shall be compensated as Additional Services to the extent the Architect incurs additional cost in providing those Construction Phase Services.
- § 4.2.5 If the services covered by this Agreement have not been completed within Twenty Four (24) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

Init.

# ARTICLE 5 OWNER'S RESPONSIBILITIES

- § 5.1 Unless otherwise provided for under this Agreement, the Owner shall provide information in a timely manner regarding requirements for and limitations on the Project, including a written program, which shall set forth the Owner's objectives; schedule; constraints and criteria, including space requirements and relationships; flexibility; expandability; special equipment; systems; and site requirements.
- § 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes: (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.
- § 5.3 The Owner shall identify a representative authorized to act on the Owner's behalf with respect to the Project. The Owner shall render decisions and approve the Architect's submittals in a timely manner in order to avoid unreasonable delay in the orderly and sequential progress of the Architect's services.
- § 5.4 The Owner shall furnish surveys to describe physical characteristics, legal limitations and utility locations for the site of the Project, and a written legal description of the site. The surveys and legal information shall include, as applicable, grades and lines of streets, alleys, pavements and adjoining property and structures; designated wetlands; adjacent drainage; rights-of-way, restrictions, easements, encroachments, zoning, deed restrictions, boundaries and contours of the site; locations, dimensions, and other necessary data with respect to existing buildings, other improvements and trees; and information concerning available utility services and lines, both public and private, above and below grade, including inverts and depths. All the information on the survey shall be referenced to a Project benchmark.
- § 5.5 The Owner shall furnish services of geotechnical engineers, which may include test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.
- § 5.6 The Owner shall provide the Supplemental Services designated as the Owner's responsibility in Section 4.1.1.
- § 5.7 If the Owner identified a Sustainable Objective in Article 1, the Owner shall fulfill its responsibilities as required in AIA Document E204™\_2017, Sustainable Projects Exhibit, attached to this Agreement.
- § 5.8 The Owner shall coordinate the services of its own consultants with those services provided by the Architect. Upon the Architect's request, the Owner shall furnish copies of the scope of services in the contracts between the Owner and the Owner's consultants. The Owner shall furnish the services of consultants other than those designated as the responsibility of the Architect in this Agreement, or authorize the Architect to furnish them as an Additional Service, when the Architect requests such services and demonstrates that they are reasonably required by the scope of the Project. The Owner shall require that its consultants and contractors maintain insurance, including professional liability insurance, as appropriate to the services or work provided.
- § 5.9 The Owner shall furnish tests, inspections and reports required by law or the Contract Documents, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials.
- § 5.10 The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.
- § 5.11 The Owner shall provide prompt written notice to the Architect if the Owner becomes aware of any fault or defect in the Project, including errors, omissions or inconsistencies in the Architect's Instruments of Service.

- § 5.12 The Owner shall include the Architect in all communications with the Contractor that relate to or affect the Architect's services or professional responsibilities. The Owner shall promptly notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect.
- § 5.13 Before executing the Contract for Construction, the Owner shall coordinate the Architect's duties and responsibilities set forth in the Contract for Construction with the Architect's services set forth in this Agreement. The Owner shall provide the Architect a copy of the executed agreement between the Owner and Contractor, including the General Conditions of the Contract for Construction.
- § 5.14 The Owner shall provide the Architect access to the Project site prior to commencement of the Work and shall obligate the Contractor to provide the Architect access to the Work wherever it is in preparation or progress.
- § 5.15 Within 15 days after receipt of a written request from the Architect, the Owner shall furnish the requested information as necessary and relevant for the Architect to evaluate, give notice of, or enforce lien rights.

#### ARTICLE 6 COST OF THE WORK

- § 6.1 For purposes of this Agreement, the Cost of the Work shall be the total cost to the Owner to construct all elements of the Project designed or specified by the Architect and shall include contractors' general conditions costs, overhead and profit. The Cost of the Work also includes the reasonable value of labor, materials, and equipment, donated to, or otherwise furnished by, the Owner. The Cost of the Work does not include the compensation of the Architect; the costs of the land, rights-of-way, financing, or contingencies for changes in the Work; or other costs that are the responsibility of the Owner.
- § 6.2 The Owner's budget for the Cost of the Work is provided in Initial Information, and shall be adjusted throughout the Project as required under Sections 5.2, 6.4 and 6.5. Evaluations of the Owner's budget for the Cost of the Work, and the preliminary estimate of the Cost of the Work and updated estimates of the Cost of the Work, prepared by the Architect, represent the Architect's judgment as a design professional. It is recognized, however, that neither the Architect nor the Owner has control over the cost of labor, materials, or equipment; the Contractor's methods of determining bid prices; or competitive bidding, market, or negotiating conditions. Accordingly, the Architect cannot and does not warrant or represent that bids or negotiated prices will not vary from the Owner's budget for the Cost of the Work, or from any estimate of the Cost of the Work, or evaluation, prepared or agreed to by the Architect.
- § 6.3 In preparing estimates of the Cost of Work, the Architect shall be permitted to include contingencies for design, bidding, and price escalation; to determine what materials, equipment, component systems, and types of construction are to be included in the Contract Documents; to recommend reasonable adjustments in the program and scope of the Project; and to include design alternates as may be necessary to adjust the estimated Cost of the Work to meet the Owner's budget. The Architect's estimate of the Cost of the Work shall be based on current area, volume or similar conceptual estimating techniques. If the Owner requires a detailed estimate of the Cost of the Work, the Architect shall provide such an estimate, if identified as the Architect's responsibility in Section 4.1.1, as a Supplemental Service.
- § 6.4 If, through no fault of the Architect, the Procurement Phase has not commenced within 90 days after the Architect submits the Construction Documents to the Owner, the Owner's budget for the Cost of the Work shall be adjusted to reflect changes in the general level of prices in the applicable construction market.
- § 6.5 If at any time the Architect's estimate of the Cost of the Work exceeds the Owner's budget for the Cost of the Work, the Architect shall make appropriate recommendations to the Owner to adjust the Project's size, quality, or budget for the Cost of the Work, and the Owner shall cooperate with the Architect in making such adjustments.
- § 6.6 If the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services is exceeded by the lowest bona fide bid or negotiated proposal, the Owner shall
  - .1 give written approval of an increase in the budget for the Cost of the Work;
  - .2 authorize rebidding or renegotiating of the Project within a reasonable time;
  - .3 terminate in accordance with Section 9.5;
  - .4 in consultation with the Architect, revise the Project program, scope, or quality as required to reduce

the Cost of the Work; or,

.5 implement any other mutually acceptable alternative.

§ 6.7 If the Owner chooses to proceed under Section 6.6.4, the Architect shall modify the Construction Documents as necessary to comply with the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services, or the budget as adjusted under Section 6.6.1. If the Owner requires the Architect to modify the Construction Documents because the lowest bona fide bid or negotiated proposal exceeds the Owner's budget for the Cost of the Work due to market conditions the Architect could not reasonably anticipate, the Owner shall compensate the Architect for the modifications as an Additional Service pursuant to Section 11.3; otherwise the Architect's services for modifying the Construction Documents shall be without additional compensation. In any event, the Architect's modification of the Construction Documents shall be the limit of the Architect's responsibility under this Article 6.

#### ARTICLE 7 COPYRIGHTS AND LICENSES

§ 7.1 The Architect and the Owner warrant that in transmitting Instruments of Service, or any other information, the transmitting party is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project.

- § 7.2 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements or for similar purposes in connection with the Project is not to be construed as publication in derogation of the reserved rights of the Architect and the Architect's consultants.
- § 7.3 The Architect grants to the Owner a nonexclusive license to use the Architect's Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations under this Agreement, including prompt payment of all sums due pursuant to Article 9 and Article 11. The Architect shall obtain similar nonexclusive licenses from the Architect's consultants consistent with this Agreement. The license granted under this section permits the Owner to authorize the Contractor, Subcontractors, Sub-subcontractors, and suppliers, as well as the Owner's consultants and separate contractors, to reproduce applicable portions of the Instruments of Service, subject to any protocols established pursuant to Section 1.3, solely and exclusively for use in performing services or construction for the Project. If the Architect rightfully terminates this Agreement for cause as provided in Section 9.4, the license granted in this Section 7.3 shall terminate.
- § 7.3.1 In the event the Owner uses the Instruments of Service without retaining the authors of the Instruments of Service, the Owner releases the Architect and Architect's consultant(s) from all claims and causes of action arising from such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the Architect and its consultants from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's use of the Instruments of Service under this Section 7.3.1. The terms of this Section 7.3.1 shall not apply if the Owner rightfully terminates this Agreement for cause under Section 9.4.
- § 7.4 Except for the licenses granted in this Article 7, no other license or right shall be deemed granted or implied under this Agreement. The Owner shall not assign, delegate, sublicense, pledge or otherwise transfer any license granted herein to another party without the prior written agreement of the Architect. Any unauthorized use of the Instruments of Service shall be at the Owner's sole risk and without liability to the Architect and the Architect's consultants.
- § 7.5 Except as otherwise stated in Section 7.3, the provisions of this Article 7 shall survive the termination of this Agreement.

#### ARTICLE 8 CLAIMS AND DISPUTES

#### § 8.1 General

§ 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but

in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.

- § 8.1.2 To the extent damages are covered by property insurance, the Owner and Architect waive all rights against each other and against the contractors, consultants, agents, and employees of the other for damages, except such rights as they may have to the proceeds of such insurance as set forth in AIA Document A201–2017, General Conditions of the Contract for Construction. The Owner or the Architect, as appropriate, shall require of the contractors, consultants, agents, and employees of any of them, similar waivers in favor of the other parties enumerated herein.
- § 8.1.3 The Architect and Owner waive consequential damages for claims, disputes, or other matters in question, arising out of or relating to this Agreement. This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination of this Agreement, except as specifically provided in Section 9.7.

#### § 8.2 Mediation

- § 8.2.1 Any claim, dispute or other matter in question arising out of or related to this Agreement shall be subject to mediation as a condition precedent to binding dispute resolution. If such matter relates to or is the subject of a lien arising out of the Architect's services, the Architect may proceed in accordance with applicable law to comply with the lien notice or filing deadlines prior to resolution of the matter by mediation or by binding dispute resolution.
- § 8.2.2 The Owner and Architect shall endeavor to resolve claims, disputes and other matters in question between them by mediation, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of this Agreement. A request for mediation shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of a complaint or other appropriate demand for binding dispute resolution but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration proceeding is stayed pursuant to this section, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.
- § 8.2.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.
- § 8.2.4 If the parties do not resolve a dispute through mediation pursuant to this Section 8.2, the method of binding dispute resolution shall be the following:

  (Check the appropriate box.)

  [ ] Arbitration pursuant to Section 8.3 of this Agreement

  [ X ] Litigation in a court of competent jurisdiction

[ ] Other: (Specify)

If the Owner and Architect do not select a method of binding dispute resolution, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, the dispute will be resolved in a court of competent jurisdiction.

#### § 8.3 Arbitration

- § 8.3.1 If the parties have selected arbitration as the method for binding dispute resolution in this Agreement, any claim, dispute or other matter in question arising out of or related to this Agreement subject to, but not resolved by, mediation shall be subject to arbitration, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of this Agreement. A demand for arbitration shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the arbitration.
- § 8.3.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the claim, dispute or other matter in question would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the claim, dispute or other matter in question.
- § 8.3.2 The foregoing agreement to arbitrate, and other agreements to arbitrate with an additional person or entity duly consented to by parties to this Agreement, shall be specifically enforceable in accordance with applicable law in any court having jurisdiction thereof.
- § 8.3.3 The award rendered by the arbitrator(s) shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

#### § 8.3.4 Consolidation or Joinder

- § 8.3.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation; (2) the arbitrations to be consolidated substantially involve common questions of law or fact; and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).
- § 8.3.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.
- § 8.3.4.3 The Owner and Architect grant to any person or entity made a party to an arbitration conducted under this Section 8.3, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Architect under this Agreement.
- § 8.4 The provisions of this Article 8 shall survive the termination of this Agreement.

# ARTICLE 9 TERMINATION OR SUSPENSION

- § 9.1 If the Owner fails to make payments to the Architect in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Architect's option, cause for suspension of performance of services under this Agreement. If the Architect elects to suspend services, the Architect shall give seven days' written notice to the Owner before suspending services. In the event of a suspension of services, the Architect shall have no liability to the Owner for delay or damage caused the Owner because of such suspension of services. Before resuming services, the Owner shall pay the Architect all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

- § 9.3 If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Architect, the Architect may terminate this Agreement by giving not less than seven days' written notice.
- § 9.4 Either party may terminate this Agreement upon seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.
- § 9.5 The Owner may terminate this Agreement upon seven days' written notice to the Architect for the Owner's convenience and without cause.
- § 9.6 If the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall compensate the Architect for services performed prior to termination, Reimbursable Expenses incurred, and costs attributable to termination, including the costs attributable to the Architect's termination of consultant agreements.
- § 9.7 In addition to any amounts paid under Section 9.6, if the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall pay to the Architect the following fees:

(Set forth below the amount of any termination or licensing fee, or the method for determining any termination or licensing fee.)

.1 Termination Fee:

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

.2 Licensing Fee if the Owner intends to continue using the Architect's Instruments of Service:

As mutually agreed

- § 9.8 Except as otherwise expressly provided herein, this Agreement shall terminate one year from the date of Substantial Completion.
- § 9.9 The Owner's rights to use the Architect's Instruments of Service in the event of a termination of this Agreement are set forth in Article 7 and Section 9.7.

# ARTICLE 10 MISCELLANEOUS PROVISIONS

Init.

- § 10.1 This Agreement shall be governed by the law of the place where the Project is located, excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 8.3.
- § 10.2 Terms in this Agreement shall have the same meaning as those in AIA Document A201-2017, General Conditions of the Contract for Construction.
- § 10.3 The Owner and Architect, respectively, bind themselves, their agents, successors, assigns, and legal representatives to this Agreement. Neither the Owner nor the Architect shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement, including any payments due to the Architect by the Owner prior to the assignment.
- § 10.4 If the Owner requests the Architect to execute certificates, the proposed language of such certificates shall be submitted to the Architect for review at least 14 days prior to the requested dates of execution. If the Owner requests the Architect to execute consents reasonably required to facilitate assignment to a lender, the Architect shall execute all such consents that are consistent with this Agreement, provided the proposed consent is submitted to the Architect for review at least 14 days prior to execution. The Architect shall not be required to execute certificates or consents that would require knowledge, services, or responsibilities beyond the scope of this Agreement.

- § 10.5 Nothing contained in this Agreement shall create a contractual relationship with, or a cause of action in favor of, a third party against either the Owner or Architect.
- § 10.6 Unless otherwise required in this Agreement, the Architect shall have no responsibility for the discovery, presence, handling, removal or disposal of, or exposure of persons to, hazardous materials or toxic substances in any form at the Project site.
- § 10.7 The Architect shall have the right to include photographic or artistic representations of the design of the Project among the Architect's promotional and professional materials. The Architect shall be given reasonable access to the completed Project to make such representations. However, the Architect's materials shall not include the Owner's confidential or proprietary information if the Owner has previously advised the Architect in writing of the specific information considered by the Owner to be confidential or proprietary. The Owner shall provide professional credit for the Architect in the Owner's promotional materials for the Project. This Section 10.7 shall survive the termination of this Agreement unless the Owner terminates this Agreement for cause pursuant to Section 9.4.
- § 10.8 If the Architect or Owner receives information specifically designated as "confidential" or "business proprietary," the receiving party shall keep such information strictly confidential and shall not disclose it to any other person except as set forth in Section 10.8.1. This Section 10.8 shall survive the termination of this Agreement.
- § 10.8.1 The receiving party may disclose "confidential" or "business proprietary" information after 7 days' notice to the other party, when required by law, arbitrator's order, or court order, including a subpoena or other form of compulsory legal process issued by a court or governmental entity, or to the extent such information is reasonably necessary for the receiving party to defend itself in any dispute. The receiving party may also disclose such information to its employees, consultants, or contractors in order to perform services or work solely and exclusively for the Project, provided those employees, consultants and contractors are subject to the restrictions on the disclosure and use of such information as set forth in this Section 10.8.
- § 10.9 The invalidity of any provision of the Agreement shall not invalidate the Agreement or its remaining provisions. If it is determined that any provision of the Agreement violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Agreement shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Agreement.

# ARTICLE 11 COMPENSATION

§ 11.1 For the Architect's Basic Services described under Article 3, the Owner shall compensate the Architect as follows:

.1 Stipulated Sum (Insert amount)

Compensation in the amount of Seventy Five Thousand Dollars and Zero Cents (\$75,000.00)

- .2 Percentage Basis (Insert percentage value)
  - ( ) % of the Owner's budget for the Cost of the Work, as calculated in accordance with Section 11.6.
- .3 Other (Describe the method of compensation)
- § 11.2 For the Architect's Supplemental Services designated in Section 4.1.1 and for any Sustainability Services required pursuant to Section 4.1.3, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation. If necessary, list specific services to which particular methods of compensation apply.)

Hourly or as mutually agreed upon in advance of services

§ 11.3 For Additional Services that may arise during the course of the Project, including those under Section 4.2, the Owner shall compensate the Architect as follows:

(Insert amount of, or basis for, compensation.)

Hourly or as mutually agreed upon in advance of services

§ 11.4 Compensation for Supplemental and Additional Services of the Architect's consultants when not included in Section 11.2 or 11.3, shall be the amount invoiced to the Architect plus percent (%), or as follows: (Insert amount of, or basis for computing, Architect's consultants' compensation for Supplemental or Additional Services.)

As mutually agreed by Owner and Architect in advance of services

§ 11.5 When compensation for Basic Services is based on a stipulated sum or a percentage basis, the proportion of compensation for each phase of services shall be as follows:

| Total Basic Compensation | one hundred | percent ( | 100 | %) |
|--------------------------|-------------|-----------|-----|----|
| Construction Phase       | Twenty Five | percent ( | 25  | %) |
| Procurement Phase        | Five        | percent ( | 5   | %) |
| Phase                    |             |           |     |    |
| Construction Documents   | Twenty Five | percent ( | 25  | %) |
| Design Development Phase | Twenty Five | percent ( | 25  | %) |
| Schematic Design Phase   | Twenty      | percent ( | 20  | %) |

- § 11.6 When compensation identified in Section 11.1 is on a percentage basis, progress payments for each phase of Basic Services shall be calculated by multiplying the percentages identified in this Article by the Owner's most recent budget for the Cost of the Work. Compensation paid in previous progress payments shall not be adjusted based on subsequent updates to the Owner's budget for the Cost of the Work.
- § 11.6.1 When compensation is on a percentage basis and any portions of the Project are deleted or otherwise not constructed, compensation for those portions of the Project shall be payable to the extent services are performed on those portions. The Architect shall be entitled to compensation in accordance with this Agreement for all services performed whether or not the Construction Phase is commenced.
- § 11.7 The hourly billing rates for services of the Architect and the Architect's consultants are set forth below. The rates shall be adjusted in accordance with the Architect's and Architect's consultants' normal review practices. (If applicable, attach an exhibit of hourly billing rates or insert them below.)

See Exhibit C - Hourly Compensation and Reimbursable Expenses

**Employee or Category** 

Rate (\$0.00)

# § 11.8 Compensation for Reimbursable Expenses

- § 11.8.1 Reimbursable Expenses are in addition to compensation for Basic, Supplemental, and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:
  - Transportation and authorized out-of-town travel and subsistence;
  - .2 Long distance services, dedicated data and communication services, teleconferences, Project web sites, and extranets;
  - .3 Permitting and other fees required by authorities having jurisdiction over the Project;
  - .4 Printing, reproductions, plots, and standard form documents;
  - .5 Postage, handling, and delivery:

- .6 Expense of overtime work requiring higher than regular rates, if authorized in advance by the Owner;
- .7 Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner or required for the Project;
- .8 If required by the Owner, and with the Owner's prior written approval, the Architect's consultants' expenses of professional liability insurance dedicated exclusively to this Project, or the expense of additional insurance coverage or limits in excess of that normally maintained by the Architect's consultants;
- All taxes levied on professional services and on reimbursable expenses;
- .10 Site office expenses;
- .11 Registration fees and any other fees charged by the Certifying Authority or by other entities as necessary to achieve the Sustainable Objective; and,
- .12 Other similar Project-related expenditures.
- § 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus Ten percent (10 %) of the expenses incurred.
- § 11.9 Architect's Insurance. If the types and limits of coverage required in Section 2.5 are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect for the additional costs incurred by the Architect for the additional coverages as set forth below:

(Insert the additional coverages the Architect is required to obtain in order to satisfy the requirements set forth in Section 2.5, and for which the Owner shall reimburse the Architect.)

N/A

# § 11.10 Payments to the Architect

§ 11.10.1 Initial Payments

§ 11.10.1.1 An initial payment of Zero (\$0) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of Zero (\$0) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred.

#### § 11.10.2 Progress Payments

§ 11.10.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid Forty-Five (45) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect. (Insert rate of monthly or annual interest agreed upon.)

1.5 % Monthly

§ 11.10.2.2 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work, unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.

§ 11.10.2.3 Records of Reimbursable Expenses, expenses pertaining to Supplemental and Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

§ 11.10.2.4 Owner Agrees to receive invoices by email, which will be sent to the following individual and email address:

Full Name: Kelly Christopherson

Email Address:

Init.

# ARTICLE 12 SPECIAL TERMS AND CONDITIONS

Special terms and conditions that modify this Agreement are as follows: (Include other terms and conditions applicable to this Agreement.)

None

# ARTICLE 13 SCOPE OF THE AGREEMENT

§ 13.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both the Owner and Architect.

§ 13.2 This Agreement is comprised of the following documents identified below:

.1 AIA Document B101™-2017, Standard Form Agreement Between Owner and Architect

(Clearly identify any other exhibits incorporated into this Agreement, including any exhibits and scopes of services

.2 AIA Document E203<sup>TM</sup> 2013, Building Information Modeling and Digital Data Exhibit, dated as indicated below:

(Insert the date of the E203-2013 incorporated into this agreement.)

| .3                    | Exhibits:<br>(Check the appropriate box for any exhibits incorporated into this Agreement.   |
|-----------------------|--|
| [ ] AIA (Insert the d | Document E204TM_2017, Sustainable Projects Exhibit, dated as indicated below: date of the E204-2017 incorporated into this agreement.) |
|                       |  |

identified as exhibits in Section 4.1.2.)

Exhibit A: Project Initial Information

Exhibit B: JLG Image Release

Exhibit C: Schedule of Hourly Compensation & Reimbursable Expenses

Other Exhibits incorporated into this Agreement:

.4 Other documents:

(List other documents, if any, forming part of the Agreement.)

This Agreement entered into as of the day and year first written above.

| ARCHITECT (Signature)                                  |
|--|
| Catherine Dekkenga, AIA, Principal                     |
| (Printed name, title, and license number, if required) |
|  |

# Additions and Deletions Report for

AIA® Document B101 - 2017

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 15:31:59 ET on 10/05/2022.

| PAGE 1   |
|--|
| AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two   |
|  |
| Huron Public School District  150 5th St. SW  Huron, SD 57350  |
| ***  |
| JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104  |
|  |
| 21392.01 – Huron School District – Track Resurfacing Resurfacing existing thrack   |
| PAGE 2   |
| TABLE OF ARTICLES  |
|  |
| (Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.) |
|  |
| Resurfacing existing track   |
|  |
| Existing track owned and operated by the School District   |

# PAGE 3 Bidding: Late fall 2022 Construction to Commence spring/summer 2023 ... N/A \*\*\* N/A \*\*\* N/A Design/Bid/Build N/A Kelly Christopheron Huron Public School District 150 5th St. SW Huron, SD 57350 Telephone: Email: kelly.christopherson@k12.sd.us \*\*\* N/A PAGE 4 N/A

.2 Civil Engineer:

\$250,000.00 - \$525,000.00

... N/A Catherine Dekkenga JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104 Telephone: 605-271-2531 Email: cdekkenga@jlgarchitects.com N/A N/A N/A PAGE 5 .2 Civil Engineer: ... Stockwell Engineers, Inc. 801 N. Phillips Ave. Suite 100 Sioux Falls, SD 57104 ... Telephone: 605-338-6668

Additions and Deletions Report for AIA Document B101" – 2017, Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects." "AIA," The AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 15:31:59 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents\* Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

N/A

N/A

#### PAGE 6

§ 2.5.1 Commercial General Liability with policy limits of not less than (S-One Million Dollars and Zero Cents (\$ 1,000,000.00 ) for each occurrence and Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) in the aggregate for bodily injury and property damage.

§ 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of not less than (\$-One Million Dollars and Zero Cents (\$1,000,000.00) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.

§ 2.5.5 Employers' Liability with policy limits not less than (\$ ) each accident, (\$ Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each accident. Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00 ) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000.00 ) policy limit.

§ 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of not less than (\$-Two Million Dollars and Zero Cents (\$2,000,000.00) per claim and Four Million Dollars and Zero Cents (\$ 4,000,000.00 ) in the aggregate.

§ 3.1 The Architect's Basic Services consist of those described in this Article 3 and include usual and eustomary structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.

#### PAGE 7

§ 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner alternative approaches approach(es) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.

§ 3.2.5.2 The Architect shall consider the value of alternative-materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.

#### PAGE 9

§ 3.6.2.1 The Architect shall visit the site at intervals appropriate to the stage points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known

Additions and Deletions Report for AIA Document B101" - 2017, Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and e used wilhout permission. This document was produced by AIA software at 15:31:59 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.

#### PAGE 10

§ 3.6.4.2 The Architect shall review and approve, review, or take other appropriate action upon, per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

#### PAGE 11

# § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- The Parties agree and understand that, during the course of the construction of the Work, the Owner,
   Contractor, or other Project participants may choose to, at their own expense and initiative, make use of
   supplemental video imagery and information sourced from devices such as project webcams, Unmanned
   Aerial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain
   visual information and documentation regarding the state of the progress and/or quality of the Work
   (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly
   agree and acknowledge the following:
  - I. Since Supplementary Video Information encompasses only certain limited visual information
    and provides neither complete nor comprehensive visual information/measurements/dimensions,
    nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video
    Information shall not be used as a substitute for or required supplement to Architect's Construction
    Phase services, herein, but rather may, to the extent provided and made available to Architect in a
    reasonably timely, legible, and complete fashion, supplement the other data and information
    obtained by Architect pursuant to Architect's Construction Phase Services, herein.
  - 2. Architect shall have no obligation to conduct a comprehensive or continuous review of any
    Supplementary Video Information, but rather may, at its sole discretion, choose to review such
    Supplementary Video Information, or any portion thereof, in connection with the performance of
    its Construction Phase Services on the Project.

# PAGE 12

3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part

#### ARTICLE thereof.

#### ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

#### PAGE 13

| § 4.1.1.1  | Programming  | Not Provided |
|------------|--|--------------|
| § 4.1.1.2  | Multiple preliminary designs   | Not Provided |
| § 4.1.1.3  | Measured drawings  | Not Provided |
| § 4.1.1.4  | Existing facilities surveys  | Not Provided |
| § 4.1.1.5  | Site evaluation and planning   | Not Provided |
| § 4.1.1.6  | Building Information Model management responsibilities               | Not Provided |
| § 4.1.1.7  | Development of Building Information Models for post construction use | Not Provided |
| § 4.1.1.8  | Civil engineering  | Architect    |
| § 4.1.1.9  | Landscape design   | Architect    |
| § 4.1.1.10 | Architectural interior design  | Not Provided |
| § 4.1.1.11 | Value analysis   | Not Provided |
| § 4.1.1.12 | Detailed cost estimating beyond that required in Section 6.3         | Not Provided |
| § 4.1.1.13 | On-site project representation                                       | Not Provided |
| § 4.1.1.14 | Conformed documents for construction                                 | Not Provided |
| § 4.1.1.15 | As-designed record drawings  | Not Provided |
| § 4.1.1.16 | As-constructed record drawings                                       | Not Provided |
| § 4.1.1.17 | Post-occupancy evaluation  | Not Provided |
| § 4.1.1.18 | Facility support services  | Not Provided |
| § 4.1.1.19 | Tenant-related services  | Not Provided |
| § 4.1.1.20 | Architect's coordination of the Owner's consultants                  | Not Provided |
| § 4.1.1.21 | Telecommunications/data design                                       | Not Provided |
| § 4.1.1.22 | Security evaluation and planning                                     | Not Provided |
| § 4.1.1.23 | Commissioning  | Not Provided |
| § 4.1.1.24 | Sustainable Project Services pursuant to Section 4.1.3               | Not Provided |
| § 4.1.1.25 | Fast-track design services   | Not Provided |
| § 4.1.1.26 | Multiple bid packages  | Not Provided |
| § 4.1.1.27 | Historic preservation  | Not Provided |
| § 4.1.1.28 | Furniture, furnishings, and equipment design                         | Not Provided |
| § 4.1.1.29 | Other services provided by specialty Consultants                     | Not Provided |
| § 4.1.1.30 | Other Supplemental Services  | Not Provided |

,,,

SS 4.1.1.8 Civil Engineering: Limited to usual and customary civil engineering services

SS 4.1.1.9 Landscape design: Limited to Schematic design phase services to indicate design intent and provide basis for cost allowance and bid alternates; based on accepted scope/alternates, remaining documentation will be provided design/build by landscaping subcontractor (Architect to review and approve final drawings for conformance with design intent and review shop drawings)

#### N/A

#### PAGE 14

.1 Two (2 ) reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor

- .2 (+Bi-weekly visits to the site by the Architect during construction

  .3 One (1 ) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents

  .4 One (1 ) inspections for any portion of the Work to determine final completion.
- § 4.2.5 If the services covered by this Agreement have not been completed within <a href="Twenty Four">Twenty Four</a> (24) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

#### PAGE 15

§ 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1; 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes: (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.

#### PAGE 18

§ 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.

[X] Litigation in a court of competent jurisdiction

#### PAGE 19

...

§ 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

#### PAGE 20

§ 9.4 Either party may terminate this Agreement upon not less than seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.

§ 9.5 The Owner may terminate this Agreement upon not less than seven days' written notice to the Architect for the Owner's convenience and without cause.

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

As mutually agreed

#### PAGE 21

Compensation in the amount of Seventy Five Thousand Dollars and Zero Cents (\$75,000.00)

#### PAGE 22

Hourly or as mutually agreed upon in advance of services

Hourly or as mutually agreed upon in advance of services

As mutually agreed by Owner and Architect in advance of services

Schematic Design Phase Twenty percent ( %) Design Development Phase Twenty Five percent ( %) Construction Documents Twenty Five percent ( 96) Phase Procurement Phase Five percent ( %) Construction Phase Twenty Five percent ( %)

See Exhibit C - Hourly Compensation and Reimbursable Expenses

#### PAGE 23

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus <u>Ten</u> percent (<u>10</u>%) of the expenses incurred.

#### N/A

§ 11.10.1.1 An initial payment of Zero (\$ 0 ) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice. § 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of Zero (\$ 0 ) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred. § 11.10.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid Forty-Five (45) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect. 1.5 % Monthly § 11.10.2.4 Owner Agrees to receive invoices by email, which will be sent to the following individual and email address: Full Name: Kelly Christopherson Email Address: ARTICLE kelly.christopherson@k12.sd.us PAGE 24 ARTICLE 12 SPECIAL TERMS AND CONDITIONS None Exhibit A: Project Initial Information Exhibit B: JLG Image Release Exhibit C: Schedule of Hourly Compensation & Reimbursable Expenses

...

Additions and Deletions Report for AIA Document B101\*\* - 2017, Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," 'AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and sod without permission. This document was produced by AIA software at 15:31:59 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

| Garret | Bischoff, | President |
|--------|-----------|-----------|

\*\*\*

Catherine Dekkenga, AIA. Principal

# Certification of Document's Authenticity

AIA® Document D401™ - 2003

I, Catherine Dekkenga, AIA, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 15:31:59 ET on 10/05/2022 under Order No. 2114257989 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA® Document B101TM - 2017, Standard Form of Agreement Between Owner and Architect, as published by the AIA in its software, other than those additions and deletions shown in the associated Additions and Deletions Report.

| (Signed) |  |  |
|----------|--|--|
|          |  |  |
| (Title)  |  |  |
|          |  |  |
| Dated)   |  |  |



## Standard Form of Agreement Between Owner and Architect

AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two (In words, indicate day, month and year.)

BETWEEN the Architect's client identified as the Owner: (Name, legal status, address and other information)

Huron Public School District 150 5th St. SW Huron, SD 57350

and the Architect; (Name, legal status, address and other information)

JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104

for the following Project: (Name, location and detailed description)

21392.01 - Huron School District - TAC Ceiling Design and replacement of damaged ceiling in the Tiger Activities Center Huron, SD

The Owner and Architect agree as follows.

#### ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

## **TABLE OF ARTICLES**

- 1 INITIAL INFORMATION
- 2 ARCHITECT'S RESPONSIBILITIES
- 3 SCOPE OF ARCHITECT'S BASIC SERVICES
- 4 SUPPLEMENTAL AND ADDITIONAL SERVICES
- 5 OWNER'S RESPONSIBILITIES
- 6 COST OF THE WORK
- 7 COPYRIGHTS AND LICENSES
- 8 CLAIMS AND DISPUTES
- 9 TERMINATION OR SUSPENSION
- 10 MISCELLANEOUS PROVISIONS
- 11 COMPENSATION
- 12 SPECIAL TERMS AND CONDITIONS
- 13 SCOPE OF THE AGREEMENT

#### ARTICLE 1 INITIAL INFORMATION

§ 1.1 This Agreement is based on the Initial Information set forth in this Section 1.1.

(For each item in this section, insert the information or a statement such as "not applicable" or "unknown at time of execution.")

## § 1.1.1 The Owner's program for the Project:

(Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.)

# Design for the replacement of damaged ceiling in the TAC.

## § 1.1.2 The Project's physical characteristics:

(Identify or describe pertinent information about the Project's physical characteristics, such as size; location; dimensions; geotechnical reports; site boundaries; topographic surveys; traffic and utility studies; availability of public and private utilities and services; legal description of the site, etc.)

The TAC is an existing pre engineered metal building owned by the Huron School District. The ceiling has been damaged and needs to be replaced.

§ 1.1.3 The Owner's budget for the Cost of the Work, as defined in Section 6.1: (Provide total and, if known, a line item breakdown.)

\$150,000.00

Init.

- § 1.1.4 The Owner's anticipated design and construction milestone dates:
  - .1 Design phase milestone dates, if any:

Design: July 1, 2022 - September 9, 2022

Construction Documents: September 9, 2022 – October 21, 2022

Bidding: October 22, 2022 - November 19, 2022

Construction Administration: March 1, 2023 - July 19, 2023

.2 Construction commencement date:

Sprin/Summer 2023

.3 Substantial Completion date or dates:

N/A

.4 Other milestone dates:

N/A

§ 1.1.5 The Owner intends the following procurement and delivery method for the Project: (Identify method such as competitive bid or negotiated contract, as well as any requirements for accelerated or fast-track design and construction, multiple bid packages, or phased construction.)

Design/Bid/Build

§ 1.1.6 The Owner's anticipated Sustainable Objective for the Project: (Identify and describe the Owner's Sustainable Objective for the Project, if any.)

N/A

(Paragraph Deleted)

§ 1.1.6.1 If the Owner identifies a Sustainable Objective, the Owner and Architect shall complete and incorporate AIA Document E204™-2017, Sustainable Projects Exhibit, into this Agreement to define the terms, conditions and services related to the Owner's Sustainable Objective. If E204-2017 is incorporated into this agreement, the Owner and Architect shall incorporate the completed E204-2017 into the agreements with the consultants and contractors performing services or Work in any way associated with the Sustainable Objective.

§ 1.1.7 The Owner identifies the following representative in accordance with Section 5.3: (List name, address, and other contact information.)

Kelly Christopherson Huron Public School District 150 5th St. SW Huron, SD 57350

Telephone:

Email:

§ 1.1.8 The persons or entities, in addition to the Owner's representative, who are required to review the Architect's submittals to the Owner are as follows:

(List name, address, and other contact information.)

N/A

§ 1.1.9 The Owner shall retain the following consultants and contractors: (List name, legal status, address, and other contact information.)

Init.

.1 Geotechnical Engineer:

N/A

.2 Civil Engineer:

N/A

.3 Other, if any:

(List any other consultants and contractors retained by the Owner.)

N/A

§ 1.1.10 The Architect identifies the following representative in accordance with Section 2.3: (List name, address, and other contact information.)

Catherine Dekkenga JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104

Telephone: 605-271-2531

Email: cdekkenga@jlgarchitects.com

§ 1.1.11 The Architect shall retain the consultants identified in Sections 1.1.11.1 and 1.1.11.2: (List name, legal status, address, and other contact information.)

§ 1.1.11.1 Consultants retained under Basic Services:

.1 Structural Engineer:

SEA, Inc., Mark Smith 401 E. 8th St. Suite 201 Sioux Falls, SD 57103 Telephone: 605-334-0188

.2 Mechanical Engineer:

Associated Consulting Engineers, Inc., Jordan deWit 340 S. Phillips Ave. Sioux Falls, SD 57104 Telephone: 605-335-3720

.3 Electrical Engineer:

Init.

Associated Consulting Engineers, Inc., Brad Shoup 340 S. Phillips Ave. Sioux Falls, SD 57104

AlA Document B101\* – 2017. Copyright © 1974, 1978, 1987, 1997, 2007 and 2017 by The American Institute of Architects. All rights reserved. The "American Institute of Architects." "AIA," Ine AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 12:21:46 ET on 10/05/2022 under Order No.2114257989 which expires on 11/13/2022, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents\* Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

Telephone: 605-335-3720

§ 1.1.11.2 Consultants retained under Supplemental Services:

N/A

§ 1.1.12 Other Initial Information on which the Agreement is based:

N/A

- § 1.2 The Owner and Architect may rely on the Initial Information. Both parties, however, recognize that the Initial Information may materially change and, in that event, the Owner and the Architect shall appropriately adjust the Architect's services, schedule for the Architect's services, and the Architect's compensation. The Owner shall adjust the Owner's budget for the Cost of the Work and the Owner's anticipated design and construction milestones, as necessary, to accommodate material changes in the Initial Information.
- § 1.3 The parties shall agree upon protocols governing the transmission and use of Instruments of Service or any other information or documentation in digital form. The parties will use AIA Document E203<sup>TM</sup>\_2013, Building Information Modeling and Digital Data Exhibit, to establish the protocols for the development, use, transmission, and exchange of digital data.
- § 1.3.1 Any use of, or reliance on, all or a portion of a building information model without agreement to protocols governing the use of, and reliance on, the information contained in the model and without having those protocols set forth in AIA Document E203<sup>TM</sup>—2013, Building Information Modeling and Digital Data Exhibit, and the requisite AIA Document G202<sup>TM</sup>—2013, Project Building Information Modeling Protocol Form, shall be at the using or relying party's sole risk and without liability to the other party and its contractors or consultants, the authors of, or contributors to, the building information model, and each of their agents and employees.

## ARTICLE 2 ARCHITECT'S RESPONSIBILITIES

- § 2.1 The Architect shall provide professional services as set forth in this Agreement. The Architect represents that it is properly licensed in the jurisdiction where the Project is located to provide the services required by this Agreement, or shall cause such services to be performed by appropriately licensed design professionals.
- § 2.2 The Architect shall perform its services consistent with the professional skill and care ordinarily provided by architects practicing in the same or similar locality under the same or similar circumstances. The Architect shall perform its services as expeditiously as is consistent with such professional skill and care and the orderly progress of the Project.
- § 2.3 The Architect shall identify a representative authorized to act on behalf of the Architect with respect to the Project.
- § 2.4 Except with the Owner's knowledge and consent, the Architect shall not engage in any activity, or accept any employment, interest or contribution that would reasonably appear to compromise the Architect's professional judgment with respect to this Project.
- § 2.5 The Architect shall maintain the following insurance until termination of this Agreement. If any of the requirements set forth below are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect as set forth in Section 11.9.
- $\S$  2.5.1 Commercial General Liability with policy limits of One Million Dollars and Zero Cents ( $\S$  1,000,000.00 ) for each occurrence and Two Million Dollars and Zero Cents ( $\S$  2,000,000.00 ) in the aggregate for bodily injury and property damage.

Init.

- § 2.5.2 Automobile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy limits of One Million Dollars and Zero Cents (\$ 1,000,000.00 ) per accident for bodily injury, death of any person, and property damage arising out of the ownership, maintenance and use of those motor vehicles, along with any other statutorily required automobile coverage.
- § 2.5.3 The Architect may achieve the required limits and coverage for Commercial General Liability and Automobile Liability through a combination of primary and excess or umbrella liability insurance, provided such primary and excess or umbrella liability insurance policies result in the same or greater coverage as the coverages required under Sections 2.5.1 and 2.5.2, and in no event shall any excess or umbrella liability insurance provide narrower coverage than the primary policy. The excess policy shall not require the exhaustion of the underlying limits only through the actual payment by the underlying insurers.
- § 2.5.4 Workers' Compensation at statutory limits.
- § 2.5.5 Employers' Liability with policy limits Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00) each accident, Five Hundred Thousand Dollars and Zero Cents (\$ 500,000.00) each employee, and One Million Dollars and Zero Cents (\$ 1,000,000.00) policy limit.
- § 2.5.6 Professional Liability covering negligent acts, errors and omissions in the performance of professional services with policy limits of Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) per claim and Four Million Dollars and Zero Cents (\$ 4,000,000.00 ) in the aggregate.
- § 2.5.7 Additional Insured Obligations. To the fullest extent permitted by law, the Architect shall cause the primary and excess or umbrella polices for Commercial General Liability and Automobile Liability to include the Owner as an additional insured for claims caused in whole or in part by the Architect's negligent acts or omissions. The additional insured coverage shall be primary and non-contributory to any of the Owner's insurance policies and shall apply to both ongoing and completed operations.
- $\S$  2.5.8 The Architect shall provide certificates of insurance to the Owner that evidence compliance with the requirements in this Section 2.5.

## ARTICLE 3 SCOPE OF ARCHITECT'S BASIC SERVICES

- § 3.1 The Architect's Basic Services consist of those described in this Article 3 and include structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental or Additional Services.
- § 3.1.1 The Architect shall manage the Architect's services, research applicable design criteria, attend Project meetings, communicate with members of the Project team, and report progress to the Owner.
- § 3.1.2 The Architect shall coordinate its services with those services provided by the Owner and the Owner's consultants. The Architect shall be entitled to rely on, and shall not be responsible for, the accuracy, completeness, and timeliness of, services and information furnished by the Owner and the Owner's consultants. The Architect shall provide prompt written notice to the Owner if the Architect becomes aware of any error, omission, or inconsistency in such services or information.
- § 3.1.3 As soon as practicable after the date of this Agreement, the Architect shall submit for the Owner's approval a schedule for the performance of the Architect's services. The schedule initially shall include anticipated dates for the commencement of construction and for Substantial Completion of the Work as set forth in the Initial Information. The schedule shall include allowances for periods of time required for the Owner's review, for the performance of the Owner's consultants, and for approval of submissions by authorities having jurisdiction over the Project. Once approved by the Owner, time limits established by the schedule shall not, except for reasonable cause, be exceeded by the Architect or Owner. With the Owner's approval, the Architect shall adjust the schedule, if necessary, as the Project proceeds until the commencement of construction.
- § 3.1.4 The Architect shall not be responsible for an Owner's directive or substitution, or for the Owner's acceptance of non-conforming Work, made or given without the Architect's written approval.

- § 3.1.5 The Architect shall contact governmental authorities required to approve the Construction Documents and entities providing utility services to the Project. The Architect shall respond to applicable design requirements imposed by those authorities and entities.
- § 3.1.6 The Architect shall assist the Owner in connection with the Owner's responsibility for filing documents required for the approval of governmental authorities having jurisdiction over the Project.

#### § 3.2 Schematic Design Phase Services

- § 3.2.1 The Architect shall review the program and other information furnished by the Owner, and shall review laws, codes, and regulations applicable to the Architect's services.
- § 3.2.2 The Architect shall prepare a preliminary evaluation of the Owner's program, schedule, budget for the Cost of the Work, Project site, the proposed procurement and delivery method, and other Initial Information, each in terms of the other, to ascertain the requirements of the Project. The Architect shall notify the Owner of (1) any inconsistencies discovered in the information, and (2) other information or consulting services that may be reasonably needed for the Project.
- § 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner approach(es) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.
- § 3.2.4 Based on the Project requirements agreed upon with the Owner, the Architect shall prepare and present, for the Owner's approval, a preliminary design illustrating the scale and relationship of the Project components.
- § 3.2.5 Based on the Owner's approval of the preliminary design, the Architect shall prepare Schematic Design Documents for the Owner's approval. The Schematic Design Documents shall consist of drawings and other documents including a site plan, if appropriate, and preliminary building plans, sections and elevations; and may include some combination of study models, perspective sketches, or digital representations. Preliminary selections of major building systems and construction materials shall be noted on the drawings or described in writing.
- § 3.2.5.1 The Architect shall consider sustainable design alternatives, such as material choices and building orientation, together with other considerations based on program and aesthetics, in developing a design that is consistent with the Owner's program, schedule and budget for the Cost of the Work. The Owner may obtain more advanced sustainable design services as a Supplemental Service under Section 4.1.1.
- § 3.2.5.2 The Architect shall consider the value of materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.
- § 3.2.6 The Architect shall submit to the Owner an estimate of the Cost of the Work prepared in accordance with Section 6.3.
- § 3.2.7 The Architect shall submit the Schematic Design Documents to the Owner, and request the Owner's approval.

#### § 3.3 Design Development Phase Services

- § 3.3.1 Based on the Owner's approval of the Schematic Design Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Design Development Documents for the Owner's approval. The Design Development Documents shall illustrate and describe the development of the approved Schematic Design Documents and shall consist of drawings and other documents including plans, sections, elevations, typical construction details, and diagrammatic layouts of building systems to fix and describe the size and character of the Project as to architectural, structural, mechanical and electrical systems, and other appropriate elements. The Design Development Documents shall also include outline specifications that identify major materials and systems and establish, in general, their quality levels.
- § 3.3.2 The Architect shall update the estimate of the Cost of the Work prepared in accordance with Section 6.3.

§ 3.3.3 The Architect shall submit the Design Development Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, and request the Owner's approval.

#### § 3.4 Construction Documents Phase Services

- § 3.4.1 Based on the Owner's approval of the Design Development Documents, and on the Owner's authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Construction Documents for the Owner's approval. The Construction Documents shall illustrate and describe the further development of the approved Design Development Documents and shall consist of Drawings and Specifications setting forth in detail the quality levels and performance criteria of materials and systems and other requirements for the construction of the Work. The Owner and Architect acknowledge that, in order to perform the Work, the Contractor will provide additional information, including Shop Drawings, Product Data, Samples and other similar submittals, which the Architect shall review in accordance with Section 3.6.4.
- § 3.4.2 The Architect shall incorporate the design requirements of governmental authorities having jurisdiction over the Project into the Construction Documents.
- § 3.4.3 During the development of the Construction Documents, the Architect shall assist the Owner in the development and preparation of (1) procurement information that describes the time, place, and conditions of bidding, including bidding or proposal forms; (2) the form of agreement between the Owner and Contractor; and (3) the Conditions of the Contract for Construction (General, Supplementary and other Conditions). The Architect shall also compile a project manual that includes the Conditions of the Contract for Construction and Specifications, and may include bidding requirements and sample forms.
- § 3.4.4 The Architect shall update the estimate for the Cost of the Work prepared in accordance with Section 6.3.
- § 3.4.5 The Architect shall submit the Construction Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, take any action required under Section 6.5, and request the Owner's approval.

### § 3.5 Procurement Phase Services

#### § 3.5.1 General

The Architect shall assist the Owner in establishing a list of prospective contractors. Following the Owner's approval of the Construction Documents, the Architect shall assist the Owner in (1) obtaining either competitive bids or negotiated proposals; (2) confirming responsiveness of bids or proposals; (3) determining the successful bid or proposal, if any; and, (4) awarding and preparing contracts for construction.

#### § 3.5.2 Competitive Bidding

§ 3.5.2.1 Bidding Documents shall consist of bidding requirements and proposed Contract Documents.

- § 3.5.2.2 The Architect shall assist the Owner in bidding the Project by:
  - .1 facilitating the distribution of Bidding Documents to prospective bidders;
  - .2 organizing and conducting a pre-bid conference for prospective bidders;
  - .3 preparing responses to questions from prospective bidders and providing clarifications and interpretations of the Bidding Documents to the prospective bidders in the form of addenda; and,
  - .4 organizing and conducting the opening of the bids, and subsequently documenting and distributing the bidding results, as directed by the Owner.
- § 3.5.2.3 If the Bidding Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective bidders.

#### § 3.5.3 Negotiated Proposals

§ 3.5.3.1 Proposal Documents shall consist of proposal requirements and proposed Contract Documents.

- § 3.5.3.2 The Architect shall assist the Owner in obtaining proposals by:
  - facilitating the distribution of Proposal Documents for distribution to prospective contractors and requesting their return upon completion of the negotiation process;

- .2 organizing and participating in selection interviews with prospective contractors;
- .3 preparing responses to questions from prospective contractors and providing clarifications and interpretations of the Proposal Documents to the prospective contractors in the form of addenda; and,
- .4 participating in negotiations with prospective contractors, and subsequently preparing a summary report of the negotiation results, as directed by the Owner.
- § 3.5.3.3 If the Proposal Documents permit substitutions, upon the Owner's written authorization, the Architect shall, as an Additional Service, consider requests for substitutions and prepare and distribute addenda identifying approved substitutions to all prospective contractors.

#### § 3.6 Construction Phase Services

#### § 3.6.1 General

- § 3.6.1.1 The Architect shall provide administration of the Contract between the Owner and the Contractor as set forth below and in AIA Document A201<sup>TM</sup> 2017, General Conditions of the Contract for Construction. If the Owner and Contractor modify AIA Document A201–2017, those modifications shall not affect the Architect's services under this Agreement unless the Owner and the Architect amend this Agreement.
- § 3.6.1.2 The Architect shall advise and consult with the Owner during the Construction Phase Services. The Architect shall have authority to act on behalf of the Owner only to the extent provided in this Agreement. The Architect shall not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, nor shall the Architect be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect shall be responsible for the Architect's negligent acts or omissions, but shall not have control over or charge of, and shall not be responsible for, acts or omissions of the Contractor or of any other persons or entities performing portions of the Work.
- § 3.6.1.3 Subject to Section 4.2 and except as provided in Section 3.6.6.5, the Architect's responsibility to provide Construction Phase Services commences with the award of the Contract for Construction and terminates on the date the Architect issues the final Certificate for Payment.

#### § 3.6.2 Evaluations of the Work

- § 3.6.2.1 The Architect shall visit the site at points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.
- § 3.6.2.2 The Architect has the authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect shall have the authority to require inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not the Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, suppliers, their agents or employees, or other persons or entities performing portions of the Work.
- § 3.6.2.3 The Architect shall interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests shall be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- § 3.6.2.4 Interpretations and decisions of the Architect shall be consistent with the intent of, and reasonably inferable from, the Contract Documents and shall be in writing or in the form of drawings. When making such interpretations and decisions, the Architect shall endeavor to secure faithful performance by both Owner and Contractor, shall not show partiality to either, and shall not be liable for results of interpretations or decisions rendered in good faith. The

Architect's decisions on matters relating to aesthetic effect shall be final if consistent with the intent expressed in the Contract Documents.

§ 3.6.2.5 Unless the Owner and Contractor designate another person to serve as an Initial Decision Maker, as that term is defined in AIA Document A201–2017, the Architect shall render initial decisions on Claims between the Owner and Contractor as provided in the Contract Documents.

## § 3.6.3 Certificates for Payment to Contractor

- § 3.6.3.1 The Architect shall review and certify the amounts due the Contractor and shall issue certificates in such amounts. The Architect's certification for payment shall constitute a representation to the Owner, based on the Architect's evaluation of the Work as provided in Section 3.6.2 and on the data comprising the Contractor's Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated, the quality of the Work is in accordance with the Contract Documents, and that the Contractor is entitled to payment in the amount certified. The foregoing representations are subject to (1) an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, (2) results of subsequent tests and inspections, (3) correction of minor deviations from the Contract Documents prior to completion, and (4) specific qualifications expressed by the Architect.
- § 3.6.3.2 The issuance of a Certificate for Payment shall not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) ascertained how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.
- § 3.6.3.3 The Architect shall maintain a record of the Applications and Certificates for Payment.

#### § 3.6.4 Submittals

- § 3.6.4.1 The Architect shall review the Contractor's submittal schedule and shall not unreasonably delay or withhold approval of the schedule. The Architect's action in reviewing submittals shall be taken in accordance with the approved submittal schedule or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time, in the Architect's professional judgment, to permit adequate review.
- § 3.6.4.2 The Architect shall review, or take other appropriate action per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- § 3.6.4.3 If the Contract Documents specifically require the Contractor to provide professional design services or certifications by a design professional related to systems, materials, or equipment, the Architect shall specify the appropriate performance and design criteria that such services must satisfy. The Architect shall review and take appropriate action on Shop Drawings and other submittals related to the Work designed or certified by the Contractor's design professional, provided the submittals bear such professional's seal and signature when submitted to the Architect. The Architect's review shall be for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect shall be entitled to rely upon, and shall not be responsible for, the adequacy and accuracy of the services, certifications, and approvals performed or provided by such design professionals.
- § 3.6.4.4 Subject to Section 4.2, the Architect shall review and respond to requests for information about the Contract Documents. The Architect shall set forth, in the Contract Documents, the requirements for requests for information. Requests for information shall include, at a minimum, a detailed written statement that indicates the specific Drawings or Specifications in need of clarification and the nature of the clarification requested. The Architect's response to such requests shall be made in writing within any time limits agreed upon, or otherwise with reasonable

promptness. If appropriate, the Architect shall prepare and issue supplemental Drawings and Specifications in response to the requests for information.

§ 3.6.4.5 The Architect shall maintain a record of submittals and copies of submittals supplied by the Contractor in accordance with the requirements of the Contract Documents.

#### § 3.6.5 Changes in the Work

§ 3.6.5.1 The Architect may order minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. Subject to Section 4.2, the Architect shall prepare Change Orders and Construction Change Directives for the Owner's approval and execution in accordance with the Contract Documents.

§ 3.6.5.2 The Architect shall maintain records relative to changes in the Work.

#### § 3.6.6 Project Completion

§ 3.6.6.1 The Architect shall:

- .1 conduct inspections to determine the date or dates of Substantial Completion and the date of final completion;
- .2 issue Certificates of Substantial Completion;
- .3 forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract Documents and received from the Contractor; and,
- 4 issue a final Certificate for Payment based upon a final inspection indicating that, to the best of the Architect's knowledge, information, and belief, the Work complies with the requirements of the Contract Documents.
- § 3.6.6.2 The Architect's inspections shall be conducted with the Owner to check conformance of the Work with the requirements of the Contract Documents and to verify the accuracy and completeness of the list submitted by the Contractor of Work to be completed or corrected.
- § 3.6.6.3 When Substantial Completion has been achieved, the Architect shall inform the Owner about the balance of the Contract Sum remaining to be paid the Contractor, including the amount to be retained from the Contract Sum, if any, for final completion or correction of the Work.
- § 3.6.6.4 The Architect shall forward to the Owner the following information received from the Contractor: (1) consent of surety or sureties, if any, to reduction in or partial release of retainage or the making of final payment; (2) affidavits, receipts, releases and waivers of liens, or bonds indemnifying the Owner against liens; and (3) any other documentation required of the Contractor under the Contract Documents.
- § 3.6.6.5 Upon request of the Owner, and prior to the expiration of one year from the date of Substantial Completion, the Architect shall, without additional compensation, conduct a meeting with the Owner to review the facility operations and performance.

## § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- 1. The Parties agree and understand that, during the course of the construction of the Work, the Owner, Contractor, or other Project participants may choose to, at their own expense and initiative, make use of supplemental video imagery and information sourced from devices such as project webcams, Unmanned Acrial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain visual information and documentation regarding the state of the progress and/or quality of the Work (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly agree and acknowledge the following:
  - Since Supplementary Video Information encompasses only certain limited visual information
    and provides neither complete nor comprehensive visual information/measurements/dimensions,
    nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video
    Information shall not be used as a substitute for or required supplement to Architect's Construction
    Phase services, herein, but rather may, to the extent provided and made available to Architect in a
    reasonably timely, legible, and complete fashion, supplement the other data and information
    obtained by Architect pursuant to Architect's Construction Phase Services, herein.

- Architect shall have no obligation to conduct a comprehensive or continuous review of any Supplementary Video Information, but rather may, at its sole discretion, choose to review such Supplementary Video Information, or any portion thereof, in connection with the performance of its Construction Phase Services on the Project.
- 3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part thereof.

#### ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

#### § 4.1 Supplemental Services

§ 4.1.1 The services listed below are not included in Basic Services but may be required for the Project. The Architect shall provide the listed Supplemental Services only if specifically designated in the table below as the Architect's responsibility, and the Owner shall compensate the Architect as provided in Section 11.2. Unless otherwise specifically addressed in this Agreement, if neither the Owner nor the Architect is designated, the parties agree that the listed Supplemental Service is not being provided for the Project.

(Designate the Architect's Supplemental Services and the Owner's Supplemental Services required for the Project by indicating whether the Architect or Owner shall be responsible for providing the identified Supplemental Service. Insert a description of the Supplemental Services in Section 4.1.2 below or attach the description of services as an exhibit to this Agreement.)

|            | ental Services   | Responsibility<br>(Architect, Owner, or not<br>provided) |
|------------|--|--|
| § 4.1.1.1  | Programming  | Not Provided   |
| § 4.1.1.2  | Multiple preliminary designs   | Not Provided   |
| § 4.1.1.3  | Measured drawings  | Not Provided   |
| § 4.1.1.4  | Existing facilities surveys  | Not Provided   |
| § 4.1.1.5  | Site evaluation and planning   | Not Provided   |
| § 4.1.1.6  | Building Information Model management responsibilities               | Not Provided   |
| § 4.1.1.7  | Development of Building Information Models for post construction use | Not Provided   |
| § 4.1.1.8  | Civil engineering  | Not Provided   |
| § 4.1.1.9  | Landscape design   | Not Provided   |
| § 4.1.1.10 | Architectural interior design  | Not Provided   |
| § 4.1.1.11 | Value analysis   | Not Provided   |
| § 4.1.1.12 | Detailed cost estimating beyond that required in Section 6.3         | Not Provided   |
| § 4.1.1.13 | On-site project representation                                       | Not Provided   |
| § 4.1.1.14 | Conformed documents for construction                                 | Not Provided   |
| § 4.1.1.15 | As-designed record drawings  | Not Provided   |
| § 4.1.1.16 | As-constructed record drawings                                       | Not Provided   |
| § 4.1.1.17 | Post-occupancy evaluation  | Not Provided   |
| § 4.1.1.18 | Facility support services  | Not Provided   |
| § 4.1.1.19 | Tenant-related services  | Not Provided   |
| § 4.1.1.20 | Architect's coordination of the Owner's consultants                  | Not Provided   |
| § 4.1.1.21 | Telecommunications/data design                                       | Not Provided   |
| § 4.1.1.22 | Security evaluation and planning                                     | Not Provided   |

| § 4.1.1.23 Cor | nmissioning   | N. D. St.    |
|----------------|---|--------------|
| § 4.1.1.24 Sus | tainable Project Services pursuant to Section 4.1.3 | Not Provided |
| § 4.1.1.25 Fas | t-track design services                             | Not Provided |
|                | tiple bid packages                                  | Not Provided |
|                | oric preservation                                   | Not Provided |
|                |   | Not Provided |
|                | niture, furnishings, and equipment design           | Not Provided |
|                | er services provided by specialty Consultants       | Not Provided |
| § 4.1.1.30 Oth | er Supplemental Services                            | Not Provided |

## § 4.1.2 Description of Supplemental Services

§ 4.1.2.1 A description of each Supplemental Service identified in Section 4.1.1 as the Architect's responsibility is provided below.

(Describe in detail the Architect's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit. The AIA publishes a number of Standard Form of Architect's Services documents that can be included as an exhibit to describe the Architect's Supplemental Services.)

N/A

§ 4.1.2.2 A description of each Supplemental Service identified in Section 4.1.1 as the Owner's responsibility is provided below.

Describe in detail the Owner's Supplemental Services identified in Section 4.1.1 or, if set forth in an exhibit, identify the exhibit.)

N/A

§ 4.1.3 If the Owner identified a Sustainable Objective in Article 1, the Architect shall provide, as a Supplemental Service, the Sustainability Services required in AIA Document E204™-2017, Sustainable Projects Exhibit, attached to this Agreement. The Owner shall compensate the Architect as provided in Section 11.2.

## § 4.2 Architect's Additional Services

The Architect may provide Additional Services after execution of this Agreement without invalidating the Agreement. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 4.2 shall entitle the Architect to compensation pursuant to Section 11.3 and an appropriate adjustment in the Architect's schedule.

- § 4.2.1 Upon recognizing the need to perform the following Additional Services, the Architect shall notify the Owner with reasonable promptness and explain the facts and circumstances giving rise to the need. The Architect shall not proceed to provide the following Additional Services until the Architect receives the Owner's written authorization:
  - .1 Services necessitated by a change in the Initial Information, previous instructions or approvals given by the Owner, or a material change in the Project including size, quality, complexity, the Owner's schedule or budget for Cost of the Work, or procurement or delivery method;
  - .2 Services necessitated by the enactment or revision of codes, laws, or regulations, including changing or editing previously prepared Instruments of Service;
  - ,3 Changing or editing previously prepared Instruments of Service necessitated by official interpretations of applicable codes, laws or regulations that are either (a) contrary to specific interpretations by the applicable authorities having jurisdiction made prior to the issuance of the building permit, or (b) contrary to requirements of the Instruments of Service when those Instruments of Service were prepared in accordance with the applicable standard of care;
  - .4 Services necessitated by decisions of the Owner not rendered in a timely manner or any other failure of performance on the part of the Owner or the Owner's consultants or contractors;
  - .5 Preparing digital models or other design documentation for transmission to the Owner's consultants and contractors, or to other Owner-authorized recipients;
  - Preparation of design and documentation for alternate bid or proposal requests proposed by the Owner;
  - Preparation for, and attendance at, a public presentation, meeting or hearing;

- .8 Preparation for, and attendance at, a dispute resolution proceeding or legal proceeding, except where the Architect is party thereto;
- .9 Evaluation of the qualifications of entities providing bids or proposals;
- .10 Consultation concerning replacement of Work resulting from fire or other cause during construction; or,
- .11 Assistance to the Initial Decision Maker, if other than the Architect.
- § 4.2.2 To avoid delay in the Construction Phase, the Architect shall provide the following Additional Services, notify the Owner with reasonable promptness, and explain the facts and circumstances giving rise to the need. If, upon receipt of the Architect's notice, the Owner determines that all or parts of the services are not required, the Owner shall give prompt written notice to the Architect of the Owner's determination. The Owner shall compensate the Architect for the services provided prior to the Architect's receipt of the Owner's notice.
  - Reviewing a Contractor's submittal out of sequence from the submittal schedule approved by the Architect;
  - .2 Responding to the Contractor's requests for information that are not prepared in accordance with the Contract Documents or where such information is available to the Contractor from a careful study and comparison of the Contract Documents, field conditions, other Owner-provided information, Contractor-prepared coordination drawings, or prior Project correspondence or documentation;
  - .3 Preparing Change Orders and Construction Change Directives that require evaluation of Contractor's proposals and supporting data, or the preparation or revision of Instruments of Service;
  - .4 Evaluating an extensive number of Claims as the Initial Decision Maker; or,
  - .5 Evaluating substitutions proposed by the Owner or Contractor and making subsequent revisions to Instruments of Service resulting therefrom.
- § 4.2.3 The Architect shall provide Construction Phase Services exceeding the limits set forth below as Additional Services. When the limits below are reached, the Architect shall notify the Owner:
  - .1 Two (2) reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor
  - .2 Bi-weekly visits to the site by the Architect during construction
  - .3 One (1) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents
  - .4 One (1) inspections for any portion of the Work to determine final completion.
- § 4.2.4 Except for services required under Section 3.6.6.5 and those services that do not exceed the limits set forth in Section 4.2.3, Construction Phase Services provided more than 60 days after (1) the date of Substantial Completion of the Work or (2) the initial date of Substantial Completion identified in the agreement between the Owner and Contractor, whichever is earlier, shall be compensated as Additional Services to the extent the Architect incurs additional cost in providing those Construction Phase Services.
- § 4.2.5 If the services covered by this Agreement have not been completed within Twenty Four (24) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

#### ARTICLE 5 OWNER'S RESPONSIBILITIES

- § 5.1 Unless otherwise provided for under this Agreement, the Owner shall provide information in a timely manner regarding requirements for and limitations on the Project, including a written program, which shall set forth the Owner's objectives; schedule; constraints and criteria, including space requirements and relationships; flexibility; expandability; special equipment; systems; and site requirements.
- § 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes; (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the

Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.

- § 5.3 The Owner shall identify a representative authorized to act on the Owner's behalf with respect to the Project. The Owner shall render decisions and approve the Architect's submittals in a timely manner in order to avoid unreasonable delay in the orderly and sequential progress of the Architect's services.
- § 5.4 The Owner shall furnish surveys to describe physical characteristics, legal limitations and utility locations for the site of the Project, and a written legal description of the site. The surveys and legal information shall include, as applicable, grades and lines of streets, alleys, pavements and adjoining property and structures; designated wetlands; adjacent drainage; rights-of-way, restrictions, easements, encroachments, zoning, deed restrictions, boundaries and contours of the site; locations, dimensions, and other necessary data with respect to existing buildings, other improvements and trees; and information concerning available utility services and lines, both public and private, above and below grade, including inverts and depths. All the information on the survey shall be referenced to a Project benchmark.
- § 5.5 The Owner shall furnish services of geotechnical engineers, which may include test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.
- § 5.6 The Owner shall provide the Supplemental Services designated as the Owner's responsibility in Section 4.1.1.
- § 5.7 If the Owner identified a Sustainable Objective in Article 1, the Owner shall fulfill its responsibilities as required in AIA Document E204™–2017, Sustainable Projects Exhibit, attached to this Agreement.
- § 5.8 The Owner shall coordinate the services of its own consultants with those services provided by the Architect. Upon the Architect's request, the Owner shall furnish copies of the scope of services in the contracts between the Owner and the Owner's consultants. The Owner shall furnish the services of consultants other than those designated as the responsibility of the Architect in this Agreement, or authorize the Architect to furnish them as an Additional Service, when the Architect requests such services and demonstrates that they are reasonably required by the scope of the Project. The Owner shall require that its consultants and contractors maintain insurance, including professional liability insurance, as appropriate to the services or work provided.
- § 5.9 The Owner shall furnish tests, inspections and reports required by law or the Contract Documents, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials.
- § 5.10 The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.
- § 5.11 The Owner shall provide prompt written notice to the Architect if the Owner becomes aware of any fault or defect in the Project, including errors, omissions or inconsistencies in the Architect's Instruments of Service.
- § 5.12 The Owner shall include the Architect in all communications with the Contractor that relate to or affect the Architect's services or professional responsibilities. The Owner shall promptly notify the Architect of the substance of any direct communications between the Owner and the Contractor otherwise relating to the Project. Communications by and with the Architect's consultants shall be through the Architect.
- § 5.13 Before executing the Contract for Construction, the Owner shall coordinate the Architect's duties and responsibilities set forth in the Contract for Construction with the Architect's services set forth in this Agreement. The Owner shall provide the Architect a copy of the executed agreement between the Owner and Contractor, including the General Conditions of the Contract for Construction.
- § 5.14 The Owner shall provide the Architect access to the Project site prior to commencement of the Work and shall obligate the Contractor to provide the Architect access to the Work wherever it is in preparation or progress.

§ 5.15 Within 15 days after receipt of a written request from the Architect, the Owner shall furnish the requested information as necessary and relevant for the Architect to evaluate, give notice of, or enforce lien rights.

#### ARTICLE 6 COST OF THE WORK

- § 6.1 For purposes of this Agreement, the Cost of the Work shall be the total cost to the Owner to construct all elements of the Project designed or specified by the Architect and shall include contractors' general conditions costs, overhead and profit. The Cost of the Work also includes the reasonable value of labor, materials, and equipment, donated to, or otherwise furnished by, the Owner. The Cost of the Work does not include the compensation of the Architect; the costs of the land, rights-of-way, financing, or contingencies for changes in the Work; or other costs that are the responsibility of the Owner.
- § 6.2 The Owner's budget for the Cost of the Work is provided in Initial Information, and shall be adjusted throughout the Project as required under Sections 5.2, 6.4 and 6.5. Evaluations of the Owner's budget for the Cost of the Work, and the preliminary estimate of the Cost of the Work and updated estimates of the Cost of the Work, prepared by the Architect, represent the Architect's judgment as a design professional. It is recognized, however, that neither the Architect nor the Owner has control over the cost of labor, materials, or equipment; the Contractor's methods of determining bid prices; or competitive bidding, market, or negotiating conditions. Accordingly, the Architect cannot and does not warrant or represent that bids or negotiated prices will not vary from the Owner's budget for the Cost of the Work, or from any estimate of the Cost of the Work, or evaluation, prepared or agreed to by the Architect.
- § 6.3 In preparing estimates of the Cost of Work, the Architect shall be permitted to include contingencies for design, bidding, and price escalation; to determine what materials, equipment, component systems, and types of construction are to be included in the Contract Documents; to recommend reasonable adjustments in the program and scope of the Project; and to include design alternates as may be necessary to adjust the estimated Cost of the Work to meet the Owner's budget. The Architect's estimate of the Cost of the Work shall be based on current area, volume or similar conceptual estimating techniques. If the Owner requires a detailed estimate of the Cost of the Work, the Architect shall provide such an estimate, if identified as the Architect's responsibility in Section 4.1.1, as a Supplemental Service.
- § 6.4 If, through no fault of the Architect, the Procurement Phase has not commenced within 90 days after the Architect submits the Construction Documents to the Owner, the Owner's budget for the Cost of the Work shall be adjusted to reflect changes in the general level of prices in the applicable construction market.
- § 6.5 If at any time the Architect's estimate of the Cost of the Work exceeds the Owner's budget for the Cost of the Work, the Architect shall make appropriate recommendations to the Owner to adjust the Project's size, quality, or budget for the Cost of the Work, and the Owner shall cooperate with the Architect in making such adjustments.
- § 6.6 If the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services is exceeded by the lowest bona fide bid or negotiated proposal, the Owner shall
  - .1 give written approval of an increase in the budget for the Cost of the Work;
  - .2 authorize rebidding or renegotiating of the Project within a reasonable time;
  - .3 terminate in accordance with Section 9.5;
  - .4 in consultation with the Architect, revise the Project program, scope, or quality as required to reduce the Cost of the Work; or,
  - .5 implement any other mutually acceptable alternative.
- § 6.7 If the Owner chooses to proceed under Section 6.6.4, the Architect shall modify the Construction Documents as necessary to comply with the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services, or the budget as adjusted under Section 6.6.1. If the Owner requires the Architect to modify the Construction Documents because the lowest bona fide bid or negotiated proposal exceeds the Owner's budget for the Cost of the Work due to market conditions the Architect could not reasonably anticipate, the Owner shall compensate the Architect for the modifications as an Additional Service pursuant to Section 11.3; otherwise the Architect's services for modifying the Construction Documents shall be without additional compensation. In any event, the Architect's modification of the Construction Documents shall be the limit of the Architect's responsibility under this Article 6.

## ARTICLE 7 COPYRIGHTS AND LICENSES

- § 7.1 The Architect and the Owner warrant that in transmitting Instruments of Service, or any other information, the transmitting party is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project.
- § 7.2 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements or for similar purposes in connection with the Project is not to be construed as publication in derogation of the reserved rights of the Architect and the Architect's consultants.
- § 7.3 The Architect grants to the Owner a nonexclusive license to use the Architect's Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations under this Agreement, including prompt payment of all sums due pursuant to Article 9 and Article 11. The Architect shall obtain similar nonexclusive licenses from the Architect's consultants consistent with this Agreement. The license granted under this section permits the Owner to authorize the Contractor, Subcontractors, Sub-subcontractors, and suppliers, as well as the Owner's consultants and separate contractors, to reproduce applicable portions of the Instruments of Service, subject to any protocols established pursuant to Section 1.3, solely and exclusively for use in performing services or construction for the Project. If the Architect rightfully terminates this Agreement for cause as provided in Section 9.4, the license granted in this Section 7.3 shall terminate.
- § 7.3.1 In the event the Owner uses the Instruments of Service without retaining the authors of the Instruments of Service, the Owner releases the Architect and Architect's consultant(s) from all claims and causes of action arising from such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the Architect and its consultants from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's use of the Instruments of Service under this Section 7.3.1. The terms of this Section 7.3.1 shall not apply if the Owner rightfully terminates this Agreement for cause under Section 9.4.
- § 7.4 Except for the licenses granted in this Article 7, no other license or right shall be deemed granted or implied under this Agreement. The Owner shall not assign, delegate, sublicense, pledge or otherwise transfer any license granted herein to another party without the prior written agreement of the Architect. Any unauthorized use of the Instruments of Service shall be at the Owner's sole risk and without liability to the Architect and the Architect's consultants.
- § 7.5 Except as otherwise stated in Section 7.3, the provisions of this Article 7 shall survive the termination of this Agreement.

## ARTICLE 8 CLAIMS AND DISPUTES

#### § 8.1 General

- § 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.
- § 8.1.2 To the extent damages are covered by property insurance, the Owner and Architect waive all rights against each other and against the contractors, consultants, agents, and employees of the other for damages, except such rights as they may have to the proceeds of such insurance as set forth in AIA Document A201–2017, General

Conditions of the Contract for Construction. The Owner or the Architect, as appropriate, shall require of the contractors, consultants, agents, and employees of any of them, similar waivers in favor of the other parties enumerated herein.

§ 8.1.3 The Architect and Owner waive consequential damages for claims, disputes, or other matters in question, arising out of or relating to this Agreement. This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination of this Agreement, except as specifically provided in Section 9.7.

#### § 8.2 Mediation

- § 8.2.1 Any claim, dispute or other matter in question arising out of or related to this Agreement shall be subject to mediation as a condition precedent to binding dispute resolution. If such matter relates to or is the subject of a lien arising out of the Architect's services, the Architect may proceed in accordance with applicable law to comply with the lien notice or filing deadlines prior to resolution of the matter by mediation or by binding dispute resolution.
- § 8.2.2 The Owner and Architect shall endeavor to resolve claims, disputes and other matters in question between them by mediation, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of this Agreement. A request for mediation shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of a complaint or other appropriate demand for binding dispute resolution but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration proceeding is stayed pursuant to this section, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.
- § 8.2.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

| dispu | te reso | lution s | do not re<br>hall be th<br>iate box.) | e follov |      | e throug | gh medi | ation p | ursuant | to this | Section | on 8.2, | the n | ethod | of bind | ing |
|-------|---------|----------|---------------------------------------|----------|------|----------|---------|---------|---------|---------|---------|---------|-------|-------|---------|-----|
| 2 12  | 2500    | 7        |                                       | 1201 W   | 2552 |          |         |         |         |         |         |         |       |       |         |     |

| ] | 1   | Arbitration pursuant to Section 8.3 of this Agreement |
|---|-----|---|
| 1 | X ] | Litigation in a court of competent jurisdiction       |
| I | ]   | Other: (Specify)                                      |

If the Owner and Architect do not select a method of binding dispute resolution, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, the dispute will be resolved in a court of competent jurisdiction.

#### § 8.3 Arbitration

- § 8.3.1 If the parties have selected arbitration as the method for binding dispute resolution in this Agreement, any claim, dispute or other matter in question arising out of or related to this Agreement subject to, but not resolved by, mediation shall be subject to arbitration, which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of this Agreement. A demand for arbitration shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the arbitration.
- § 8.3.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the claim, dispute or other matter in question would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the

Init.

arbitration shall constitute the institution of legal or equitable proceedings based on the claim, dispute or other matter in question.

- § 8.3.2 The foregoing agreement to arbitrate, and other agreements to arbitrate with an additional person or entity duly consented to by parties to this Agreement, shall be specifically enforceable in accordance with applicable law in any court having jurisdiction thereof.
- § 8.3.3 The award rendered by the arbitrator(s) shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof:

#### § 8.3.4 Consolidation or Joinder

- § 8.3.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation; (2) the arbitrations to be consolidated substantially involve common questions of law or fact; and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).
- § 8.3.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.
- § 8.3.4.3 The Owner and Architect grant to any person or entity made a party to an arbitration conducted under this Section 8.3, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Architect under this Agreement.
- § 8.4 The provisions of this Article 8 shall survive the termination of this Agreement.

#### ARTICLE 9 TERMINATION OR SUSPENSION

- § 9.1 If the Owner fails to make payments to the Architect in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Architect's option, cause for suspension of performance of services under this Agreement. If the Architect elects to suspend services, the Architect shall give seven days' written notice to the Owner before suspending services. In the event of a suspension of services, the Architect shall have no liability to the Owner for delay or damage caused the Owner because of such suspension of services. Before resuming services, the Owner shall pay the Architect all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.
- § 9.3 If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Architect, the Architect may terminate this Agreement by giving not less than seven days' written notice.
- § 9.4 Either party may terminate this Agreement upon seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.
- § 9.5 The Owner may terminate this Agreement upon seven days' written notice to the Architect for the Owner's convenience and without cause.
- § 9.6 If the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall compensate the Architect for services performed prior to

termination, Reimbursable Expenses incurred, and costs attributable to termination, including the costs attributable to the Architect's termination of consultant agreements.

§ 9.7 In addition to any amounts paid under Section 9.6, if the Owner terminates this Agreement for its convenience pursuant to Section 9.5, or the Architect terminates this Agreement pursuant to Section 9.3, the Owner shall pay to the Architect the following fees:

(Set forth below the amount of any termination or licensing fee, or the method for determining any termination or licensing fee.)

.1 Termination Fee:

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

.2 Licensing Fee if the Owner intends to continue using the Architect's Instruments of Service:

As mutually agreed

- § 9.8 Except as otherwise expressly provided herein, this Agreement shall terminate one year from the date of Substantial Completion.
- § 9.9 The Owner's rights to use the Architect's Instruments of Service in the event of a termination of this Agreement are set forth in Article 7 and Section 9.7.

## ARTICLE 10 MISCELLANEOUS PROVISIONS

- § 10.1 This Agreement shall be governed by the law of the place where the Project is located, excluding that jurisdiction's choice of law rules. If the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 8.3.
- § 10.2 Terms in this Agreement shall have the same meaning as those in AIA Document A201–2017, General Conditions of the Contract for Construction.
- § 10.3 The Owner and Architect, respectively, bind themselves, their agents, successors, assigns, and legal representatives to this Agreement. Neither the Owner nor the Architect shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement, including any payments due to the Architect by the Owner prior to the assignment.
- § 10.4 If the Owner requests the Architect to execute certificates, the proposed language of such certificates shall be submitted to the Architect for review at least 14 days prior to the requested dates of execution. If the Owner requests the Architect to execute consents reasonably required to facilitate assignment to a lender, the Architect shall execute all such consents that are consistent with this Agreement, provided the proposed consent is submitted to the Architect for review at least 14 days prior to execution. The Architect shall not be required to execute certificates or consents that would require knowledge, services, or responsibilities beyond the scope of this Agreement.
- § 10.5 Nothing contained in this Agreement shall create a contractual relationship with, or a cause of action in favor of, a third party against either the Owner or Architect.
- § 10.6 Unless otherwise required in this Agreement, the Architect shall have no responsibility for the discovery, presence, handling, removal or disposal of, or exposure of persons to, hazardous materials or toxic substances in any form at the Project site.
- § 10.7 The Architect shall have the right to include photographic or artistic representations of the design of the Project among the Architect's promotional and professional materials. The Architect shall be given reasonable access to the completed Project to make such representations. However, the Architect's materials shall not include the Owner's confidential or proprietary information if the Owner has previously advised the Architect in writing of the specific information considered by the Owner to be confidential or proprietary. The Owner shall provide professional credit for the Architect in the Owner's promotional materials for the Project. This Section 10.7 shall

survive the termination of this Agreement unless the Owner terminates this Agreement for cause pursuant to Section 9.4.

§ 10.8 If the Architect or Owner receives information specifically designated as "confidential" or "business proprietary," the receiving party shall keep such information strictly confidential and shall not disclose it to any other person except as set forth in Section 10.8.1. This Section 10.8 shall survive the termination of this Agreement.

§ 10.8.1 The receiving party may disclose "confidential" or "business proprietary" information after 7 days' notice to the other party, when required by law, arbitrator's order, or court order, including a subpoena or other form of compulsory legal process issued by a court or governmental entity, or to the extent such information is reasonably necessary for the receiving party to defend itself in any dispute. The receiving party may also disclose such information to its employees, consultants, or contractors in order to perform services or work solely and exclusively for the Project, provided those employees, consultants and contractors are subject to the restrictions on the disclosure and use of such information as set forth in this Section 10.8.

§ 10.9 The invalidity of any provision of the Agreement shall not invalidate the Agreement or its remaining provisions. If it is determined that any provision of the Agreement violates any law, or is otherwise invalid or unenforceable, then that provision shall be revised to the extent necessary to make that provision legal and enforceable. In such case the Agreement shall be construed, to the fullest extent permitted by law, to give effect to the parties' intentions and purposes in executing the Agreement.

#### ARTICLE 11 COMPENSATION

§ 11.1 For the Architect's Basic Services described under Article 3, the Owner shall compensate the Architect as follows:

.1 Stipulated Sum (Insert amount)

Compensation in the amount of Eight Thousand Five Hundred Dollars and Zero Cents (\$8,500.00)

.2 Percentage Basis (Insert percentage value)

( )% of the Owner's budget for the Cost of the Work, as calculated in accordance with Section 11.6.

.3 Other (Describe the method of compensation)

§ 11.2 For the Architect's Supplemental Services designated in Section 4.1.1 and for any Sustainability Services required pursuant to Section 4.1.3, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation. If necessary, list specific services to which particular methods of compensation apply.)

Hourly or as mutually agreed upon in advance of services

§ 11.3 For Additional Services that may arise during the course of the Project, including those under Section 4.2, the Owner shall compensate the Architect as follows: (Insert amount of, or basis for, compensation.)

Hourly or as mutually agreed upon in advance of services

§ 11.4 Compensation for Supplemental and Additional Services of the Architect's consultants when not included in Section 11.2 or 11.3, shall be the amount invoiced to the Architect plus percent (%), or as follows:

(Insert amount of, or basis for computing, Architect's consultants' compensation for Supplemental or Additional Services.)

As mutually agreed by Owner and Architect in advance of services

§ 11.5 When compensation for Basic Services is based on a stipulated sum or a percentage basis, the proportion of compensation for each phase of services shall be as follows:

| Total Basic Compensation | one hundred | percent ( | 100 | %) |
|--------------------------|-------------|-----------|-----|----|
| Construction Phase       | Twenty Five | percent ( | 25  | %) |
| Procurement Phase        | Five        | percent ( | 5   | %) |
| Phase                    |             |           |     |    |
| Construction Documents   | Twenty Five | percent ( | 25  | %) |
| Design Development Phase | Twenty Five | percent ( | 25  | %) |
| Schematic Design Phase   | Twenty      | percent ( | 20  | %) |

- § 11.6 When compensation identified in Section 11.1 is on a percentage basis, progress payments for each phase of Basic Services shall be calculated by multiplying the percentages identified in this Article by the Owner's most recent budget for the Cost of the Work. Compensation paid in previous progress payments shall not be adjusted based on subsequent updates to the Owner's budget for the Cost of the Work.
- § 11.6.1 When compensation is on a percentage basis and any portions of the Project are deleted or otherwise not constructed, compensation for those portions of the Project shall be payable to the extent services are performed on those portions. The Architect shall be entitled to compensation in accordance with this Agreement for all services performed whether or not the Construction Phase is commenced.
- § 11.7 The hourly billing rates for services of the Architect and the Architect's consultants are set forth below. The rates shall be adjusted in accordance with the Architect's and Architect's consultants' normal review practices. (If applicable, attach an exhibit of hourly billing rates or insert them below.)

N/A

| Employee or Category | Rate (\$0.00) |
|----------------------|---------------|
| N/A                  | N/A           |

#### § 11.8 Compensation for Reimbursable Expenses

- § 11.8.1 Reimbursable Expenses are in addition to compensation for Basic, Supplemental, and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:
  - Transportation and authorized out-of-town travel and subsistence;
  - .2 Long distance services, dedicated data and communication services, teleconferences, Project web sites, and extranets;
  - .3 Permitting and other fees required by authorities having jurisdiction over the Project;
  - .4 Printing, reproductions, plots, and standard form documents;
  - .5 Postage, handling, and delivery;
  - .6 Expense of overtime work requiring higher than regular rates, if authorized in advance by the Owner;
  - .7 Renderings, physical models, mock-ups, professional photography, and presentation materials requested by the Owner or required for the Project;
  - .8 If required by the Owner, and with the Owner's prior written approval, the Architect's consultants' expenses of professional liability insurance dedicated exclusively to this Project, or the expense of additional insurance coverage or limits in excess of that normally maintained by the Architect's consultants;
  - .9 All taxes levied on professional services and on reimbursable expenses;
  - .10 Site office expenses;
  - .11 Registration fees and any other fees charged by the Certifying Authority or by other entities as necessary to achieve the Sustainable Objective; and,
  - .12 Other similar Project-related expenditures.

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus Ten percent (10 %) of the expenses incurred.

§ 11.9 Architect's Insurance. If the types and limits of coverage required in Section 2.5 are in addition to the types and limits the Architect normally maintains, the Owner shall pay the Architect for the additional costs incurred by the Architect for the additional coverages as set forth below:

(Insert the additional coverages the Architect is required to obtain in order to satisfy the requirements set forth in Section 2.5, and for which the Owner shall reimburse the Architect.)

## § 11.10 Payments to the Architect

#### § 11.10.1 Initial Payments

§ 11.10.1.1 An initial payment of Zero (\$ 0 ) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of Zero (\$ 0 ) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred.

#### § 11.10.2 Progress Payments

§ 11.10.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid Forty-Five (45) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect. (Insert rate of monthly or annual interest agreed upon.)

#### 1.5 % Monthly

§ 11.10.2.2 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work, unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.

§ 11.10.2.3 Records of Reimbursable Expenses, expenses pertaining to Supplemental and Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

§ 11.10.2.4 Owner Agrees to receive invoices by email, which will be sent to the following individual and email address:

Full Name: Kelly Christopherson

Email Address: kelly.christopherson@k12.sd.us

## ARTICLE 12 SPECIAL TERMS AND CONDITIONS

Special terms and conditions that modify this Agreement are as follows: (Include other terms and conditions applicable to this Agreement.)

None

Init.

## ARTICLE 13 SCOPE OF THE AGREEMENT

§ 13.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both the Owner and Architect.

§ 13.2 This Agreement is comprised of the following documents identified below:

- .1 AIA Document B101TM\_2017, Standard Form Agreement Between Owner and Architect
- .2 AIA Document E203<sup>TM</sup>–2013, Building Information Modeling and Digital Data Exhibit, dated as indicated below:

(Insert the date of the E203-2013 incorporated into this agreement.)

| -  | F  |      |      |        |
|----|----|------|------|--------|
| .3 | Ex | Pro- | 13.1 | Port 1 |
|    |    |      |      |        |

(Check the appropriate box for any exhibits incorporated into this Agreement.)

[ ] AIA Document E204TM\_2017, Sustainable Projects Exhibit, dated as indicated below: (Insert the date of the E204-2017 incorporated into this agreement.)

[ ] Other Exhibits incorporated into this Agreement:

(Clearly identify any other exhibits incorporated into this Agreement, including any exhibits and scopes of services identified as exhibits in Section 4.1.2.)

Exhibit A: Project Initial Information

Exhibit B: JLG Image Release

Exhibit C: Schedule of Hourly Compensation & Reimbursable Expenses

.4 Other documents:

(List other documents, if any, forming part of the Agreement.)

This Agreement entered into as of the day and year first written above.

OWNER (Signature)

Garret Bischoff, President

(Printed name and title)

ARCHITECT (Signature)

Catherine Dekkenga, AIA, Principal

(Printed name, title, and license number, if required)

# Additions and Deletions Report for

AIA® Document B101™ - 2017

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 12:21:46 ET on 10/05/2022.

#### PAGE 1

AGREEMENT made as of the Twenty Third day of September in the year Two Thousand and Twenty Two

Huron Public School District 150 5th St. SW Huron, SD 57350

JLG Architects
232 S. Main Ave.
Sioux Falls, SD 57104

21392.01 – Huron School District – TAC Ceiling

Design and replacement of damaged ceiling in the Tiger Activities Center

Huron, SD

#### PAGE 2

#### TABLE OF ARTICLES

(Insert the Owner's program, identify documentation that establishes the Owner's program, or state the manner in which the program will be developed.)

Design for the replacement of damaged ceiling in the TAC.

The TAC is an existing pre engineered metal building owned by the Huron School District. The ceiling has been damaged and needs to be replaced.

```
$150,000.00
PAGE 3
        Design: July 1, 2022 - September 9, 2022
...
        Construction Documents: September 9, 2022 - October 21, 2022
        Bidding: October 22, 2022 - November 19, 2022
***
        Construction Administration: March 1, 2023 - July 19, 2023
        Sprin/Summer 2023
***
        N/A
...
        N/A
Design/Bid/Build
N/A
Kelly Christopherson
Huron Public School District
150 5th St. SW
Huron, SD 57350
Telephone:
Email:
```

N/A

# PAGE 4 N/A N/A N/A Catherine Dekkenga JLG Architects 232 S. Main Ave. Sioux Falls, SD 57104 Telephone: 605-271-2531 Email: cdckkenga@jlgarchitects.com SEA, Inc., Mark Smith \*\*\* 401 E. 8th St. Suite 201 Sioux Falls, SD 57103 Telephone: 605-334-0188 ... Associated Consulting Engineers, Inc., Jordan deWit 340 S. Phillips Ave. Sioux Falls, SD 57104

Telephone: 605-335-3720

|                                    | Associated Consulting Engineers, Inc., Brad Shoup  |
|------------------------------------|--|
| ;; <del>;;;</del>                  |  |
|                                    | 340 S. Phillips Ave.   |
|                                    |  |
|                                    | Sioux Falls, SD 57104  |
| PAGE 5                             |  |
|                                    | Telephone: 605-335-3720  |
| ***                                |  |
| N/A                                |  |
|                                    |  |
|                                    |  |
| N/A                                |  |
| ****                               |  |
| Cents (\$ 1,00                     | ercial General Liability with policy limits of not less than (\$-One Million Dollars and Zero 00,000.00 ) for each occurrence and Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) in the bodily injury and property damage.  |
| PAGE 6                             |  |
| limits of not le<br>death of any p | obile Liability covering vehicles owned, and non-owned vehicles used, by the Architect with policy as than (\$\frac{One Million Dollars and Zero Cents (\\$1,000,000.00)}{\}) per accident for bodily injury, erson, and property damage arising out of the ownership, maintenance and use of those motor with any other statutorily required automobile coverage. |
|                                    |  |
| and Zero Cent                      | rers' Liability with policy limits not less than (\$\sigma\$) each accident, (\$\sigma\$-Five Hundred Thousand Dollars s (\$\frac{500,000.00}{0.00}) each accident, Five Hundred Thousand Dollars and Zero Cents (\$\frac{500,000.00}{0.00}) each accident, Five Hundred Thousand Dollars and Zero Cents (\$\frac{500,000.00}{0.00}) policy limit.                 |
|                                    |  |
| services with p                    | ional Liability covering negligent acts, errors and omissions in the performance of professional policy limits of not less than (\$-Two Million Dollars and Zero Cents (\$ 2,000,000.00 ) per claim on Dollars and Zero Cents (\$ 4,000,000.00 ) in the aggregate.   |
| •••                                |  |
| § 3.1 The Arch                     | itect's Basic Services consist of those described in this Article 3 and include usual and oustomary  |

structural, mechanical, and electrical engineering services. Services not set forth in this Article 3 are Supplemental

or Additional Services.

#### PAGE 7

§ 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner alternative approach(es) to design and construction of the Project. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.

...

§ 3.2.5.2 The Architect shall consider the value of alternative-materials, building systems and equipment, together with other considerations based on program and aesthetics, in developing a design for the Project that is consistent with the Owner's program, schedule, and budget for the Cost of the Work.

#### PAGE 9

§ 3.6.2.1 The Architect shall visit the site at intervals appropriate to the stage points mutually agreed upon with Owner and Contractor prior to or at the start of construction, or as otherwise required in Section 4.2.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and promptly report to the Owner (1) known deviations from the Contract Documents, (2) known deviations from the most recent construction schedule submitted by the Contractor, and (3) defects and deficiencies observed in the Work.

#### PAGE 10

§ 3.6.4.2 The Architect shall review and approve, review, or take other appropriate action upon, per specifications, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

#### PAGE 11

#### § 3.6.7 Disclaimer of Duty to Review Supplementary Video Information

- The Parties agree and understand that, during the course of the construction of the Work, the Owner,
  Contractor, or other Project participants may choose to, at their own expense and initiative, make use of
  supplemental video imagery and information sourced from devices such as project webcams, Unmanned
  Aerial Vehicles, including but not limited to drones, and/or other videography equipment, in order to obtain
  visual information and documentation regarding the state of the progress and/or quality of the Work
  (collectively "Supplementary Video Information"). Notwithstanding the foregoing, the Parties expressly
  agree and acknowledge the following:
  - Since Supplementary Video Information encompasses only certain limited visual information
    and provides neither complete nor comprehensive visual information/measurements/dimensions,
    nor any tactile, auditory, or olfactory feedback whatsoever, such Supplementary Video
    Information shall not be used as a substitute for or required supplement to Architect's Construction
    Phase services, herein, but rather may, to the extent provided and made available to Architect in a
    reasonably timely, legible, and complete fashion, supplement the other data and information
    obtained by Architect pursuant to Architect's Construction Phase Services, herein.
  - Architect shall have no obligation to conduct a comprehensive or continuous review of any
    Supplementary Video Information, but rather may, at its sole discretion, choose to review such
    Supplementary Video Information, or any portion thereof, in connection with the performance of

#### its Construction Phase Services on the Project.

#### PAGE 12

3. Architect's receipt of any Supplementary Video Information shall not in any way change the provisions of the Construction Phase Services described, herein, to the effect that Architect's observation of the Work shall be limited to becoming generally familiar with the progress and quality of the Work and to determine, in general, if the Work observed appears to be in general conformance with the Contract Documents based upon the Architect's site visits of the Project during construction pursuant to this Agreement, and that Architect shall not be required to make continuous or exhaustive on-site inspections of any kind with respect to the quality, quantity, or progress of the Work and, under no circumstances shall Architect have any responsibility with respect to the Contractor's construction means, methods, techniques, sequences, or procedures with respect to the Work or any portion or part

#### ARTICLE thereof.

...

#### ARTICLE 4 SUPPLEMENTAL AND ADDITIONAL SERVICES

#### PAGE 13

| § 4.1.1.1  | Programming  | Not Provided |
|------------|--|--------------|
| § 4.1.1.2  | Multiple preliminary designs   | Not Provided |
| § 4.1.1.3  | Measured drawings  | Not Provided |
| § 4.1.1.4  | Existing facilities surveys  | Not Provided |
| § 4.1.1.5  | Site evaluation and planning   | Not Provided |
| § 4.1.1.6  | Building Information Model management responsibilities               | Not Provided |
| § 4.1.1.7  | Development of Building Information Models for post construction use | Not Provided |
| § 4.1.1.8  | Civil engineering  | Not Provided |
| § 4.1.1.9  | Landscape design   | Not Provided |
| § 4.1.1.10 | Architectural interior design  | Not Provided |
| § 4.1.1.11 | Value analysis   | Not Provided |
| § 4.1.1.12 | Detailed cost estimating beyond that required in Section 6.3         | Not Provided |
| § 4.1.1.13 | On-site project representation                                       | Not Provided |
| § 4.1.1.14 | Conformed documents for construction                                 | Not Provided |
| § 4.1.1.15 | As-designed record drawings  | Not Provided |
| § 4.1.1.16 | As-constructed record drawings                                       | Not Provided |
| § 4.1.1.17 | Post-occupancy evaluation  | Not Provided |
| § 4.1.1.18 | Facility support services  | Not Provided |
| § 4.1.1.19 | Tenant-related services  | Not Provided |
| § 4.1.1.20 | Architect's coordination of the Owner's consultants                  | Not Provided |
| § 4.1.1.21 | Telecommunications/data design                                       | Not Provided |
| § 4.1.1.22 | Security evaluation and planning                                     | Not Provided |
| § 4.1.1.23 | Commissioning  | Not Provided |
| § 4.1.1.24 | Sustainable Project Services pursuant to Section 4.1.3               | Not Provided |
| § 4.1.1.25 | Fast-track design services   | Not Provided |
| § 4.1.1.26 | Multiple bid packages  | Not Provided |

| § 4.1.1.27 | Historic preservation                            | Not Provided |
|------------|--|--------------|
| § 4.1.1.28 | Furniture, furnishings, and equipment design     | Not Provided |
| § 4.1.1.29 | Other services provided by specialty Consultants | Not Provided |
| § 4.1.1.30 | Other Supplemental Services                      | Not Provided |

N/A

N/A

#### PAGE 14

- .1 <u>Two (2)</u> reviews of each Shop Drawing, Product Data item, sample and similar submittals of the Contractor
- .2 () Bi-weekly visits to the site by the Architect during construction
- .3 One (1) inspections for any portion of the Work to determine whether such portion of the Work is substantially complete in accordance with the requirements of the Contract Documents.
- .4 One (1) inspections for any portion of the Work to determine final completion.

§ 4.2.5 If the services covered by this Agreement have not been completed within <u>Twenty Four</u> (24) months of the date of this Agreement, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

#### PAGE 15

§ 5.2 The Owner shall establish the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1; 5.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. The Owner acknowledges that the contingency allowance is to cover funding of unpredictable changes in the work or items of work during the construction phase, serving three core purposes: (1) To account for coordination issues in the Construction Documents, including errors and omissions, (2) to modify or change the scope of the project, (3) to accommodate unknown conditions. The Owner shall update the Owner's budget for the Project as necessary throughout the duration of the Project until final completion. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree to a corresponding change in the Project's scope and quality.

#### PAGE 17

§ 8.1.1 The Owner and Architect shall commence all claims and causes of action against the other and arising out of or related to this Agreement, whether in contract, tort, or otherwise, in accordance with the requirements of the binding dispute resolution method selected in this Agreement and within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1. Any such claim, dispute

or matter in question shall be subject to Informal Negotiation as first course of resolution. Either party may initiate informal negotiation of its claim, assuming the claim has been asserted within the time limits set forth in this Agreement by giving the other party written notice of such demand. Within ten (10) days of receipt of demand for informal negotiations, representatives of the parties with authority to settle the claim will meet to determine if the claim can be resolved informally. If informal negotiation fails to resolve the claim, mediation as a condition precedent to binding dispute resolution. If the parties do not resolve a claim, dispute, or matter in question through mediation, the method of binding dispute resolution shall be as selected in the sections to follow.

#### PAGE 18

[X] Litigation in a court of competent jurisdiction

#### PAGE 19

§ 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension and as planned and communicated per schedule for two (2) weeks following suspension. When the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

...

§ 9.4 Either party may terminate this Agreement upon not less than-seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.

300

§ 9.5 The Owner may terminate this Agreement upon not less than seven days' written notice to the Architect for the Owner's convenience and without cause.

#### PAGE 20

An amount equal to the average of the Architect's previous two invoices, prior to the date of termination.

•••

As mutually agreed

#### PAGE 21

Compensation in the amount of Eight Thousand Five Hundred Dollars and Zero Cents (\$8,500.00)

\*\*\*

Hourly or as mutually agreed upon in advance of services

...

Hourly or as mutually agreed upon in advance of services

#### PAGE 22

As mutually agreed by Owner and Architect in advance of services

555

| Schematic Design Phase                  | Twenty              | percent (              | 20             | 9/6)     |
|---|---------------------|------------------------|----------------|----------|
| Design Development Phase                | Twenty Five         | percent (              | 25             | %)       |
| Construction Documents<br>Phase         | Twenty Five         | percent (              | 25             | %)       |
| Procurement Phase<br>Construction Phase | Five<br>Twenty Five | percent (<br>percent ( | <u>5</u><br>25 | %)<br>%) |

N/A

\*\*\*

....

PAGE 23

N/A

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus <u>Ten</u> percent (<u>10</u> %) of the expenses incurred.

\*\*\*

§ 11.10.1.1 An initial payment of Zero\_(\$ 0\_) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

222

§ 11.10.1.2 If a Sustainability Certification is part of the Sustainable Objective, an initial payment to the Architect of <a href="Zero">Zero</a> (\$ 0 ) shall be made upon execution of this Agreement for registration fees and other fees payable to the Certifying Authority and necessary to achieve the Sustainability Certification. The Architect's payments to the Certifying Authority shall be credited to the Owner's account at the time the expense is incurred.

...

§ 11.10.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid Forty-Five (45\_) days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect.

...

1.5 % Monthly

...

§ 11.10.2.4 Owner Agrees to receive invoices by email, which will be sent to the following individual and email address:

...

Full Name: Kelly Christopherson

Email Address: kelly.christopherson@k12.sd.us

| ***  |                                    |
|--|------------------------------------|
| None   |                                    |
| PAGE 24  |                                    |
| Exhibit A: Project Initial Information         |                                    |
| 106  |                                    |
| Exhibit B: JLG Image Release                   |                                    |
|  |                                    |
| Exhibit C: Schedule of Hourly Compensation & F | Reimbursable Expenses              |
|  |                                    |
| Garret Bischoff, President                     | Catherine Dekkenga, AIA, Principal |

# Certification of Document's Authenticity

AIA® Document D401™ - 2003

I, Catherine Dekkenga, AIA, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 12:21:46 ET on 10/05/2022 under Order No. 2114257989 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA® Document B101TM - 2017, Standard Form of Agreement Between Owner and Architect, as published by the AIA in its software, other than those additions and deletions shown in the associated Additions and Deletions Report.

| (Signed) |  |
|----------|--|
|          |  |
|          |  |
| Title)   |  |
|          |  |
|          |  |
| Dated)   |  |